Developing an Airport Energy System Optimisation Framework

The Example of Amsterdam Schiphol

Master Thesis Aerospace Engineering - Control & Operations June 2024 - February 2025

Christoph Pabsch



Developing an Airport Energy System Optimisation Framework

The Example of Amsterdam Schiphol

Master Thesis Aerospace Engineering - Control & Operations

by

Christoph Pabsch

to obtain the degree of Master of Science at the Delft University of Technology, to be defended publicly on Tuesday February 18, 2025.

Student number:5064015Project duration:June 2024 - February 2025Thesis committee:Ir. P.C. Roling, TU Delft, supervisorDr. A. Bombelli, TU DelftDr. J. Sun, TU DelftDr. A.C. Braz, EPFL, supervisor

An electronic version of this thesis is available at http://repository.tudelft.nl/.

Cover image: © Christoph Pabsch, 2023.



Acknowledgements

As the nine months of writing my thesis come to an end, I want to thank the many people who have shaped my work and supported me during this time. First and most importantly, I want to thank my supervisors, Catarina Braz and Paul Roling. Over the past months, we have had numerous meetings, both in Sion and Delft, online and in person, and you have constantly supported me with feedback, new ideas, and valuable discussions. You always made me believe in myself and that I could achieve the results I have now accomplished! Moreover, I want to thank Prof. François Maréchal for permitting me to work on this thesis topic at IPESE and for your countless ideas for my project.

This thesis would not have been possible without the data I obtained on energy demand at airports. A big thank you goes to Koen from KLM, who supported me with trip data for the ground support equipment of KLM at Amsterdam Schiphol. Another important thank you goes to Cora from Amsterdam Schiphol Airport, who provided me with essential energy consumption data for the airport, without which this work would not have been possible to this extent. Furthermore, I also want to thank Junzi Sun from TU Delft, who helped me to obtain flight data from OpenSky.

Next, I want to thank those who allowed me to work on this thesis at EPFL and TU Delft. Most importantly, I want to thank Jonas for motivating me for the energy domain in the Advanced Energetics course during my Bachelor exchange semester at EPFL, for keeping contact while setting up the semester project I worked on in Autumn 2023 at IPESE, and without whom I would never have come to EPFL for my Master's thesis. Thanks also to Sylvie for handling all the organisational matters and welcoming me to the lab. At TU Delft, I especially thank Jeanet Rink and Prof. Carlos Simao Ferreira, who supported my plans to perform my Master's thesis abroad in Switzerland.

During the last nine months, I spent countless days working on my thesis in the open space of IPESE, together with all other postdocs, PhD candidates, Master exchange students, etc. Thank you, Catarina, Mael, David, Arthur, Cyrille, Joao, Luc, Cédric, Arthur, Lorenzo, Pullah, Yi and everyone else at IPESE, for your support, help, ideas, discussions and time spent together. I have felt like a part of IPESE since my semester project in 2023. Beyond that, I want to particularly say thank you to my fellow (mostly) Master students at IPESE: Oktay, Giulia, Ziqian, Amir, Sai, Ettore and Tayna. You have shaped my time at IPESE, and we have become friends over the past weeks and months, motivating and supporting but also distracting each other, for example, during numerous ping pong matches.

In addition to IPESE, it has been the people from my student residence who have made me feel at home in Sion, with whom I have created many wonderful memories and who thus always helped me to stay motivated for my Master's thesis. Thanks especially to Justine, Noémie, Margaux, Emelie, Pauline, Zac and Johan for the many dinners, evenings and other activities we could experience together and the many conversations we shared. Merci pour tout ce que nous avons vécu ensemble, vous êtes devenus de bons amis que j'espère revoir très bientôt!

At TU Delft, I would like to thank all professors, staff members, teaching assistants, etc., who made me live an unforgettable five years at the faculty of Aerospace Engineering, which will always be a part of me. A huge thanks goes to my fellow students, with whom I spent my whole Master's and Bachelor's. Most importantly, thank you, Wouter and Constança, for the time we spent together on various projects, lectures, and other things besides our studies and for our friendship. Bedankt and obrigado! I am sure that we will continue to keep in touch even from a distance, as we have done over the past year already.

Last but not least, I want to thank my friends from high school, my friends from my scholarship, Cusanuswerk and my family, who have made me the person I am today and with whom I could create unforgettable memories besides my studies over the past years. Above all, I would like to thank my parents: you have always been there for me, allowed me to do what I enjoy and what interests me, and supported me in my plans, just like when I am again moving to new destinations. Danke, dass ihr immer für mich da seid!

Christoph Pabsch Sion, February 2025

Contents

	Li	of Abbreviations	vii
	In	oduction	ix
I	Sc	ntific Paper	1
II	Li	erature Study	63
	1	introduction	65 65
	2	 Airport Energy Demands 2.1 Overview of Energy Consumption of Real Airports 2.2 Characteristics of Energy Consumption of Real Airports 2.3 Determining Overall Airport Energy Demand 2.4 Demands of Different Airport Areas 2.4.1 Landside Area 2.4.2 Airside Area 2.5 Factors Defining Airport Energy Demand 2.6 Methods for Quantifying Airport Energy System 3.1 Buildings 	68 69 70 70 70 72 73 75
		3.2 Operations . <t< td=""><td>78</td></t<>	78
	4	Energy Technologies Suitable for Airport Environment4.1Electricity Generation.4.2Heating and Cooling.4.3Energy Storage.4.4Production of Hydrogen4.5Production of Sustainable Aviation Fuels.	83 86 89
	5	Research Gap	91
	6	Research Plan 5.1 Research Question	

List of Abbreviations

APU	Auxiliary Power Unit
EU	European Union
HFSP	Hybrid Fuzzy-Stochastic Programming
HOMER	Hybrid Optimisation Model for Electric Renewable
HVAC	Heating, Ventilation and Air-Conditioning
IPESE	Industrial Process and Energy Systems Engineering group at EPFL
KPI	Key Performance Indicators
LP	Linear Programming
MCDM	Multiple Criteria Decision Making
MILP	Mixed Integer Linear Programming
PEM	Polymer Electrolyte Membranes
PSO	Particle Swarm Optimisation
PV	Photovoltaic Cells
REMM	Rational Exergy Management Model
SAF	Sustainable Aviation Fuel
SMR	Steam Methane Reforming
SOEC	Solid Oxide Electrolysers
SQP	Sequence Quadratic Programming

Introduction

Airports are key mobility hubs in the current century. They connect humanity across different continents and allow fast transport of passengers and goods. Airports are thus enabling aviation and have a significant impact on the economic development of their surrounding region [1]. Nevertheless, airports also have negative impacts, both on their neighbouring environment and on the climate. They create noise and environmental pollution [2] and are a significant energy consumer. Reducing their energy consumption could thus have an important local and global effect towards a more emission-free world.

The Industrial Process and Energy Systems Engineering (IPESE) group at EPFL, where this thesis has been conducted, has substantial expertise in the domain of energy system engineering, the integration of renewables and the design of complex energy systems. Research domains of this group include the topics of process integration, process design, urban systems and energy planning [3]. Several tools were developed by the group for modelling and optimising energy systems. One of them is OSMOSE, a tool for designing and analysing integrated energy systems, which has been used for the optimisations performed in this study.

In the thesis, an energy system optimisation framework is developed, allowing the analysis and comparison of various technology configurations for different airports. The demand is estimated using a simple model applicable to any airport. It is developed based on data obtained from contacts at KLM and Amsterdam Schiphol Airport. Next, a superstructure of energy technologies is built up, which includes models defined based on literature as well as (adapted) models from the IPESE group. Together, the energy demand model and the energy system optimisation framework in OSMOSE allow to answer the research question of this thesis, which is defined as follows:

"To what extent could European airports, such as Schiphol airport, benefit from the concept of renewable energy hubs in reducing their sustainability impact and increasing their energy supply self-sufficiency while fulfilling the varying energy demand profile from airport operations and infrastructure?"

This thesis report is split into two separate parts. Part I includes the scientific article, which contains the research performed during the thesis, including the methodology and results for estimating the energy demand and optimising the energy system. In Part II, the literature review is presented, the research gap is identified and a research plan is developed.

J Scientific Paper

Developing an Airport Energy System Optimisation Framework: The Example of Amsterdam Schiphol

Christoph Pabsch^{1,2}

Supervisors: Ana Catarina Gouveia Braz², Paul Roling¹

Abstract

Airports play a significant role in economic development while presenting considerable energy consumption with demands from various stakeholders. Their important environmental impact can be reduced through an optimised energy system design. In this study, a systematic framework for the generation and comparison of airport energy systems is developed. Mixed integer linear programming and multi-objective optimization are employed to generate different system configurations, which are then compared using multi-criteria analysis. The airport energy demands are estimated based on a simple model for the building and the airfield handling area, using a limited set of parameters available for any airport. First results for Amsterdam Schiphol and five other European hub airports indicate that fully renewable energy systems can be competitive, especially when integrating battery storage for daily energy management. Self-sufficiencies of around 80% can be economically viable, relying on energy systems employing photovoltaic cells, a CO_2 network, batteries and grid electricity.

Nomenclature

Abbreviations

			v
AEC	Alkaline electrolyser	A	surface area
BAT	Battery	ACPH	air changes j
BOI	Boiler	C	cost
CC	Combined cycle	C_p	specific heat
CHI	Chiller	ЃОР	coefficient of
$\rm CO2N$	CO ₂ network	d	distance
DHW	Domestic hot water	D	Frequency of
EDDM	Munich airport	DA	people degre
EHAM	Amsterdam Schiphol airport	ė	electricity lo
ENGM	Oslo-Gardermoen airport	E	electricity
\mathbf{FT}	Fischer-Tropsch	E_{sp}	specific elect
GSE	Ground support equipment	En	energy
GWP	Global warming potential	es	energy store
H_2S	Hydrogen storage system	Ex	exergy
HP	Heat pump	f	unit sizing fa
HVAC	Heating, ventilation and air conditioning	F	fixed unit siz
Inv	Investment cost	G	global warm
LEMD	Madrid-Barajas airport	GI	global incide
LIRF	Rome-Fiumicino airport	'n	natural gas l
LSZH	Zurich airport	H	height
Op	Operating cost	hc	heat exchang
PEM	Proton-exchange membrane	ir	interest rate
PV	Photovoltaic cell	Ι	irradiance
SOEC	Solid oxide electrolyser	\bar{k}	factor
SOFC	Solid oxide fuel cell	l	layer

Latin Symbols

A	surface area	m^2
ACPH	air changes per hour	-
C	$\cos t$	€
C_p	specific heat capacity	kJ/kg/K
\hat{COP}	coefficient of performance	-
d	distance	m
D	Frequency of occurrence	-
DA	people degree of activity	-
\dot{e}	electricity load	kW
E	electricity	kW
E_{sp}	specific electricity demand	Wh/m^3
\dot{En}	energy	kW
es	energy stored	kWh
Ex	exergy	kW
f	unit sizing factor	-
F	fixed unit size	-
G	global warming potential	tCO_2eq
GI	global incident radiation	W/m^2
\dot{h}	natural gas load	kW
H	height	m
hc	heat exchanged heat cascade	kW
ir	interest rate	-
Ι	irradiance	W/m^2
k	factor	-
l	layer	-

¹Department of Control & Operations, Faculty of Aerospace Engineering, TU Delft

²Industrial Process and Energy Systems Engineering, EPFL

lt	lifetime	у
LF	load factor	-
MET	standard metabolic heat human	W
n	number	-
P	power	kW
\dot{q}	heat load	kW
\overline{RES}	renewable electricity share	-
rs	resource stream	kg/h
SS	self-sufficiency	-
t	time	s
T	temperature	$^{\circ}\mathrm{C}$
U	thermal transmittance	$\mathrm{W/m^2/K}$
v	velocity	km/h
\dot{V}	volume flow	m^3/h
y	binary variable unit installation	-
YC	yearly annualised costs	€/y
YG	yearly greenhouse gas pot.	$tCO_2 eq/y$

Greek Symbols

α	weight objective function	-
ϵ	fraction/share	-
η	efficiency	-
ξ	recovery rate	-
ρ	density	$ m kg/m^3$
ϕ	solar transmission factor	-
Φ	heat input power	W/m^2
Ψ	fresh air ratio	-

Subscripts

b	building
с	consumption
cell	cell
$^{\rm ch}$	charging
cw	cold water before heating
cond	condenser
d	daily
dhw	domestic hot water
dr	driving
ea	electric appliances
el	electrical/electricity
en	energy
evap	evaporator
ex	exergy
ext	exterior
f	flight
fa	fresh air
g	gains (solar and interior)
gen	generator efficiency
h	hourly
hc	heat cascade
hw	hot water after heating
II	second law
in	

int	interior
irr	solar irradiation
level	level of energy storage
max	maximum
med	median
\min	minimum
out	going out
р	period
pax	passenger
\mathbf{pr}	process
r	rated
rad	radiation
real	real
\mathbf{S}	stream
STC	standard test condition
\mathbf{SW}	swept
t	time instance
$^{\mathrm{th}}$	theoretical
Т	Temperature
u	unit
\mathbf{ut}	utility
v	vehicle
\mathbf{vs}	ventilation system
wa	building wall
wi	building window
wt	wind turbine
У	yearly

Superscripts

+	imported
-	exported
cap	capital
dem	demand
el	electricity
fuel	fuel
gr	grid
inv	investment
GSE	ground support equipment
loc	local
\min	minimum
ng	natural gas
nr	non-renewable
op	operating
ren	renewable
wt	wind turbine
\mathbf{PV}	photovoltaic
Sets	

${old U}$	set of units
\boldsymbol{L}	set of layers
\boldsymbol{P}	set of periods (typical days)
T	set of time instances
$oldsymbol{S}$	set of streams

1 Introduction

Airports have a significant importance for economical development of their surrounding region due to their function of moving cargo and people [1]. Overall, there are more than 2000 airports in Europe [2], moving 2.3 billion passengers [3]. On the other hand, airports also have an important environmental impact, which is not only coming from the aircraft emissions but also from other sources such as ground support equipment, heating and cooling of the airport or construction work for renewing its infrastructure [4]. In fact, Amsterdam Schiphol emitted more than 75,000tCO₂eq in 2023 [5], being equivalent to the emissions per passenger for 160,000 flights from Amsterdam to New York [6].

The airport's energy system is the main contributor to the airport's emissions. Amsterdam Schiphol consumed over 254GWh of energy in 2023 [5]. This is equivalent to the electricity consumption of nearly 100,000 households per year [7] and thus twice the household electricity consumption of a small city like Delft [8], [9]. Data from 20 European airports allows to make an estimate of an average energy demand of 6.3kWh passenger [5], [10]–[32], which would lead together with the yearly total number of 2.3 billion passengers in 2023 [3] to an energy consumption of 14.5TWh for all European airports together in one year, which is equivalent to the annual electricity production of smaller European countries like Croatia [33].

In consequence, there is a huge potential to reduce energy consumption and CO_2 emissions of airports by a smart design of integrated airport energy systems, both through a good choice of energy technologies as well as by reusing waste or excess energy. Creating a universally applicable optimisation model for designing energy systems, which can be used in different environmental characteristics, could thus help in creating a baseline for developing and renewing energy systems of airports.

The goal of this research is to develop an energy system model to analyse to what extent European airports, such as Amsterdam Schiphol, could benefit from the concept of renewable energy hubs in reducing their sustainability impact and increasing their energy supply self-sufficiency while fulfilling the varying demands from airport operations and infrastructure. A renewable energy hub connects different energy storage and conversion technologies to optimise their performance and maximise renewable energy use [34].

The energy demand is estimated from airport operational, structural and weather data, which can be gathered for any European airport, and includes the building as well as the airfield handling demand. Hourly profiles are obtained, which are clustered in typical days as input for the energy system optimisation. An extensive superstructure is constructed, including various technologies for heating, cooling, local electricity production and storage as well as the possible production of synthetic fuels. The energy system is optimised and the energy demand is balanced using the OSMOSE tool for designing integrated energy systems [35]. Possible configurations are analysed using a set of key performance indicators including costs, global warming potential (GWP), exergy efficiency, self-sufficiency and renewable electricity share.

A simple but robust model for estimating the energy demand and designing the energy system has been successfully developed and has been applied to the case of Amsterdam Schiphol and five other European hub airports. First results indicate an energy system consisting of a combination of photovoltaic cells (PV), electricity from the grid, a battery, a CO_2 network and a combined cycle from natural gas to be most robust. A fully renewable energy system is achievable at only slightly higher costs. The introduction of energy storage technologies is beneficial, especially when not using natural gas. Batteries for daily energy storage appear to be the most beneficial energy storage option. A fully self-sufficient airport is possible at high investment costs, whereas self-sufficiencies of 80% are observed to be economically viable.

This paper will start with a description of relevant literature in section 2. Next, the methodology, both for estimating the energy demand and designing the energy system, is described in section 3. The case studies employed in this research are depicted in section 4. Results are presented and analysed in section 5, before section 6 concludes this research with overall conclusions and recommendation for future work.

2 Related Work

Airports are complex systems, having a diverse set of stakeholders, energy demands and limitations to take into account. A description of the environment in section 2.1 is followed by a summary of relevant research in the fields of airport energy demand estimation and system optimisation in section 2.2 and section 2.3 respectively. To conclude, section 2.4 identifies the research gaps addressed by this study.

2.1 Airport Environment and Energy Demand

Aviation, being a main driver of globalisation, relies on globally present airport infrastructures, with large economic centres often also being close to large airports [36]. They thus have a key role in the current transportation system. The largest airports move more than 100 million passengers per year [37]. Many different stakeholders are involved in an airport, including the passengers, the air carriers, the airport authority and service providers as well as local organisations, communities and governments [38]. Their interests are diverse, ranging from efficient and smooth airport operation, minimal costs, growing passengers numbers and maximum revenue up to a minimised impact on the environment and local communities.

The airport itself can be divided in two main parts: the landside area consisting of for example passenger terminal buildings, freight terminals and parking lots as well as the airside area consisting of the terminal airspace, taxiways, runways, etc. [39].

Due to their size, airports have a high energy consumption, with Amsterdam Schiphol having an energy consumption of 254GWh for moving 61.9 million passengers in 2023 [5]. The yearly energy consumption for 20 European airports can be seen in Figure 1. While in general the demands increase with the number of passengers, significant variations can be observed due to different airport layouts, energy systems and measurement methods. Looking at the energy usages for six airports where data is available [12]–[14], [16], [17], [21], it can be observed that direct electricity demand is the most important with 57% of the overall demand on average, followed by heating with 31%, cooling with 12% and fuel with 12%. The most important energy resources are observed to be electricity from the grid, natural gas and diesel [10], [11], [15], [18], [23], [24].



Figure 1: Overview of energy demands per number of passengers of a selection of 15 individual European airports and five European airport groups [5], [10]–[32]

Reducing an airport's energy demand is of crucial importance for reducing its environmental impact. Possible pathways would be for example optimised operation [40] or the introduction of new technologies such as renewable electricity sources, energy-efficient technologies or electric vehicles [41].

2.2 Energy Demand Estimation

The energy demand of the airport can be divided in the airside and landside parts of the airport. Ortega Alba [42] further subdivided the areas and the main facilities, being for example lighting, heating, ventilation, etc. and defined hourly up to seasonal demand profiles, finding that the time of day and the season determine most the energy demand. Lin, Liu, Zhang, et al. [43] observed that the highest energy demands occur in climate zones with hot summers and/or cold winters. The energy consumption in a freezing environment can be two to four times higher than in mild environments [44]. Furthermore, the terminal surface area and the flow of passengers have an important influence on the buildings' energy demand [43], as well as the spatio-temporal distribution of the passengers in the building [45].

An optimal energy system that is based on different types of requirements and resources can be computed after having detailed information and estimation on the energy demand, in terms of the type of demand (electricity, thermal, fuel), the location of the demand and the variation over the year, as is also done in the approach of Rubeis, Nardi, Paoletti, et al. [46]. Studies on modern terminal buildings, such as the one performed by Xianliang, Jingchao, Zhiwen, et al. [47], suggested that the systems related to heating, cooling and ventilation are the largest energy consumers and the ones with the most significant distribution over space and time. Moreover, Yildiz, Yilmaz, and Celik [48] suggested changing the set-point temperature for heating and cooling and renewing those technologies as some of the main strategies for reducing energy demand and emissions. However, substantial reductions in emissions can also be achieved by optimising other systems in the building environment, such as the baggage handling system, as is studied by Lodewijks, Cao, Zhao, et al. [49]. As regards the vehicles around an airport, it can be seen that the majority operates on the airside area with activities related to aircraft loading, towing, take-off preparation or others, as is described by Liu, Li, Liu, et al. [50]. Vehicle energy demand is found to be dependent on the flight schedule especially for the airside vehicles related to aircraft handling [50]. The demand of airside ground support equipment can be decreased through more efficient operation. Sigler, Wang, Liu, et al. [51] for example optimised routes and schedules of shuttles at the airport to decrease the energy consumption of the airport. Moreover, Zoutendijk and Mitici [52] are optimising the schedules of aircraft towing vehicles at limited energy supply to study the effect of an increased electricity demand of the airport due to the introduction of electric towing vehicles. Indeed, many airports are working towards fully emission-free ground handling, such as Amsterdam Schiphol, where the ambition is to achieve emission-free ground operations by the year 2030 and where, in 2023, 30 new electric ground power units are introduced in order to reach this goal [53].

Finally, future energy demands of green aviation can be considered when estimating the energy demand of airports. Radhakrishnan [54] studied the effect of additional electricity demand due to electric aviation at Stavanger airport, which might double the current energy demand within the next 20 years, and found that it is difficult to transform the airport into a positive energy district only by the introduction of renewable power generation from PV at the airport.

2.3 Energy System Optimisation

In parallel to energy demand estimation, energy system optimisation can be done either for the terminal buildings, the airport operations or a combination of both. Most studies focus on the terminal building energy system. For optimisation, linear programming is mostly used, allowing to find a solution from an objective function and set of constraints, both being linear functions of the decision variables [55]. Mixed-integer linear programming furthermore allows to determine an optimal solution from both integer and non-integer variables. Sequence quadratic programming [56], particle swarm optimisation [57], hybrid fuzzy-stochastic programming [58] and multiple criteria decision making [59] are observed to be alternative options. Moreover, specific tools for the optimisation of energy systems exist and are used, such as the Hybrid Optimisation Model for Electric Renewables developed by the National Renewable Energy Laboratory in the United States [60] or the modular open-source OSeMOSYS framework employed by Prussi, Laveneziana, Misul, et al. [61].

Multiple studies investigated energy system optimisation of airport buildings. Most of them looked into a combination of heating, cooling and power supply [58], [62]–[64], while some of them only considered individual demands such as only power supply [59], [60], only power and cooling supply [56] or only thermal demands [65]. Some of the studies consider in addition to the building energy demands the vehicle demands at the airport [61] or additional demands due to electric aircraft, as was done by Tian, Zhang, and Balta-Ozkan [57], who optimised the full energy system including buildings and mobility demand.

Different conclusions are observed for the building energy system. Jin and Li [58] identified that the optimal share of grid electricity significantly depended on the grid electricity price. Similarly, Zhang, Si, Feng, et al. [56] described how it can be beneficial to use different cooling technologies during night and day based on the natural gas and electricity prices. Renewable energy generation at the airport can reduce the dependence of the grid. In fact, local renewable electricity production technologies often have lower levelised costs than grid electricity, as was found for PV energy by Thiem, Danov, Metzger, et al. [62]. However, for PV it is important to consider the limitations due to glare effects on the pilots at the airport. Next to PV, wind energy can be considered as an alternative local resource for emission-free airports [64]. Rubeis, Nardi, Paoletti, et al. [46] propose the combination of solar and wind power with energy storage technologies for load management and specifically suggest to replicate this approach to other European airports. Natural gas generators could be a non-renewable alternative for levelling electricity supply [57]. For heating and cooling, combined technologies of heating, cooling and/or power are observed to show a high potential [58], [62], [66].

From the different studies, it is observed that through new integrated building cooling, heating and power systems, the costs of the energy system can be decreased or even profitable energy systems might be obtained [56], [62], [63]. However, Zhang, Si, Feng, et al. [56] also noted that the most cost-effective solutions are not necessarily equivalent to the most energy-efficient solution [56], indicating a trade-off to be made between different parameters. In general, these results must be analysed with caution, as the simulations have limitations, due, for example, to the assumptions made, which might lead to good but not necessarily globally optimal results, as is mentioned by Baek, Kim, and Chang [60].

For the ground support equipment, Kirca, McGordon, and Dinh [67] investigate how emissions could be reduced through electrified ground support equipment as well as optimized charging schedules and battery pack sizes. For the future, it is suggested to look into adding vehicle-to-grid operations. Another option for ensuring electricity supply for the ground support equipment is the introduction of hydrogen. Including a hydrogen storage system would allow to power for example the aircraft auxiliary power unit and to increase the overall energy system efficiency, as was seen by Xiang, Cai, Liu, et al. [68]. Furthermore, in combination with PV energy, hydrogen allows to compensate for insufficient availabilities of grid electricity [69].

2.4 Research Gap

Multiple studies already examined the energy demand of airports and introduced methods for optimising their energy systems. However, most of these studies only focused on the demand estimation or the energy system optimisation, while for the latter, most studies did not optimise the full energy system but focused on one part, being for example the heating and cooling system, the buildings or the ground support equipment charging. Furthermore, most of the studies focused on one airport. The research gap identified for this study lied thus in an all-encompassing approach for estimating the airports' energy demand as well as constructing the airports' energy system, by using an approach globally applicable to any airport location with a limited set of pre-defined airport parameters. The model shall allow to analyse the performance of a variety of energy technologies, including possible production methods of sustainable aviation fuels, which has not been explored in the studies discussed.

3 Methodology

The methodology for this thesis comprises both the energy demand estimation as well as the construction of the energy system for the airport. Thus, in this section, first the methods for estimating the energy demand and for obtaining and clustering the optimisation input data are explained in section 3.1 and section 3.2 respectively. Subsequently, section 3.3 describes in detail the energy system superstructure constructed for this study. Finally, section 3.4, section 3.5 and section 3.6 explain the employed optimisation framework, the possible objective functions and the chosen key performance indicators.

3.1 Energy Demand Estimation

The energy demand of an airport consists of several components. Figure 2 shows an example of the yearly energy consumption distribution of Amsterdam Schiphol [70]. It can be observed that the main consumers of energy are the airport terminals and piers. Other significant consumers are buildings such as hotels around the airport, hangars and warehouses, charging and shore power, parking and aircraft handling. The other consumers show only a minor share of 2% or less. This study is limited to the energy consumption at the airport related to passenger and flight handling and thus to the terminal and pier buildings and aircraft handling using ground support equipment (GSE), making up 61% of the total energy consumption at Amsterdam Schiphol airport. The methods for estimating those demands are outlined in the following subsections.



Figure 2: Total Energy Consumption Distribution of Amsterdam Schiphol Airport in 2023 (corrected for primary energy factor in numbers provided) [70]

3.1.1 Terminal and Pier Buildings

The main energy consumption contributors considered are the terminal and pier buildings. An example of the energy consumption distribution of these buildings is shown in Figure 3 [71]. Overall, it can be estimated that 19% of the energy demand is related to air handling, 9% to cooling, 27% to heating, 2% to domestic hot water, and 43% to other electricity consumers, such as lighting or baggage handling. Three types of energy

demands will be defined in this study: heating consisting of building heating and domestic hot water, cooling, and electricity consisting of ventilation and other electricity appliances.



Figure 3: Energy Consumption Distribution Terminals and Piers at Amsterdam Schiphol Airport in 2023 (Note: Terminal 2 also provides climate heating and cooling for Terminal 1, Plaza and Skyport) [71]

Heating and Cooling

In order to estimate the building heating and cooling energy demand, different heat exchanges of the building with its environment as well as its interior are considered. An overview of this is shown in Figure 4. As can be seen, the heat losses from the ventilation system and through the building walls as well as the heat gains from solar irradiance on the building walls and windows, from electric appliances and people in the building are considered to determine the overall heating or cooling demand required to ensure an interior temperature of $T_{int} = 22^{\circ}$ C, the typical value to which public buildings are heated [72].



Figure 4: Illustration of the heat exchanges considered for estimating the building heating, cooling and DHW energy demands

$$\dot{q}_{sh} = \dot{q}_g - \dot{q}_{wa} - \dot{q}_{fa} \tag{1}$$

The different contributions seen in the illustration are also reflected in Equation 1, which determines the overall hourly space heating demand \dot{q}_{sh} from the interior and solar gains \dot{q}_g , the heat lost through the walls \dot{q}_{wa} and the heat lost through fresh air entering the building \dot{q}_{fa} . If \dot{q}_{sh} is negative, no heating but cooling is required to ensure the set-point interior temperature. The model for obtaining the heat and cooling demands is based on a similar building envelope model for residential buildings [34].

$$\dot{q}_g = \dot{q}_{int} + \dot{q}_{irr} = (A_b \cdot \Phi_{ea} + n_{pax} \cdot DA \cdot MET) + A_{wi} \cdot \phi_{wi} \cdot I \tag{2}$$

As shown in Equation 2, the heat gain of the building \dot{q}_g is obtained from two different contributions: the interior gains \dot{q}_{int} and the solar irradiance gains \dot{q}_{irr} . The interior gains are on the one hand gains from power

appliances, which are calculated from the surface area of the building A_b as well as an estimate of $\Phi_{ea} = 3 \text{W/m}^2$ [72] for the heat input power of appliances in similar large-surface areas such as shopping malls or restaurants. On the other hand, there are gains from the building occupiers. They are estimated from the hourly number of passengers in the building n_{pax} , an estimate of DA = 1.2 [72] for the degree of activity for people in similar buildings such as shopping malls and a value for the standard metabolic heat (MET) for a human body of MET = 104 W [73].

As a second contribution, building heat gains are occurring due to solar irradiance on the windows, \dot{q}_{irr} . They are computed from a window solar transmission factor $\phi_{wi} = 0.65$ [72], an estimate that 80% of the wall area of the buildings is composed of windows, and the wall area A_{wa} . The wall area is obtained from an estimate for the building height of 11*m*, based on data, and the building contour length obtained from [74].

$$\dot{q}_{wa} = U_b \cdot (A_b + A_{wa}) \cdot (T_{int} - T_{ext}) \tag{3}$$

Next, the heat lost because of heat exchange through the building walls, \dot{q}_{wa} , is determined using Equation 3. This estimate is based on the thermal transmittance of the building envelope, U_b , the surface area of the building, the wall area A_{wa} as well as the interior set-point temperature and the external temperature, T_{ext} . For the thermal transmittance, a value of $U_{b_{roof}} = 0.8 \text{W/m}^2/\text{K}$ is used for the building roof and of $U_{b_{wall}} = 1.36 \text{W/m}^2/\text{K}$ for the wall area (being composed of 80% windows with $U_{b_{window}} = 1.5 \text{W/m}^2/\text{K}$ and 20% opaque surfaces with $U_{b_{opaque}} = 0.8 \text{W/m}^2/\text{K}$) [72].

$$\dot{q}_a = ACPH \cdot \Psi_{out} \cdot (1 - \xi_{vs}) \cdot \rho_{air} \cdot C_{p_{air}} \cdot (T_{int} - T_{ext}) \tag{4}$$

Finally, the heat lost due to the fresh air entering with the ventilation system, \dot{q}_a , is estimated using Equation 4. The volume flow of fresh air entering the building is calculated by multiplying the required air changes per hour (*ACPH*) of the ventilation system with $\Psi_{out} = 1/3$ of the air going through the ventilation system being replaced with outdoor air, an estimate based on similar ratios of health care public buildings [75]. Furthermore, it is assumed that $\xi_{vs} = 0.73$ [72] of the heat from the replaced air is recovered to the new air entering the building, thus reducing the heating demand. Next to that, an air density of $\rho_{air} = 1.225 \text{kg/m}^3$ and a specific heat capacity at constant pressure of $C_{pair} = 1.006 \text{kJ/kg/K}$ are used for the exchanged air [76].

Domestic Hot Water

The heating demand of domestic hot water is estimated as follows.

$$\dot{q}_{dhw} = \rho_{dhw} \cdot C_{p_{dhw}} \cdot \dot{V}_{dhw} \cdot (T_{hw} - T_{cw}) \tag{5}$$

As can be seen in Equation 5, the heating demand \dot{q}_{dhw} for domestic hot water is estimated from the required hourly volume of hot water \dot{V}_{dhw} , the temperature of the water before being heated up of $T_{cw} = 12.5^{\circ}$ C [77] as well as the target temperature of the water after heating and before delivery of $T_{hw} = 60.0^{\circ}$ C [78]. Furthermore, a density of water of $\rho_{dhw} = 1,000 \text{kg/m}^3$ and a specific heat capacity at constant pressure of $C_{pdhw} = 4.186 \text{kJ/kg/K}$ are employed [79].

$$\dot{V}_{dhw} = n_{pax} \cdot V_{dhw_{pax}} \tag{6}$$

The hourly volume flow of domestic hot water \dot{V}_{dhw} is computed using Equation 6 from the number of passengers as well as an estimate $V_{dhw_{pax}}$ for the required volume of domestic hot water per passenger. For the latter, a combination of the standard demand of domestic hot water per person in a restaurant (15l/d/pers) and in a shopping mall (1.5l/d/pers) is used [72]. As it can be assumed that 50% of all passengers eat in a restaurant at the airport [80], 100% of the value for a shopping mall and 50% of the value for the restaurant are summed up, leading to a domestic hot water demand per person of 9l/d/pers.

Ventilation

To complete the heating, ventilation and cooling system, the electricity demand for the ventilation system is estimated.

$$\dot{e}_{vs} = \dot{V}_{vs} \cdot E_{sp_{vs}} = (ACPH \cdot A_b \cdot \overline{H}_{b_{int}}) \cdot E_{sp_{vs}} \tag{7}$$

The electricity demand of the ventilation system \dot{e}_{vs} is calculated using Equation 7 from the required overall volume flow of the ventilation system \dot{V}_{vs} and an expected average consumption of new ventilation systems of $E_{sp_{vs}} = 0.28 \text{Wh/m}^3$ [81]. The overall volume flow is determined from the building surface area, an estimate of the average inner height of the buildings of $\overline{H}_{b_{int}} = 8.8 \text{m}$, being equal to 80% of the estimate for the overall building height, and the required ACPH. For the latter, a value of 6 is used, the same as the standard for shopping malls (6), slightly higher than the standard for public waiting rooms (4) and slightly lower than for restaurants (8-12) [82], of which the airport can be estimated to be a combination of.

Electricity

The electricity demand of the remaining energy consumers in the terminal and pier buildings is estimated based on the data of Amsterdam Schiphol airport shown in Figure 3 [71]. Different parameters are used in order to quantify the electricity demand and to determine its profile for the different consumers.

	Sizing parameters	Profile determining parameters
Baggage handling/process	Passenger capacity, surface area	Number of flights
ICT	Passenger capacity	Number of flights
Lighting	Surface area	Constant, daylight hours
Pantry/kitchen	Passenger capacity, surface area	Number of flights
Servers MER+SER	Passenger capacity	Constant
Other	Surface area	Constant

Table 1: Parameters defining the electricity demand of an airport

In Table 1, an overview of the different parameters for the different consumers is given. The sizing parameters for the electricity consumption are the yearly passenger capacity of the airport, the surface area or an equally weighted combination of these two. A normalized yearly electricity consumption is subsequently determined, in order to determine the electricity demand for any airport based on these two parameters.

In order to obtain an hourly profile of electricity demand, different parameters as shown in Table 1 are again used. These are either the number of flights, the daylight hours or none of the two, depending on which parameter is assumed to be most relevant. The normalized profile of these parameters is then multiplied with the yearly electricity consumption of every consumer to get a yearly profile.

3.1.2 Ground Support Equipment (GSE)

The second energy consumption contributor considered in this study is the GSE on the airport. For this, the energy demand is estimated based on data obtained from KLM for their GSE employed at Amsterdam Schiphol airport. The data obtained includes all equipment trips in the airport required to serve the KLM and SkyTeam member flights [83]. The daily sums of these trips for the different types of vehicles considered are shown in Figure 5.



Figure 5: Distances driven by GSE of KLM at Amsterdam Schiphol airport in August 2024 (computed from KLM GSE trip data [83])

In order to obtain the energy demand for the GSE, the average energy demand of GSE per flight is computed. Using the flight profile for a full year, the daily charging demand can then be determined.

$$d_{f_d}^{GSE_v} = \frac{\sum_{i=1}^{tr_d} d_i^{GSE_v}}{N_{f_d}} \tag{8}$$

First, the distance driven by each GSE vehicle type v per flight f on each day $d d_{f_d}^{GSE_v}$ in the period for which the data was obtained can be calculated using Equation 8. This is done by summing the distances $d_i^{GSE_v}$ of every individual GSE trip i on the respective day and dividing them by the number of flights N_{f_d} on that day. Only the flights of KLM and other SkyTeam member airlines are considered, as the GSE for which the data is available only operated these airlines.

$$\bar{d}_{f}^{GSE_{v}} = \frac{\sum_{d=1}^{n_{d}} d_{f_{d}}^{GSE_{v}}}{n_{d}}$$
(9)

Next, the average GSE distance per flight $\overline{d}_f^{GSE_v}$ for every GSE type is calculated using Equation 9 by determining the average distance of all days for which data was obtained.

$$\overline{E}_{c_f}^{GSE_v} = P_r^{GSE_v} \cdot \frac{LF^{GSE_v}}{\eta^{GSE_v}} \cdot t_{dr_f}^{GSE_v} , \qquad where \ t_{dr_f}^{GSE_v} = \frac{\overline{d}_f^{GSE_v}}{v_{med}^{GSE_v}}$$
(10)

Subsequently, the energy consumption for every GSE type per flight $\overline{E}_{c_f}^{GSE_v}$ can be determined using Equation 10. For this, the rated power $P_r^{GSE_v}$, the load factor LF^{GSE_v} and the efficiency η^{GSE_v} for each vehicle type is needed together with the time $t_{dr_f}^{GSE_v}$ the type of GSE is driving. The latter can be obtained using the average GSE distance per flight and an estimate $v_{med}^{GSE_v}$ for the median velocity.

Data for the GSE rated power has been drawn from a GSE manufacturer [84]. For most vehicles, the average rated power from the vehicle portfolio has been determined. For the dispenser and lower deck loader, no data is available and thus the average of all GSE has been used. The load factor has been determined using the average load factor for all activities per vehicle category, which is LF = 4 for vehicles of type B and C [67]. Finally, for the efficiency a value of $\eta = 0.75$ is assumed for electric GSE [85].

Vehicle type	Vehicle category	Rated power [kW]
Baggage / Cargo tractor	В	49.5
Beltloader	С	49.8
Dispenser	В	76.4 (GSE average estimate)
Lavatory truck	В	108.4
Lower deck loader	С	76.4 (GSE average estimate)
Transporter	В	49.5
Potable water truck	В	108.4
Passenger stairs	С	92.9

Table 2: Vehicle category [67] and rated power of GSE for which data is available [84]

$$\overline{E}_{c_d}^{GSE} = \sum_{f=1}^{n_{f_d}} \sum_{v=1}^{GSE} \overline{E}_{c_f}^{GSE_v}$$
(11)

In order to obtain the average energy consumption $\overline{E}_{c_d}^{GSE}$ per day for all GSE, the average estimated consumptions of all GSE types for all flights on a day can be summed as shown in Equation 11, with n_f being the number of flights on that day. An extra 40% of energy consumption is added for missing types of vehicles such as the aircraft tractors, which is estimated based on the overall aircraft handling energy consumption at Amsterdam Schiphol airport [70]. Accuracy of the model can be improved in the future by using complete data including all types of GSE vehicles instead of an estimate for the missing vehicles' energy consumption.

$$\dot{e}_{ch_t}^{GSE} = \frac{\overline{E}_{c_d}^{GSE}}{t_{ch_d}} \tag{12}$$

Finally, the required electricity for charging needs to be determined. As the GSE need to operate during the whole day, it is assumed that they are only charging during nightime from 00:00 to 05:00. In consequence, as can be seen in Equation 12, the electricity $\dot{E}_{ch_t}^{GSE}$ required for charging in each of these five hours can be determined by dividing the average energy consumption per day by the possible charging duration per day t_{ch_d} of five hours.

3.2 Input Data Used for Optimisation

To compute the energy demand, different input data are required, as for example weather or flight data. Furthermore, prices for resources are required for later optimising the energy system. It is decided to optimise the energy system for hourly intervals, for all 8,760 hours of the year. Thus, hourly input data for the airport and weather characteristics is required and obtained as described in section 3.2.1. Next, this hourly data is clustered in typical days as described in section 3.2.2, in order to reduce the complexity of the optimisation problem.

3.2.1 Airport, Weather and Resource Cost Profiles

The energy system optimization is performed for an hourly profile. This entails that, for the data varying over the year, an hourly profile is required. This will be the case for performance data of the airport including the number of flights and passengers, for weather data including the position of the sun, as well as for resource prices data.

For the airport performance data, the hourly number of flights can be obtained from the OpenSky API [86]. Using the time instances of all departures and arrivals at the airport in 2023, the number of flight movements (including both departures and arrivals) can be computed for every hour of the year.

It is more difficult to compute the hourly number of passengers at an airport. Most of the airports only publicly provide the monthly passenger statistics, such as Amsterdam Schiphol airport [87]. In consequence, the hourly number of passengers has to be estimated from the hourly number of flights and the monthly passenger profile. The method used is shown in Figure 6. First, for every month the scaling factor between the total number of passengers and flights is computed. Subsequently, average scaling factors in between all months are computed and linearly connected to obtain the variation of the scaling factor at every hour in the month. Finally, in order to obtain the correct overall number of passengers for every month, a parabolic function is added to the scaling factor, adjusting the number of passengers to the monthly passenger profile in an iterative procedure.



Figure 6: Method for obtaining hourly passenger profile from hourly flight profile and monthly passenger profile

Hourly profiles for the weather data can be obtained from Open-Meteo [88]. The historical API database includes an extensive set of weather data for any location on the Earth, including the solar radiation, the wind speed at different heights, the ambient temperature as well as the daylight duration as required for this thesis. Next to that, the SolTrack Python package [89] can be used to obtain the precise solar position at any location and time in the year, to determine the exact potential of PV power.

For the electricity prices, the Energy-Charts API [90] can be employed. This API database allows to obtain the historical hourly variation of the electricity prices on the different European electricity spot markets. Furthermore, it also allows to obtain the share of renewable electricity at every hour in the year for the different spot markets. Together with the yearly electricity emissions in every country [91], this also allows to obtain the hourly CO_2 emissions of the electricity from the grid, as can be seen in Equation 13.

$$\left(\frac{G_{el}}{E_{el}}\right)_h = \frac{G_{el_y}}{E_{el_y}^{nr}} \cdot \epsilon_{el_h}^{nr} \tag{13}$$

Where G_{el_y} are the overall CO₂ emissions in a year, $E_{nr_{el_y}}$ is the yearly amount of non-renewable electricity and $\epsilon_{el_p}^{nr}$ is the share of non-renewable electricity in that hour.

3.2.2 Clustering Input Data in Typical Days

Optimising the energy system for the hourly profiles obtained for the entire year would lead to high model complexity with in total 8,760 data points. In consequence, it is important to reduce the data size [34] to less data points, which can for example be done by clustering the data in typical days. Using this method, every day in the year is assigned a typical day. The few typical days chosen to represent all days of the year are characteristic days of the year, for example in terms of their weather, and allow to give an approximate image of the entire year, by reducing the size of the problem to less repeating data points. For the final model results, such as the yearly operating costs, the results obtained for the distinct typical days are multiplied by the number of occurrence of every typical day in the full year.

Different methods exist for clustering all days of the year into typical days. From the most common aggregation methods, the k-medoids method, which is minimising the sum of the deviations of all clusters, displays overall the best performance [92]. The code used for clustering has been adapted from the code employed in the REHO energy optimisation framework [93]. For clustering, a number of 12 typical days are chosen. Two extreme hours are added, which complement the data by unrepresented hours with extreme values such as for example solar radiation. The typical days have been clustered based on three weather parameters: solar radiation, wind speed and ambient temperature. Next to those, the electricity price is defining the results and is variable over the year. However, it is assumed that it is dependent on the wind speed and solar radiation and thus its variation is already partly represented by including those two variables.

Having chosen the typical days using clustering methods, the remaining hourly data, both for the demand as well as for other relevant parameters such as the electricity price or grid electricity emissions, is added to the weather data for the typical days. It is used from the same days as the 12 days chosen using the typical day clustering, which is possible as all of the data used is from 2023. This clustered data is the input data used for the optimisation problem, for which energy system structures are computed.

3.3 Energy System Superstructure

The energy system superstructure, as shown in Figure 7, is implemented in order to serve the energy demands from section 3.1 for the typical days determined in section 3.2.2 constructed with the hourly data profiles from section 3.2.1. Next to the pre-defined demands, it consists of the resources and technologies, which will be described in the following subsections. An overview of the costs and impacts defined for all technologies described in this subsection can be found in section 3.3.6.



Figure 7: Overall superstructure of the energy technologies implemented with simplified resource flows

3.3.1 Resources

Different resources can be imported to the energy system. First of all, the grid electricity is included in the superstructure with variable prices and variables CO_2 emissions, which are computed as described in section 3.2.1 from Energy-Charts data [90].

The second main resource is natural gas. For its price, the average value of the Dutch TTF Natural Gas price for 2023 of $41.28 \notin MWh$ [94] is employed for the full year and no hourly profile is applied. The reason for this is that the natural gas price is not related to varying weather such as the electricity price but depends also on geopolitical factors such as the Russian war against Ukraine [95], which are difficult to predict.

Water, mainly required for the production of hydrogen and synthetic fuels, is for this study assumed to be available at a zero price. It is considered that it could be harvested from the rain on the large airport (buildings) surface or could be obtained from treated wastewater. Additional water would however require the water price of the grid.

The air rich in CO_2 is assumed to be available at zero price from close-by industry. For the case of Amsterdam, it could come from steel industry, which is for example present in Ijmuiden [96]. Also other emission-intensive industries close to the airport would be possible sources of CO_2 being used for co-electrolysis and synthetic fuel production.

Finally, the availability and the cost of biomass are determined based on a tool for biomass chains in the different European countries [97]. The biomass is however not available at a constant price, but different types of biomass are available at different costs, ranging in the Netherlands for 2020 from approximately 1,200kt available at a cost of slightly less than $5 \notin /t$ up to more than 3,600kt of biomass available when paying costs up to $120 \notin /t$. As not a single value can be deduced from the biomass cost profile, varying values will be looked at when analysing the production of synthetic fuels.

3.3.2 Electricity Generation and Storage Technologies

In this subsection, the working principle of the technologies used for local electricity generation and storage is explained. PV and mini wind turbines are technologies implemented for the production of renewable electricity at the airport. It must be added that also the combined cycle can locally produce electricity, which is described later. Furthermore, batteries are installed for storing electricity. Next to that, a model for a hydrogen storage system with electrolysers and fuel cells is included in the superstructure, being described separately.

\mathbf{PV}

Local generation of electricity can be performed through solar cells placed on the airport area. Even when limited in locations for placement due to glare and radio effects, there are still sufficient locations on the areas around the airport where the installation of PV is possible [98]. For the PV, a model taking into account the solar irradiation and the position of the sun is used, which calculates the energy produced \dot{e}_t^{PV} based on the following Equation 14 [99].

$$\dot{e}_t^{PV} = \eta_{el}^{PV} \cdot k_{gen}^{PV} \cdot k_{rad,t}^{PV} \cdot \left[1 - k_T^{PV} \cdot \left(T_{cell,t}^{PV} - T_{STC}\right)\right] \cdot GI_t \cdot A^{PV} , \qquad where \ t = time \ in \ the \ year$$
(14)

where $\eta_{el}^{PV} = 0.178$ [99] is the electrical efficiency of the PV module, $k_{gen}^{PV} = 0.95$ [99] is the efficiency factor of the generator electrical conversion, $k_{rad,t}^{PV}$ is a factor taking into account the radiation intensity based on the incident of solar radiation, $k_T^{PV} = 0.004/\text{K}$ [99] is the temperature reduction factor, $T_{cell,t}^{PV}$ and $T_{STC} = 25^{\circ}\text{C}$ [99] are the actual and standard test condition cell temperature, GI_t is the global incident radiation on the surface of the PV and A^{PV} is the area of the PV.

The costs of the PV are determined based on an estimate of the current cost for commercial high area PV installation [100].

Mini Wind Turbines

The second local production method for electricity are mini wind turbines. They are small wind turbines suitable to the airport environment, which imposes strict physical and electromagnetic or radio easements on their placement [101] not allowing the installation of conventional large wind turbines close to the runways. The energy produced from such small scale wind turbines \dot{E}_p^{wt} is computed from Equation 15 [102].

$$\dot{e}_t^{wt} = \frac{1}{2} \cdot A_{sw} \cdot \rho_{air} \cdot \eta^{wt} \cdot v_t^3 , \qquad where \ t = time \ in \ the \ year \tag{15}$$

An efficiency of $\eta_{wt} = 0.2$ is assumed for the wind turbines for an estimated average wind speed of $\overline{v_t} = 4$ m/s [102]. The swept area A_{sw} is calculated for an average rotor radius of 9.8m for horizontal axis mini wind turbines, which is used for assuming the cost for each installed wind turbine [103].

Battery

A battery model is implemented in the superstructure as a method for short-term, especially daily, electricity storage. Nickel-Cobalt-Aluminium Lithium-ion batteries are chosen as a technology due to their relatively low current installation cost. They are observed to have a round trip efficiency of 95% and a daily self-discharge rate of 0.2% [104].

3.3.3 Heating and Cooling

Different technologies are implemented for heating and cooling of the airport terminal buildings. These include conventional heating technologies running on natural gas, such as boilers of combined cycles, heat pumps and chillers for electricity-based heating and cooling as well as a CO_2 network as a combined heating and cooling solution. The working principle of the models of all these technologies is described in this subsection.

Boiler Natural Gas

The natural gas boiler is implemented as a simple model [105] for a centralised boiler with a thermal efficiency of 95%. It has a minor consumption of 0.02kW of electricity for every 1kW of useful heat output. As investment costs, costs for similar boilers used in district heating systems are assumed due to the significant scale of the airport's surface area [100]. Due to the scope-1 CO₂eq emissions from burning the natural gas, an impact value is assigned to the boiler [106].

Combined Cycle Natural Gas

As an alternative to the natural gas boiler, a combined cycle is introduced in the superstructure both generating heat and electricity from burning natural gas. The implemented technology model has a thermal efficiency of 40% and an electrical efficiency of 49% [105]. For the costs, investment costs values for conventional combined cycle gas turbines are used [100]. Again, scope-1 CO₂eq emissions have to be added to the technology model by assigning an impact value [106].

Heat Pump

For the heat pump, a simple model is implemented taking heat from water available in rivers, lakes or the sea. The model is based on the calculation of the coefficient of performance COP_{real} displayed in Equation 16 [65].

$$COP_{real} = \eta_{II} \cdot COP_{th} = \eta_{II} \cdot \frac{T_{cond} + 273.15}{T_{cond} - T_{evan} + \Delta T_{evan}^{min}}$$
(16)

For the assumed supply temperature which is equal to the exit temperature at the condenser $T_{cond} = 60^{\circ}$ C, a second law efficiency of $\eta_{II} = 0.423$ can be assumed [34]. A minimum temperature difference of $\Delta T_{evap}^{min} = 2^{\circ}$ C can be used at the evaporator [65]. The heat source outlet temperature being the temperature at the exit of the evaporator $T_{evap} = 1^{\circ}$ C — 2°C minus the subtracted superheating temperature difference [65] is inspired from similar values for rivers, lakes and seawater [107].

The investment cost is estimated based on the costs for similar types of heat pumps and their maximal electric power [108].

Chiller

The chiller has a model very similar to the heat pump, as it basically is a heat pump running in reverse mode, releasing the heat to the environment. Hence, Equation 16 [65] still applies for calculating the coefficient of performance COP_{real} of the chiller. Still using the same condenser exit temperature of $T_{cond} = 60^{\circ}$ C, but now with a minimum temperature difference of $T_{cond}^{min} = 1.5^{\circ}$ C at the condenser [65], a second law efficiency of $\eta_{II} = 0.436$ can be applied [34]. Next to that, a minimum temperature difference of $T_{evap}^{min} = 2^{\circ}$ C at the evaporator is now applied [65]. The evaporators now have an exit temperature of $T_{evap} = 6^{\circ}$ C, which is based on the heat source outlet temperatures of district cooling systems [107].

For the investment cost of the chiller, the same cost estimate as used for the heat pump is employed [108].

CO₂ Network

Next to the conventional technologies for heating and cooling, a CO_2 network is implemented in the superstructure as a more innovative solution, combining both heating and cooling services. The CO_2 network [108] consists of several decentralized units which can serve the users' heating and cooling demands, of pipes delivering the heat transfer fluid being CO_2 , as well as of a central plant required for balancing the CO_2 network and for exchanging heat with the environment. The latter, both being a heat source and sink, could for example be a river, lake or sea such as for the heat pumps previously discussed. A simple overview of the CO_2 network units implemented in the superstructure can be seen in Figure 8.



Figure 8: Overall structure of the CO₂ network (inspired from [109])

The CO_2 network uses CO_2 as a refrigerant transporting heat from the central plant throughout the network [109]. It is based on the latent heat of evaporation and condensation of CO_2 . The working principle is described in [110]. When running in heating mode, the CO_2 is arriving in liquid form at the central plant where it is evaporated and compressed, in order to leave again as CO_2 vapour to the energy user units. In the cooling mode, the CO_2 arrives as vapour at the central plant where it is condensed with a pump and leaves again as liquid CO_2 . Two pipes connect the central plant to the decentralized heating and cooling units, one for liquid and one for vaporized CO_2 . The heating units for space heating and domestic hot water consist of a heat pump, which takes the heat from the vaporized CO_2 and makes it condense. The air conditioning unit uses a heat exchanger in which the CO_2 is evaporated with heat exchanged from the building. As the pipe with liquid CO_2 has a slightly higher pressure than the pipe with vaporized CO_2 , no heat pump is needed here. For refrigeration cooling demands however, a heat pump again is needed for ensuring the cooling demands by vaporizing the CO_2 .

The costs for the different units, namely the heat exchanger and heat pump at the central plant, the CO_2 pipes connecting the whole network and the distributed heat pumps (heating, refrigeration) and heat exchangers (air conditioning) around the building, are used from [108]. Furthermore, building data from OpenStreetMap [74] is used to estimate the required length for the pipes, which need to supply every part of the building with two pipes.

3.3.4 Hydrogen Technologies

Hydrogen is implemented as an electricity storage medium, no demand of hydrogen is included in the superstructure. In order to serve as an electricity storage medium, water needs to be converted with electricity to hydrogen, stored and then converted back from hydrogen, while producing electricity. Three units are implemented for electrolysis from water and electricity to oxygen and storable hydrogen: A solid oxide electrolyser cell, a proton-exchange membrane and an alkaline electrolyser cell. For storage, a pressurized compressed hydrogen tank is included. Finally, generation of electricity from the stored hydrogen can be done using a solid oxide fuel cell. An overview of the implemented technologies can be seen in Figure 9.



Figure 9: Overall structure of the implemented hydrogen storage system

Electrolysers

Three types of electrolyser are implemented for the production of hydrogen. Their models are based on the conversion efficiency, which is next to the cost the main difference of the different technologies. The models all consume water and electricity and produce oxygen, hydrogen and heat, some of them however also having other side products visible in Figure 9. The model for the alkaline electrolyser has the lowest efficiency of 73% [111] but appears to be the cheapest of the implemented electrolysis technologies [100]. The Proton-exchange membrane model has a medium efficiency of 82% [112] and also a significantly higher cost [100]. Finally, the solid oxide electrolyser cell has both the highest efficiency of 89% [113] but also a high cost [114].

Fuel Cell

The model of the solid oxide fuel cell is the reverse option of solid oxide electrolyser cells, both being two types of solid oxide cells. The models of these two technologies have been developed by colleagues at the IPESE lab [115]. The model of the solid oxide fuel cell displays an efficiency of 57%. Costs for this technology are obtained from [116].

Pressurized Hydrogen Storage

Hydrogen storage is ensured in the superstructure using a pressurized hydrogen storage tank. Out of different overground hydrogen technologies, such as also liquid cryogenic hydrogen storage or metal hydrides, pressurised hydrogen tanks have the lowest investment costs [100]. Only underground hydrogen storage has a slightly lower investment cost [100]. Next to the battery, hydrogen storage is included in the superstructure as a long-term seasonal energy storage method. The implemented pressurised hydrogen storage has a storage efficiency of 88% [100].

3.3.5 Synthetic Fuel Production

Two pathways [117] are implemented for the production of synthetic fuels. The first option is producing synthetic fuels from captured CO_2 using co-electrolysis. The second option employs biomass as a resource to produce synthetic fuels using biogasification. They subsequently both employ a Fischer-Tropsch synthesis for transforming the syngas into synthetic fuels. Finally, a separation unit allows to obtain the product fuels. An overview of the different pathways can be seen in Figure 10, with the individual technologies being described more in detail as follows. The costs of the different technologies are drawn from [100], [118]–[120].



Figure 10: Overall structure of the implemented technology models for the production of synthetic fuels

Carbon Capturing

Different technologies exist for carbon capturing at different stages of development, which are mainly the following four: pre-combustion carbon capture, post-combustion carbon capture, oxyfuel combustion and direct air capture [121]. Direct air capture using adsorption has been developed significantly over the last year, but further technology developments are still needed in order to be deployed at large scale [122]. Overall, absorption-based technologies are currently still cheaper [100] and are thus used in this study. They can be used as post-combustion technologies for high-carbon industries and are the basis for the model implemented [123], [124]. This monoethanolamide carbon capture can for example be installed to capture emissions of the cement or steel industry [117] and could for example use fumes from the steel production site of Tata Steel in Ijmuiden [96]. For the future, it might be interesting to look into different technologies of carbon capture, as also direct-air capture from the ventilation system.

CO₂ Co-Electrolysis

The captured CO_2 can be used in the co-electrolysis model in the superstructure to produce synthesis gas [117]. In a first step, co-electrolysis uses CO_2 and water in the form of steam and the addition of heat to produce a synthesis product gas [125]. From this gas, water contents need to be separated in a subsequent water separation step. Finally, the optimal ratio of H₂ and CO needs to be obtained using a water-gas-shift reaction, which allows the synthesis gas to be forwarded to the Fischer-Tropsch synthesis [117].

Biogasification

The model of the biogasification pathway [119] starts with drying the biomass and torrefaction, which allows to upgrade the biomass product, to produce high-quality solid biomass products and thus in the end to improve the usage of the biomass in the subsequent gasification step [126]. Gasification subsequently allows to transform the biomass at very high temperatures into syngas, which is a composition of hydrogen, carbon monoxide, methane and carbon dioxide. The syngas can then be treated using cold gas cleaning, in order to remove for example tar, metal or sulfur impurities. Furthermore, its composition can be changed using a water-gas shift reaction for obtaining the optimal ratio of H_2 and CO for the Fischer Tropsch synthesis. The solid carbon produced from biogasification and other side products can be burnt and the heat can be reused within the biogasification process [125].

Fischer-Tropsch

The synthesis gas obtained from both co-electrolysis and biogasification needs to be further processed in order to obtain Fischer-Tropsch fuels, which are a mixture of straight-chain hydrocarbons similar to semi-refined crude oil [119]. They are obtained from the syngas using a catalytic non-selective exothermal reaction [119]. The crude Fischer-Tropsch fuels obtained still need to be upgraded in order to obtain the desired fuel products, such as for example kerosene, jet fuels or diesel [119]. Side product gases which are obtained throughout the process can be burnt to produce heat for the whole process, similar to what is done with the solid biomass obtained from gasification.

Synthetic Fuel Separation

Finally, the synthetic fuels need to be processed in refineries, where the Fischer-Tropsch crude fuel can be separated for example in sustainable aviation fuel. Average values for the investment costs of syncrude refineries have been used in this study [120]. While the heat streams have been neglected for the synthetic fuel separation unit, a simple model is constructed based on the possible output fuel composition estimated in the models of the AIDRES project [117], which is shown in Table 3. It can be seen that the share of kerosene and thus sustainable aviation fuel is quite low for this type of refinery. Refineries capable of generating more than 50% of kerosene are possible [127], their model implementation is however out of the scope of this study. It is recommended to look into this topic in future projects.

In this study, a simple estimation is made for the possible profit from synthetic fuel production based on the prices without taxes for the different fuel types in the Netherlands [128], [129]. For the light distillate fuels, the heavy reformates and isomerates, the price of gasoline is assumed, as they can often be blended with gasoline, as observable for typical refineries [130]. The prices assumed can also be found in Table 3.

Table 3: Output fuels from synthetic fuel production with estimated fuel share [117] and estimated prices [128], [129]

	Assumed fuel share	Assumed
Fuel type	in output synthetic fuel	prices
	[%]	[EUR/l]
Gasoline	11	0.81
Light distillate fuel	3	0.81
Gas oil	69	0.91
Kerosene	1	0.55
Heavy reformate	6	0.81
Isomerate	8	0.81
Fuel oil	2	0.64

3.3.6 Costs and Environmental Impact of Implemented Technologies

For all technologies explained in the previous subsections, a cost and an environmental impact are defined. The cost is composed of an annual investment cost (Cinv1), an annualised specific investment cost per reference unit size (Cinv2) and a variable operating cost per reference unit size (Cop2). The investment costs are in general annualised for an interest rate of 6% and a lifetime of 25 years. The impact is defined as the variable scope-2 GWP per reference unit size (Impact2). Table 4 gives an overview of the costs and the environmental impact for all technologies, which are compiled on the basis of the costs and impacts collected from multiple references.

3.4 OSMOSE Optimisation Framework

The energy system is optimised using the OSMOSE energy system integration framework which has been developed at the Industrial Process and Energy Systems Engineering (IPESE, formerly LENI) group of EPFL [35]. The mixed integer linear programming problem, the technology, resource and demand units as well as the optimisation framework are defined using the Lua programming language. The full energy system is modelled as a mathematical optimisation problem using AMPL and is solved using CPLEX. The optimisation framework ensures resource balance, process integration and correct sizing of the units [131]. Process integration is achieved using a heat cascade approach, allowing to obtain the minimum energy requirement [132].

The optimisation framework of OSMOSE [35] including the sets, the method for defining a unit and the main constraints will be explained based on and adapted from [131]. Several sets are employed in the optimisation, with the main ones listed as follows.

- U: set of all units defining the technology models in the superstructure, with one model possibly containing several units
- L: set of all layers connecting the units in the superstructure (for example electricity, natural gas or water)
- **P**: set of all periods occurring recurrently over the year, with one period being equivalent to a typical day in this study

		$\mathbf{D}_{c}\mathbf{f}_{c}$						Dafamana
Technology		rence	Reference stream	$\begin{array}{c} \mathbf{Cinv1} \\ \left[\mathbf{\widehat{E}}/\mathbf{y}\right] \end{array}$	$\operatorname{Cinv2}\left(C_{u}^{inv,2} ight)$ $\left[\mathbf{\in}/\operatorname{ref}/\mathbf{y} ight]$	$\operatorname{Cop2}\left(C_{u}^{op,1} ight) \ \left[10^{-3} \mathbf{\in}/\mathrm{ref} ight]$	Impact2 $(G_u^{op,1})$ [kgCO ₂ eq/ref]	kererence costs and
		unit						impact
PV	_	m^2	solar cell area	0	7.09	0	0	[100]
Mini wind turbine	Je	ı	number wind turbines	0	3,396.92	0	0	[103]
Battery		kWh	electricity stored	0	31.49	0	0	[104]
Boiler natural gas	IS	kW	heat produced	0	11.92	0.44	0.21	[100], [105], [106]
Combined cycle natural gas	natural gas	kW	electricity produced	0	63.02	2.31	0.39	[100], [105], [106]
Heat pump		kW	electricity used	444.33	97.00	0	0	[108]
Chiller		kW	electricity used	444.33	97.00	0	0	[108]
	Heat pump	kW	electricity used	444.33	97.00	0	0	[108]
CO ₂ network	Heat exchanger	kW	heat exchanged	15.89	4.14	0	0	[108]
	Piping	m	length pipes and excavation	133.77	0	0	0	[108]
Alkaline electrolyzer	yzer	kW	electricity required	0	135.37	0	0	[100]
Proton-exchange membrane	membrane	kW	electricity required	0	173.68	0	0	[100]
Solid oxyde electrolyzer cell	rolyzer cell	kW	electricity required	0	171.48	0	0	[114]
Solid oxyde fuel cell	cell	kW	electricity produced	0	188.69	0	0	[116]
Pressurized hydrogen storage	ogen storage	kWh	H_2 stored	0	0.47	09.0	0	[100]
Carbon capturing	60	$\rm kg/h$	CO ₂ captured	0	$\sim 5,518.80$	0	0	[100]
Co-electrolysis	Co-electrolyser	$\rm kg/h$	CO_2 entering	0	731.31	0	0	[118]
	Water-gas shift	m kg/h	syngas (H_2+CO) entering	380, 399.42	199.85	0	0	[119]
	Dry pyrolysis	m kg/h	biomass entering	34,760.98	102.36	0	0	[119]
	Gasification	m kg/h	torrefied biomass entering	10,841.05	15.11	0	0	[119]
Biogasification	Cold gas cleaning	m kg/h	syngas (H_2+CO) entering	12,591.06	4.67	0	0	[119]
	Water-gas shift	kg/h	syngas (H_2+CO) entering	380, 399. 42	199.85	0	0	[119]
	Solid carbon burn.	m kg/h	solid carbon entering	195.20	0.01	0	0	[119]
Fischer-Tropsch Synthesis	Synthesis	m kg/h	Fischer Tropsch fuel leaving	0	463.43	0	0	[133]
Synthetic fuel separation	paration	$\rm kg/h$	Fischer Tropsch fuel entering	0	311.73	0	0	[120]

Table 4: Overview of costs and environmental impacts defined for different technologies in energy system after annualisation

- T(p): set of all time instances in period p with distinct input data, with one time interval being one hour in this study
- TS(p,t): set of all streams at time instance t in period p, including a number of n_s resource streams (with the resource stream being associated to a layer), n_{ut} utility streams and n_{pr} process streams (with the latter two being part of the heat cascade)

Every technology unit u in U is defined with a set of state variables. First, all units have a minimum load f_{min_u} and a maximum load f_{max_u} , with f = 1 being the reference load of the unit. All units have a set of streams s going in and out of the unit. The resource streams are defined using the layer l they are belonging to as well as the quantity going in or out of the unit. The heat streams are defined using the enthalpy they exchange, the inlet and outlet temperature and the minimum temperature difference to the supplying or receiving heat streams.

Moreover, the units have investment and operating costs as well as a GWP assigned to them. The following cost parameters are being used for every unit u.

- $C_u^{inv,1}$ [€/y]: annualized fixed investment costs
- $C_u^{inv,2}$ [€/ref/y]: annualized specific investment costs per reference load
- $C_u^{op,1}$ [\notin /ref]: variable operating costs per reference load
- $G_{u}^{op,1}$ [kgCO₂eq/ref]: variable GWP per reference load

Multiple decision variables are used for scaling the different units as well as their connections. The most important decision variables, which are used in the main constraints, are listed as follows.

- f: load of the the different units, which is defined for each time as well as for the maximum load installed
- y: binary value defining presence of the unit in the energy system configuration
- rs: size of resource stream exchanged between two units
- q: heat exchanged by a utility or process unit within the heat cascade
- hc: heat cascaded at different temperature intervals in the heat cascade

The sets and decision variables together with the state variables defined in all technology models are used in a diverse set of constraints. The main constraints employed for optimising the energy system are the following (taken and adapted from [131]). Equation 17 is used for dimensioning of the technology. It is complemented with Equation 18 and Equation 19 constraining unit size and presence.

$$F_{min_u} \cdot y_{u,p,t} \le f_{u,p,t} \le F_{max_u} \cdot y_{u,p,t} \qquad \forall t \in \mathbf{T}(p), \quad \forall p \in \mathbf{T}, \quad \forall u \in \mathbf{U}$$
(17)

$$f_{u,p,t} \le f_u \qquad \forall t \in \mathbf{T}(p), \quad \forall p \in \mathbf{P}, \quad \forall u \in \mathbf{U}$$
 (18)

$$y_{u,p,t} \le y_u \qquad \forall t \in \mathbf{T}(p), \quad \forall p \in \mathbf{P}, \quad \forall u \in \mathbf{U}$$

$$\tag{19}$$

At every instance of time t within period p, the load $f_{u,p,t}$ of the unit u must be in the range between its minimum load F_{min_u} and its maximum load F_{max_u} . If the binary value $y_{u,p,t}$ defining if the unit is present is equal to zero, the load of the unit is also constrained to zero. Furthermore, at any instance of time, the load $f_{u,p,t}$ must be smaller than the maximum load f_u of the optimised configuration. The same applies to the presence of the unit y_u in the optimised configuration.

Furthermore, Equation 20 ensures the balance of the resources, which can for example be electricity (\dot{e}) or natural gas (\dot{h}) .

$$\sum_{s=1}^{n_s(l)} rs_{in_{s,p,t}} = \sum_{s=1}^{n_s(l)} rs_{out_{s,p,t}} \qquad \forall t \in \mathbf{T}(p), \quad \forall p \in \mathbf{P}, \quad \forall l \in \mathbf{L}$$
(20)

At every time t in period p, the resources rs going in must be equal to the resources going out of the system. This balance is ensured by summing up all streams s within a layer l.

Next to the resource streams, also the balance of the heat streams needs to be ensured, which is done using the heat cascade constraint in Equation 21 as well as complementing the constraints shown in Equation 22.

$$\sum_{w=1}^{n_{ut}} f_{w,p,t} q_{ut_{w,p,t,r}} + \sum_{i=1}^{n_{pr}} q_{pr_{i,p,t,r}} + hc_{p,t,r+1} - hc_{p,t,r} = 0 \qquad \forall r = 1, ..., n_{hc}, \quad \forall t \in \mathbf{T}(p), \quad \forall p \in \mathbf{P}$$
(21)

$$hc_{p,t,1} = 0$$
; $hc_{p,t,n_r+1} = 0$; $hc_{p,t,r} \ge 0$ $\forall r = 1, ..., n_{hc}, \forall t \in T(p), \forall p \in P$ (22)

The heat cascade constraint ensures at any instance of time t in period p the balance of heat for all n_r temperature intervals r. Within each interval r, the heating or cooling flow $q_{ut_{w,p,t,r}}$ from all the utility units w with load $f_{w,p,t}$ is added to the heat exchanged by the streams $q_{pr_{i,p,t,r}}$ of all processes i, including the fixed heat demands. The balance is closed with heat $hc_{p,t,r+1}$ which is cascaded from higher temperature intervals r+1 and heat $hc_{p,t,r}$ from lower intervals r. Moreover, it is ensured that no heat can be cascaded for the lowest and highest temperature interval and that any heat cascaded must be larger than or equal to zero.

When including storage units in the optimisation, additional sets and constraints are used (reproduced from the constraints implemented in OSMOSE [35]). The following adapted sets are needed for units and time instances.

- U_s : set of all storage units implemented in the superstructure
- T_r : set of all time instances from correctly sorted and repeating periods during the full year (one typical day can occur multiple times in a year, meaning that every time instance occurring once in T can occur for multiple days in T_r)

Furthermore, the subsequent additional decision variables are needed for an optimisation including storage units.

- es_{level} : energy storage level of storage unit at the end of time instance
- $es_{in/out}$: in- and outgoing energy to and from the storage unit

The main constraint for implementing long-term storage units in OSMOSE is shown in Equation 23.

$$es_{level_{u_s,t_r}} = \begin{cases} \sigma_{u_s} \cdot es_{level_{u_s,-1}} + es_{in_{u_s,t}} - es_{out_{u_s,t}} & \text{if } t_r = 0\\ \sigma_{u_s} \cdot es_{level_{u_s,t_r-1}} + es_{in_{u_s,t}} - es_{out_{u_s,t}} & \text{if } t_r \neq 0 \end{cases} \quad \forall t_r \in \mathbf{T}_r, \quad \forall u_s \in \mathbf{U}_s \tag{23}$$

This storage constraint ensures that for every storage unit u_s , the storage level $es_{level_{u_s,t_r}}$ at the end of time instance t_r within the full year is equal to the streams $es_{in_{u_s,t}}$ and $es_{out_{u_s,t}}$ going in and out of the storage unit during this time instance and the storage level $es_{level_{u_s,t_r-1}}$ at the end of the previous time instance multiplied with the storage efficiency σ_{u_s} , indicating the losses from discharge during every time instance.

A diverse set of tests is applied to verify the functioning of the model, which confirms the unit implementations, working of the constraint definitions and correct computations of the systems' characteristics, such as the overall costs. An overview of the methods employed for verifying the optimisation framework is given in Appendix A.

3.5 Objective Functions

The implemented energy system model can be optimised for different objectives [34]: operating costs, capital investment costs, a combination of the latter two, or the GWP. The definitions (adapted from [34]) of these will be given in this section.

The first objective are the yearly operating costs YC^{op} , for which the objective function is shown in Equation 24.

$$YC^{op} = \sum_{p \in \mathbf{P}} D_p \cdot \sum_{t \in \mathbf{T}(p)} \left(C_{p,t}^{el,+} \cdot \dot{e}_{p,t}^{gr,+} + C^{ng,+} \cdot \dot{h}_{p,t}^{gr,+} + \sum_{u \in \mathbf{U}} \left(C_u^{op,1} \cdot f_{u,p,t} \right) \right)$$
(24)

The operating costs are defined as the sum of the individual operating costs for all periods P, each with their respective hourly divisions T(p). They are multiplied with the frequency of occurrence D_p of every period p. For each hour, the operating cost is composed of the grid electricity used $\dot{e}_{p,t}^{gr,+}$ multiplied with the price of grid electricity $C_{p,t}^{el,+}$ during that hour, the natural gas used $\dot{h}_{p,t}^{gr,+}$ multiplied with the price of natural gas $C^{ng,+}$ and the operating costs of the units. The operating costs of the units are calculated as the sum of the usage factor $f_{p,t,u}$ during every hour and the operating cost $C_u^{op,1}$ for all units U.

The annualised yearly investment costs YC^{cap} can be calculated from the overall investment costs YC^{inv} by annualizing them as shown in Equation 25. The overall investment costs are computed from Equation 26

$$YC^{cap} = \frac{ir \cdot (1+ir)}{(1+ir)^{lt} - 1} \cdot YC^{inv}$$

$$\tag{25}$$

An interest rate of ir = 6% is in general used for this study, with an assumed lifetime of lt = 25 years.

$$YC^{inv} = \sum_{u \in U} \left(C_u^{inv,1} \cdot y_u + C_u^{inv,2} \cdot f_u \right)$$
(26)

The investment costs of all units U can be calculated from the sum of the fixed investment costs $C_u^{inv,1}$ multiplied with a binary variable y_u indicating the presence of the unit and the variable investment costs $C_u^{inv,2}$ multiplied with the usage factor of the unit.

Finally, Equation 27 for obtaining the yearly GWP YG^{op} of the energy system looks very similar to the equation for the operating costs. For the emissions, it must be noted that in this study only scope-1 and -2 emissions are considered. These are emissions directly generated from imported or produced energy resources on site (scope-1) or emissions released in the generation process of imported electricity (scope-2) [134]. Other upstream or downstream emissions (scope-3) are neglected in this study.

$$YG^{op} = \sum_{p \in \mathbf{P}} D_p \cdot \sum_{t \in \mathbf{T}(p)} \left(G_{p,t}^{el,+} \cdot \dot{e}_{p,t}^{gr,+} + G_{p,t}^{ng,+} \cdot \dot{h}_{p,t}^{gr,+} + \sum_{u \in \mathbf{U}} \left(G_u^{op,1} \cdot f_{u,p,t} \right) \right)$$
(27)

Now, $G_{p,t}^{el,+}$ are the emissions assigned to every kWh of electricity at any hour of the year, $G_{p,t}^{ng,+}$ the emissions assigned to every kWh of natural gas, and $G_u^{op,1}$ the direct on-site emissions from using unit u.

3.6 Key Performance Indicators

Different key performance indicators are used next to the objectives to assess the different energy system configurations obtained. First, the efficiency of the energy system is addressed. This is done using energy and exergy efficiency as can be seen in Equation 28 and Equation 29 [34].

$$\eta_{en} = \frac{En^{fuel,-} + E^{dem,-} + Q^{dem,-}}{En^{fuel,+} + E^{gr,+} + E^{ren,-}}$$
(28)

Energy efficiency η_{en} can be computed as the sum of the useful energy leaving the energy system, which are the fuels sold $En^{fuel,-}$, the electricity demand of the airport $E^{dem,-}$ and the heat demand of the buildings $Q^{dem,-}$, divided by the energy going in the energy system, being the sum of fuels $En^{fuel,+}$ and electricity $E^{gr,+}$ consumed by the energy system and renewable energy generated $E^{ren,-}$ on site.

$$\eta_{ex} = \frac{Ex^{fuel,-} + Ex^{dem,-} + Ex^{Q^{dem},-}}{Ex^{fuel,+} + Ex^{gr,+} + Ex^{ren,-}}$$
(29)

Exergy efficiency η_{ex} is computed in a similar way, now dividing the sum of the useful exergy leaving the system, which is composed of the exergy $Ex^{fuel,-}$ of the sold fuels, the exergy $Ex^{dem,-}$ of the electricity demand as well as the exergy $Ex^{Q^{dem},-}$ of the heating demand, divided by the exergy entering the system, being the sum of the exergy $Ex^{fuel,+}$ of the fuels used, the exergy $Ex^{gr,+}$ of electricity from the grid and the exergy $Ex^{ren,-}$ of the renewable electricity generated at the airport.

Energy and exergy efficiency can be computed both including or excluding the efficiency of the local renewable electricity production methods. For the PV, the energy upstream of them is computed by multiplying their area with the solar irradiation [34]. The exergy per square meter upstream of the solar cells is computed using the following Equation 30 [135].

$$Ex_{irr} = I \cdot \left(1 - \frac{4}{3} \frac{T_0}{T} + \frac{1}{3} \left(\frac{T_0}{T} \right)^4 \right)$$
(30)

Where I is the solar radiation, T_0 is the temperature of the environment and T the temperature of the sun. For the wind energy, the exergy and the energy upstream of the wind turbine are computed using the following Equation 31 [136], which determines the amount of energy available across the swept area of the wind turbine, and which is equivalent to Equation 15 without the efficiency of the wind turbine.

$$Ex_{wt} = \frac{1}{2} \cdot A_{sw} \cdot \rho_{air} \cdot v_t^{\ 3} \tag{31}$$

Next, self-sufficiency of the energy system in terms of electricity consumption can be looked at, which is defined as can be seen in Equation 32 [34]. It indicates how much of the electricity demand of the airport is fulfilled with electricity generated at the airport.
$$SS = \frac{E^{loc,+} - E^{gr,-}}{E^{loc,+} + E^{gr,+} - E^{gr,-}}$$
(32)

Where $E^{loc,+}$ is electricity locally produced at the airport.

Finally, the renewable electricity share can be looked at, as displayed in Equation 33 [34]. This indicator allows to see how much of the electricity consumed at the airport is renewable electricity.

$$RES = \frac{\sum_{p \in \mathbf{P}} \sum_{t \in \mathbf{T}(p)} \dot{e}_t^{gr,+} \cdot (\epsilon_t^{gr,ren}) + E^{loc,ren,+}}{E^{gr,+} + E^{loc,+}}$$
(33)

In this indicator, both the electricity generated on site as well as the electricity imported from the grid is looked at. For the locally produced electricity, the amount of renewable locally produced electricity $E^{loc,ren,+}$ is included. For the electricity coming from the grid, the share of renewable electricity $\epsilon_t^{gr,ren}$ at every hour in the year is included.

4 Energy Demand Estimation

Having described the full methodology, this section presents the case study airports to which the energy demand estimation and system optimisation will be applied. The main airport, on which the analysis is based, is Amsterdam Schiphol airport, for which the input data is shown in section 4.1 and the estimated energy demand is displayed in section 4.2. Next, section 4.3 explains which other European hub airports are chosen to test the model. Finally, section 4.4 depicts the typical days constructed from the estimated energy demand.

4.1 Input Data Amsterdam Schiphol

The main case study looked at in this study is on Amsterdam Schiphol airport, the largest airport of the Netherlands. It welcomes more than 60 million passengers every year and operates more than 400,000 air traffic movements per year [137]. The airport is used as the main hub of the largest Dutch airline, KLM [138]. An overview of the main figures of Amsterdam Schiphol airport can be seen in Table 5.

Airport characteristic	Value	Reference
Building surface area $[m^2]$	$253,\!481.1$	computed from [74] data
Building wall length $[m]$	11,543.0	computed from [74] data
Number of passengers in 2023 [-]	$61,\!885,\!367$	[87]
Number of flights in 2023 [-]	460,035	[86]

Table 5: Airport characteristics of Amsterdam Schiphol airport

A surface area of more than $250,000m^2$ is estimated using data from [74]. They are distributed over 8 piers and 3 terminals, which are displayed in red in Figure 11. The other buildings, coloured in black on the map, are excluded from the analysis, as is explained in section 3.1.



Figure 11: Map of considered terminal and pier buildings at Amsterdam Schiphol airport (using data from [74])

As described in section 3.2, different input data needs to be collected before determining the energy demand and optimising the airport's energy system. The number of flights and the estimated number of passengers over the year of 2023 can be seen in Subfigure 12(a). Next, Subfigure 12(b) shows the variation of the electricity price as well as the estimated emissions per kWh of electricity in 2023 in the Netherlands.



Figure 12: Characteristics of case study of Amsterdam Schiphol airport in 2023: (a) number of flights [86] and passengers (monthly passenger numbers from [87]) and (b) electricity price [90] and emissions (average yearly emissions per kWh from [91], yearly profile of renewable energy share from [90]) in the Netherlands

Furthermore, weather data needed to be obtained. Figure 13 shows the profile of the temperature, solar radiation, wind speed and sunshine duration in 2023 at Amsterdam Schiphol airport.



Figure 13: Weather data at Amsterdam Schiphol airport in 2023 [88]: (a) temperature and solar radiation and (b) wind speed and sunshine duration

4.2 Energy Demand Amsterdam Schiphol

With the input demand from section 4.1 as well as the methods described in section 3.1, the energy demand profile for Amsterdam Schiphol airport for the case study year 2023 can be determined. Figure 14 shows an overview of the weekly moving average of the input energy demand which is used for optimising the airport's energy system. For the heating and cooling demand, clear seasonal variations can be seen, with the heating needs being higher during winter and the cooling needs during summer. The electricity and vehicle energy demand are more constant over the full year, as they are less dependent on the weather and more on the number of passengers and flights. The vehicles make up around 10% of the overall energy demand. The smallest demand is the domestic hot water demand, which is added to the heating demand of the buildings.



Figure 14: Weekly moving average of the estimated input energy demand for Amsterdam Schiphol airport

The data for the vehicle as well as building electricity demand has been determined based on known values for the year of 2023. The energy demand of the heating, ventilation and cooling system has however been estimated based on the building envelope. The values obtained can be compared to actual energy consumption data of Amsterdam Schiphol airport for the year of 2023 [71], which is done in Table 6. It can be seen that the heating, ventilation and cooling demand are all within less than 15% of the approximated demands from the statistics. Only the estimated domestic hot water appears to be 24% lower than the approximated real airport demand. However, its value being an order of magnitude smaller than the overall heating demand, this difference can be neglected. Thus, the energy demand model results obtained can be validated for the case of Amsterdam Schiphol airport.

HVAC consumer	Model results energy demand [GWh]	Estimated real airport data energy consumption [GWh]
Heating	37.1	37.6
Cooling	13.6	12.0
Domestic hot water	1.9	2.5
Ventilation	30.5	26.2

Table 6: HVAC system comparison model results and estimate based on real airport data [71]

4.3 Estimated Energy Demand European Hub Airports

As the energy demand model is a universal model applicable to different airports when knowing the number of flights and passengers, the building characteristics as well as weather data, the model is also applied to other airports than Amsterdam Schiphol. In consequence five other major European hub airports from different climate zones are chosen. The selection of case study airports can be seen on Figure 15.



Figure 15: Map with the locations of the case-study airports and their respective climate zone (adapted from [139], [140])

Including Amsterdam Schiphol, the six airports chosen represent well the different climate zones [139] existing in Europe. Madrid and Rome have a warm Mediterranean climate with mild winters and hot summers, with Madrid also having a minor maritime influence. Munich and Zurich lie in central Europe characterised by a continental and more stable climate, with Zurich also having a slight alpine climate influence. Amsterdam is located in the Northern maritime climate zone with varying weather conditions. Finally, Oslo represents the Northern part of Europe with a southern boreal climate, having colder winters than the other locations.

The airport characteristics, including the building sizes as well as the number of flights and passengers, can be found in Table 7. Additional input data, including the airport map with the considered buildings as well as the flight, passenger, weather, electricity and energy demand profiles for all airports, are provided in Appendix B. Whilst all airports have a comparable magnitude, it should be noted that Amsterdam and Madrid are the largest airports of this sample, both having more than 60 million passengers per year. On the other hand, Oslo and Zurich are the smallest two airports with 25 to 30 million passengers per year. When computing the relative sizes, it can be observed that the number of passengers per m² range from 154 at Munich airport to 252 at Zurich, having an influence on the determined energy demand. Next to that, the average number of passengers per flight ranges from 117 in Zurich to 158 in Madrid.

 Table 7: Airport characteristics in 2023 for airports chosen as case studies: building surface area [74], building wall length [74], number of passengers [87], [141]–[145] and number of flights [86]

Airport	ICAO code	Building surface area $[m^2]$	Building wall length $[m]$	- · · · · · · · · · · · · · · · · · · ·			
Amsterdam Schiphol	EHAM	253,481.1	11,543.0	61.9	460.0		
Madrid-Barajas	LEMD	370,226.4	12,561.9	60.2	380.3		
Munich	EDDM	240,035.6	14,373.8	37.0	296.9		
Oslo-Gardermoen	ENGM	106,172.9	7,904.5	25.1	211.6		
Rome-Fiumicino	LIRF	193,935.6	11,373.6	40.5	261.3		
Zurich	LSZH	114,443.6	6,087.9	28.9	245.8		

From the airport building, passenger, flight and weather characteristics, the energy demands for the different airports are again computed using the methods described in section 3.1. The resulting profiles can be seen in Figure 16.



Figure 16: Comparison of energy demand for the different case study airports: (a) heating demand, (b) cooling demand, (c) eletricity demand and (d) vehicle energy demand

For the heating demand in Subfigure 16(a), it can be seen that Munich displays the highest value of 47.5GWh/y/pax per passenger, due to its continental climate and the large surface area per passenger. This large surface area per person is also the explanation why Madrid has a relatively high heating requirement per passenger of 38.4GWh/y/pax even whilst being located in a Mediterranean climate, whereas Zurich has a relatively low requirement per passenger of 19.5GWh/y/pax despite being located in an Alpine climate.

Looking at the cooling demand in Subfigure 16(b), as expected due to the Mediterranean climate a very high demand per passenger can be observed for Madrid and Rome with values of 39.3GWh/y/pax and 28.6GWh/y/pax respectively. Their cooling needs are particularly high during the spring and summer period. On the other hand, as expected the lowest values are observed for Zurich and Oslo with values of 9.2GWh/y/pax and 6.1GWh/y/pax, respectively.

For the electricity demands, the highest values are observed for Madrid, Amsterdam and Munich with values from 113.1GWh/y/pax to 72.6GWh/y/pax. It can be noted from Subfigure 16(c), that Munich has the highest per passenger electricity demand. Oslo and Zurich have the lowest values per passenger of 35.0GWh/y/pax and 38.2GWh/y/pax. It can be seen that the electricity demand is mostly influenced by the buildings surface area due to the large share of the ventilation requirements in the overall electricity demand, whereas heating and cooling requirements are mostly influence by the weather.

Finally, Subfigure 16(d) displays the vehicle energy demand. Similar profiles can be obtained, with Amsterdam having the largest overall requirement of 17.1GWh/y/pax and Oslo having the lowest requirement of 7.9GWh/y/pax. The values for the vehicle energy consumption fully depend on the number of flights, different to the previously mentioned demands influenced by the number of passengers.

To validate the model for the energy demands of other airports than Amsterdam Schiphol, the overall yearly demands are compared to overall yearly demands of European airports of similar sizes for the last years, which are obtained in the literature study. The comparison can be seen in Figure 17.



Figure 17: Comparison of overall airports energy consumption per passenger between model results and real airport data from [5], [10]–[32]

From the comparison with real European airport demand data, it can be concluded that the estimated values fall in the same range, but are comparable rather to the lower than to the higher airport energy demands. When drawing a trend line, it can be seen that the average model results are approximately 40% lower than the actual data points. Considering that the terminal building and airfield handling energy demand at Amsterdam Schiphol airport made up 61% of the overall airport demand, this deviation from the real airport data seems to confirm the correct working of the model. However, conclusions can only be drawn to a limited extent from this comparison, as airport demands or consumptions are published in slightly different ways for all airports. Thus, this comparison is not able to confirm precisely the estimated airport energy consumptions but can confirm results of correct magnitude for the energy demand model.

4.4 Typical Days

From the estimated energy demands, typical days are finally constructed in order to reduce the optimisation complexity, as is described in section 3.2.2. The resulting typical days, based on the temperature, solar irradiation and wind speed, can be seen in Figure 18. It can be observed that the typical days 1 to 6 are more winter days with low temperatures and irradiation, whereas days 7 to 12 represent summer days with high temperatures and radiation. Across both types of days, there are days with more and less average wind speed.



Figure 18: Clustering of the typical days for Amsterdam Schiphol airport

Figure 19 displays the range of temperature, solar radiation and wind speed covered by the typical days.

It can be observed that the typical days cover well all days but lack at the edges of the curves, where the highest and lowest values occur. Thus, it is important to include the extreme days in the analysis so that the technologies are correctly scaled to also produce sufficient energy during extreme days.



Figure 19: Clustering of the typical days for Amsterdam Schiphol airport

5 Optimisation Results

With the typical days obtained from the input energy data, the estimated energy demand shown in section 4 and the superstructure and optimisation framework described in section 3, the airport's energy system can be designed. First, the energy system will be optimised in section 5.1 for different scenarios for Amsterdam Schiphol. Next, the possibility of producing synthetic fuels is looked into in section 5.2. Finally, section 5.3 explores the option of applying the optimisation framework to other European hub airports.

5.1 Energy System Optimisation Amsterdam Schiphol

In a first step, the energy system for Amsterdam Schiphol is analysed and possible solutions are generated. First, a set of different solutions is generated using a Pareto analysis in section 5.1.1. Next, the different configurations observed are further analysed in section 5.1.2. The option of a completely independent energy system is looked into in section 5.1.3. Finally, section 5.1.4 displays the sensitivity of the optimal energy system on several technology investment costs.

5.1.1 Generation of Energy System Solutions for Different Scenarios

Different solutions of possible energy systems fulfilling the energy demands shown in Figure 14 need to be generated, allowing to observe possible technology configurations and to select specific solutions for further analysis. This is done using a Pareto analysis approach. This approach allows to generate distinct configurations performing optimal for different objectives. In this way, on the one hand results are obtained performing particularly well with maximal usage of the grid, while on the other hand configurations which produce more electricity on site are created. The range of results can then be studied and promising results can be selected for more detailed analysis. Six different scenarios are used for generating the Pareto curves, which differ by the type of energy storage technologies implemented. The six scenarios are the following:

- No storage: No storage technology implemented, all other technologies allowed to be used.
- No storage, no natural gas: No storage technology implemented, technologies employing natural gas (boiler, combined cycle) prohibited.

- **Battery:** Battery implemented as storage technology, all other non-storage technologies allowed to be used.
- **Battery, no natural gas:** Battery implemented as storage technology, technologies employing natural gas (boiler, combined cycle) prohibited.
- Hydrogen: Pressurised hydrogen tank together with electrolysis and fuel cell implemented as storage technology, all other non-storage technologies allowed to be used, no reduction in hydrogen technology costs.
- Hydrogen, -50% costs: Pressurised hydrogen tank together with electrolysis and fuel cell implemented as storage technology, all other non-storage technologies allowed to be used, 50% reduction in the hydrogen technology investment costs ($c_u^{inv,1}$ and $c_u^{inv,2}$).

The three base scenarios are the options of no storage, battery or hydrogen storage. For the scenario with no storage unit, it is furthermore investigated if not having any non-renewable resources would alter the optimisation results. As the battery is observed to be the more competitive storage technology, it is analysed how performance would change when not including non-renewable resources. For the hydrogen storage scenario, it is investigated whether a 50% reduction in the costs of the hydrogen related technologies (hydrogen tank, electrolysers, fuel cell) would improve competitiveness compared to the battery. It must be noted that the production of synthetic fuels is left out in those scenarios and will be looked at specifically in section 5.2.

For all of the scenarios, a six-point Pareto curve is constructed. The Pareto analysis is performed by optimising the total costs of the energy system with a varying weight on the annualised investment and operating costs, as shown in Equation 34.

$$Objective \ Function = \alpha^{inv} \cdot YC^{inv} + (1 - \alpha^{inv}) \cdot YC^{op} \tag{34}$$

Variation of the importance of investment and operating costs is obtained by altering the weight on the investment cost α^{inv} from 1.0 in steps of 0.1 to 0.001. The last value is chosen to avoid technologies being scaled to their maximum size, despite not being in use and thus having no operating cost assigned.

The results of the Pareto analysis for the six scenarios can be seen in Figure 20, with the detailed cost distributions in terms of operating costs and investment costs shown in Figure 21. The lowest overall cost obtained is of $9.33M\notin/y$ and is obtained when using a battery and non-renewable technologies for an $\alpha^{inv} = 0.5$. It is composed of an operating cost of $5.46M\notin/y$ and an investment cost of $3.87M\notin/y$. Slightly higher minimum total costs of $9.40M\notin/y$ are obtained in the scenario where no energy storage is allowed. The battery is observed to perform well in further decreasing operating costs at only a slight increase of total costs. For instance, reducing the operating costs by 47.7% to $2.86M\notin/y$ leads only to a 7.2% increase of the total costs to $10.00M\notin/y$. When not allowing battery storage, even an operating cost of $3.79M\notin/y$ can only be achieved with a total cost of $16.42M\notin/y$. In fact, this is higher than the required investment cost of $15.87M\notin/y$ for achieving zero operating cost with the battery. The latter displays that with 70.1% total cost increase compared to the optimal solution, the battery allows to completely avoid operating cost.

In the fully renewable scenarios, where natural gas is not allowed, the minimum total costs increase both with and without battery storage. In the scenario not allowing any energy storage, the minimum costs increase by 22.7% to $11.54M \notin /y$. When allowing for battery storage, the minimum costs only increase by 16.2% to $10.84M \notin /y$. The advantage of using the battery for minimising total costs is thus increased from a 0.8% to a 6.1% lower cost achievable when not allowing non-renewable technologies anymore. It can thus be concluded that while the battery has no significant advantage for conventional energy systems using fossil resources, in fully renewable energy systems relying partially on intermittent energy sources, the battery is a competitive technology for minimising the total costs of the system.

When allowing for the integration of hydrogen technologies, the hydrogen storage system is not employed in the optimisation minimising total costs with an $\alpha^{inv} = 0.5$. It is only employed for $\alpha^{inv} \leq 0.2$, which displays the important investment cost needed for levelling intermittent energy supply using a hydrogen storage system, especially for the electrolysers and fuel cells. Even when reducing investment costs of all hydrogen technologies by 50%, this type of energy storage is only employed for $\alpha^{inv} \leq 0.3$. In the most cost optimal solution where hydrogen storage is employed for the scenario without a cost reduction for hydrogen technologies ($\alpha^{inv} = 0.2$), the implementation of the hydrogen technologies requires total costs of $13.59M \notin/y$, being 45.6% higher than the minimum total costs when using a battery. Even when reducing the investment costs by 50%, the minimum total costs of $10.78M \notin/y$ are 15.5% higher than when using a battery. Moreover, from Figure 20 it can clearly be seen that the hydrogen storage system does not allow to reduce operating costs as efficiently as the battery. Thus, it can be concluded that the high investment costs and the lower round-trip-efficiency of the hydrogen storage system make the battery the more competitive energy storage option. The latter is mostly used to level intermittency of energy production on a daily and not seasonal basis, as will be seen in section 5.1.3.



Figure 20: Pareto front for different scenarios analysed

In Figure 21, the investment and operating cost distributions per technology of all 66 generated solutions are displayed. It can be observed that when optimising only on investment costs, energy systems relying on natural gas boilers, chillers and electricity from the grid are the most common option, as visible for example in Subfigure 21(a). When prohibiting the use of natural gas, as done in Subfigure 21(b), the natural gas boiler is replaced with a heat pump to minimise investment costs.

When reducing overall costs, the best solutions obtained are employing a combination of a combined cycle running on natural gas, a CO_2 network, PV and electricity from the grid, as again visible for example in Subfigure 21(a). As discussed before, even lower total costs are observed when allowing the use of a battery, as visible in Subfigure 21(c). The combined cycle is replaced with a larger CO_2 network and an increased battery capacity when prohibiting the use of natural gas.

Finally, when optimising operating costs with $\alpha^{inv} = 0.001$, except when integrating batteries, solutions with unreasonably high investment costs are obtained (visible in Subfigures 21(a), 21(b), 21(e) and 21(f)). In those solutions, the optimisation is choosing to import a maximum amount of electricity when grid prices are negative and is then dissipating the electricity using additional units not necessary for fulfilling the demand. Despite constraining that the yearly sum of the costs for buying grid electricity cannot be negative, solutions with unreasonably high investment costs are still obtained. In consequence, it is decided to disregard those solutions in the future analysis, as they are neither competitive, nor sustainable or energy efficient.

5.1.2 Selected Solutions for Detailed Analysis

From the Pareto analysis depicted in section 5.1.1, the most common technology combinations are obtained. For each configuration characterised by the main technologies employed, the solutions with minimum total costs are selected, in order to observe the performance of all technologies employed. Furthermore, for the configurations including storage units, solutions with different storage sizes are chosen to observe the impact of a higher storage capacity. The solutions are analysed for different key performance indicators relating to investment and operating costs, as was done in the previous subsection.

Table 8 gives an overview of the selected solutions. It can be seen that all technologies are employed in at least one of the solutions as a main technology. Five technology configurations not using natural gas and seven technology combinations using natural gas are selected. Furthermore, six technology combinations employ no storage unit, four technology combinations employ a battery and two technology combinations the hydrogen tank. The most frequent solution for the scenario with no storage unit uses PV, a combined cycle and a CO_2 network. When employing the battery, the most frequent solutions are using, next to the battery, PV and a CO_2 network, with an additional combined cycle when allowing non-renewable technologies. One of the solutions employing the hydrogen tank furthermore make use of a mini wind turbine. An overview of the installed technology capacities can be found in Appendix C, which also allows the verification of these configurations, including the costs.



Figure 21: Cost distribution for the different points shown in the pareto front plot in Figure 20 for the scenarios with (a) no storage units, (b) no storage units and no natural gas, (c) only battery storage, (d) only battery storage and no natural gas, (e) only hydrogen storage and (f) only hydrogen storage with hydrogen units at 50% reduced costs

Table 8:	Overview	of selected	solutions	for further	analysis from	the Pareto	front in	Figure 20

Solution name	Number	Optimisation Main technologies											
solution name of occur- rences		Scenario	α^{inv}	Battery (BAT)	Hydrogen storage (H_2S)	Grid electricity	PV (PV)	Mini wind turbine (WT)	Combined cycle (CC)	Boiler (BOI)	Heat Pump (HP)	CO ₂ network (CO ₂ N)	Chiller (CHI)
BOI+CHI	8	No storage	0.9			Х				Х			Х
CC+BOI+CHI	8	No storage	0.7			Х			Х	Х			Х
HP+CHI	6	No storage, no nat. gas	0.8			Х					Х		Х
$PV+CO_2N$	4	No storage, no nat. gas	0.5			Х	Х					Х	
PV+CC+CO ₂ N	13	No storage	0.5			Х	Х		Х			Х	
PV+WT+CO ₂ N	1	No storage, no nat. gas	0.1			Х	Х	Х				Х	
$BAT+PV+CC+CO_2N$	4	Batteries	0.5	Х		Х	Х		Х			Х	
$BAT+PV+CO_2N$	9	Batteries, no nat. gas	0.5	Х		Х	Х					Х	
XBAT+PV+CC+CO ₂ N	(4)	Batteries	0.3	Х		Х	Х		Х			Х	
XBAT+PV+CO ₂ N	(9)	Batteries, no nat. gas	0.2	Х		Х	Х					Х	
H ₂ S+PV+CC+CO ₂ N	4	Hydrogen storage	0.2		Х	Х	Х		Х			Х	
$XH_2S+PV+CC+CO_2N$	(4)	Hydrogen storage	0.1		Х	Х	Х		Х			Х	

From the overview of the technologies, it can already be concluded that the PV, the combined cycle and the CO_2 network are the most commonly employed technologies in the optimised energy systems. They are being used both in different scenarios as well as for different objective functions, which proves their competitiveness under varying circumstances. The robustness of the configurations will further be analysed in section 5.3, where the optimisation will be extended to different airport environments.

It is important to be aware of the limitations of this approach when analysing the results for the selected energy systems. With the Pareto analysis, a number of feasible configurations are created that perform variously well either in terms of operating, investment costs or a combination of both. When looking at different key performance indicators, it is important to note that the configurations have not been optimized for those. Although the configurations give a good overview of achievable values, the range of results does not allow final conclusions to be drawn about the maximum possible performance. A further analysis in future studies would be required that specifically optimises the performance of individual key performance indicators and creates configurations for them.

For the selected solutions, the key performance indicators are computed and displayed in Figure 22. These include the yearly total costs as already shown before, the GWP, the share of renewable energy used, the energy system self-sufficiency and the exergy efficiency of the energy system. The latter efficiency is calculated excluding the exergy efficiency of PV, to avoid penalising solutions with a high area of PV installed.



Figure 22: Key performance indicators of selected solutions from Pareto fronts from different scenarios, as described in Table 8

Different observations can be made. First, the solutions employing natural gas boilers (BOI+CHI and CC+BOI+CHI) are looked at. While having overall costs comparable to the remaining energy system configurations, they display the highest GWP of 36.4ktCO₂eq/y and 34.5ktCO₂eq/y, but the lowest share of renewable energy with not more than 15.1% and a self-sufficiency of 0%. When replacing the boiler with a heat pump, as is done in the configuration HP+CHI, very similar results are obtained with a slightly lower GWP (4.9% lower

than for the solution with a combined cycle, a boiler and a chiller) and an improved exergy efficiency of 78.0%, being in the higher range of efficiencies from all configurations.

The GWP is significantly reduced when adding PV to the energy system, thus avoiding part of the emissions associated to the electricity from the grid. In combination with a CO₂ network and both without (PV+CO₂N) and with a combined cycle for local electricity generation from natural gas (PV+CC+CO₂N), a medium to low GWP (21.9 and 26.0ktCO₂eq/y), renewable energy share (39.9% and 16.3%) and self-sufficiency (23.2% and 15.7%) are obtained compared to the other solutions. Nevertheless, the solution including the combined cycle scores second best for the total costs (9.4M€/y), whereas the one only with PV and a CO₂ network scores highest for the exergy efficiency, with a value of 88.8%.

When adding a battery to those two configurations, as is done in BAT+PV+CC+CO₂N and BAT+PV+CO₂N similar exergy efficiencies and costs are obtained, with the former solution having the lowest total costs of $9.3M \notin$ /y. However, in the case with no combined cycle, the GWP is further reduced down to 17.7ktCO₂eq/y and the renewable energy and self-sufficiency shares are improved to 52.1% and 36.7% respectively.

While having only a slightly higher total cost of $11.1M \notin/y$ than the previous battery solutions, the configuration XBAT+PV+CC+CO₂N, employing a battery with a significantly higher capacity of 146.2MWh, has a reduced GWP of 10.9ktCO₂eq/y and a relatively high renewable energy share and self-sufficiency of 57.0% and 56.4% respectively. The solution without using a storage but adding mini wind turbines (PV+WT+CO₂N) allows to achieve a close GWP of 14.6ktCO₂eq/y, which is the best of all configurations without storage technologies. Furthermore, this solution allows to obtain a similar renewable energy share of 63.0% and self-sufficiency of 54.9%. However, the total costs when introducing mini wind turbines are 22.0M /y and thus 98.2% higher than for the comparable battery configuration (XBAT+PV+CC+CO₂N).

The last battery configuration XBAT+PV+CO₂N with a 35.9% higher battery capacity of 198.7MWh strikes out for three of the key performance indicators. It has the lowest GWP (4.5ktCO₂eq/y), displays the highest share of 88.4% renewable energies and displays the second highest self-sufficiency of 83.1%. Meanwhile, it has an average exergy efficiency and only a 61.8% higher cost of $15.1M \notin$ /y than the configuration with the lowest total cost. Thus, this configuration shows that self-sufficiencies of above 80% are possible at acceptable costs, whereas self-sufficiencies of 90% or more require investment costs which are at least twice the minimum investment costs.

Finally, the configurations employing a hydrogen tank can be looked at. The least expensive hydrogen configuration $(H_2S+PV+CC+CO_2N)$ with a hydrogen storage capacity of 104.7MWh displays similar results to the remaining solutions, due to the relatively small scale of the hydrogen system. However its exergy efficiency is only 58.5% and thus the second lowest of all configurations due to the low round trip efficiency of hydrogen storage. The lowest exergy efficiency of 48.9% is obtained for the second configuration including a hydrogen storage system $(XH_2S+PV+CC+CO_2N)$. While furthermore displaying the highest total costs of 24.3M \notin /y, this configuration on the other hand has the second lowest GWP, renewable and self-sufficiency due to a hydrogen storage capacity of 276.9MWh.

Overall, it can be seen that including a natural gas boiler particularly decreases the performance of the energy system by provoking a low self-sufficiency and GWP. While having comparable emission values, when employed, the combined cycle from natural gas often complements PV in electricity generation and thus most configurations including it display better performances in terms of GWP, renewable energy share and self-sufficiency. Next, it can be concluded that the CO_2 network is a significant contributor to solutions scoring well in terms of GWP, renewable energy share and self-sufficiency, while having costs and efficiencies comparable to the remaining solutions, and is thus a very efficient and attractive solution for heating and cooling of the building. Furthermore, it can be observed that PV have a crucial importance for increasing the self-sufficiency and reducing the GWP by avoiding the emissions associated with the grid. Finally, the most important contribution towards a low GWP and high self-sufficiency is the introduction of storage units, which allow to store on-site generated green electricity and to use it during periods with no production but high demand. For the two types of storage units implemented, batteries show better performance than hydrogen storage in terms of costs and exergy efficiency.

Below, the composition of the scope-1 and -2 emissions contributing to the GWP can be seen in Figure 23. It can be observed that for most configurations, the largest contribution comes from the emissions associated to the grid electricity due to the large demand of electricity, representing approximately 2/3 of the overall energy demand. In fact, the share of the grid electricity in the overall emissions ranges from 25.2% for a configuration involving a battery to 77.4% for the solution with a natural gas boiler and 100% for the solutions with no natural gas consumption. The overall yearly emissions range from $36.4\text{ktCO}_2\text{eq/y}$ for the configuration using a boiler to $4.5\text{ktCO}_2\text{eq/y}$ for the solution only using the largest battery, a PV, a CO₂ network and a small amount of grid electricity supply in off-peak periods. Overall, it can be concluded that considerable CO₂ emission reductions can be obtained with an optimised energy system, but an important contribution also needs to be made by the grid electricity providers in transitioning to renewable electricity generation technologies and reducing emissions associated to the grid.



Figure 23: GWP distribution from yearly emissions of selected solutions from Pareto fronts from different scenarios, as described in Table 8

The yearly results obtained for the GWP can be compared to the greenhouse gas emissions of the Royal Schiphol Group for 2023, which provides combined scope-1 and -2 emissions of 75.7ktCO₂eq/y [16]. The emissions of the proposed energy system configurations are observed to be 2 to 17 times smaller, indicating the reduction potential through integrating renewable technologies and local electricity production. However, this comparison should be treated with caution, as the stated emissions may include additional emission sources and airport components that are not modelled in this optimisation framework. Furthermore, the emissions indicated by the Royal Schiphol group also include the emissions of Rotterdam The Hague airport and Eindhoven airport, which are, however, significantly smaller.

Moreover, the electricity mix of the airport over the full year for the different scenarios is shown in Figure 24. It allows to observe both differences in the electricity mix as well as for the overall amount of electricity consumed.

Different seasonal trends can be observed. For the PV, supply is highest during summer, when solar radiation also is strongest, as can be seen in Figure 13. On the other hand, the combined cycle as well as the electricity grid are compensating intermittency of PV electricity supply and are thus highest during winter, when being in use together, as can be seen for example in Subfigure 24(e). Moreover, they are also supplementary to each other with the combined cycle being used most when electricity from the grid is expensive. As can be seen in Subfigure 24(f), the supplementary behaviour of the combined cycle could be replaced by the mini wind turbine at an increased investment cost, as can be seen in Figure 22.

The battery can be observed to supply electricity more or less constantly over the full year, as can for example be seen in Subfigure 24(j). The nearly constant supply from the batteries shows that it is mostly used for daily and not for seasonal storage, as will be explained also in section 5.1.3. Furthermore, from Subfigure 24(i) it can be seen that the share in supply from the battery is lower in periods with higher electricity production from the combined cycle of natural gas, as the latter is used to compensate the smaller potential of renewable electricity production, which also causes a smaller energy storage potential in those periods. Similar observations can be made for the hydrogen storage system with the SOFC supplying electricity, which can for example be seen in Subfigure 24(k). However, the hydrogen storage system displays much smaller overall supply than the battery.

When looking at the overall supply of electricity, it can be seen that it is lowest for the solution employing boilers or combined cycles with 102GWh to 109GWh as they are not requiring any electricity for heating. Adding heating technologies based on electricity as well as energy storage technologies increases the overall electricity consumption of the energy system, leading to overall yearly consumptions of 111GWh to 125GWh. The highest yearly electricity consumptions of 132GWh to 192GWh are observed for the energy systems involving hydrogen storage due to its low round trip efficiency, as can be seen especially in Subfigure 24(1).

Finally, the overall yearly production of PV electricity can be looked at. It is observed to range from 24GWh for the configuration PV+CC+CO₂N in Subfigure 24(e) to 192GWh for XH₂S+PV+CC+CO₂N in Subfigure 24(l). In 2023, 2.3GWh of PV energy were produced at Amsterdam Schiphol [5]. The proposed configurations including PV energy would thus at least multiply this capacity by a factor of 10. They would require total PV surface areas of 0.13km² to 1.13km². It needs to be confirmed whether enough surface around the airport can be used for such a large-scale installation of PV, also taking into account the glare and radio effects of their deployment. However, it can be noted that studies for smaller airports have already identified potential of a similar magnitude [98]. Similarly, for the energy system configuration PV+WT+CO₂N as shown in Subfigure 24(f), it needs to be verified whether the installation of 2535 mini wind turbines is possible around the airport.



Figure 24: Weekly moving average of electricity supply distribution and electricity supply from storage technologies over the year for the scenarios described in Table 8

5.1.3 Option of Fully Independent Energy System

All of the solutions presented in the previous subsection include the electricity grid as a source of electricity. Even when minimising for zero operating costs, grid electricity at negative and minor positive prices is used. In this subsection, the option of a fully independent energy system without any connection to both the electricity and natural gas grid is analysed.

Whilst having a zero GWP, running fully on renewable energy and being self-sufficient, this energy system configuration displays total investment costs of $32.5M \notin/y$, which are 248.3% higher than the lowest cost obtained from the configurations in section 5.1.1. It can be seen that obtaining a fully-self-sufficient energy system comes at a significant investment cost, whereas self-sufficiencies of around 80% can be obtained at reasonable costs of up to $15.5M \notin/y$, as is shown in Figure 22.

As can be seen in Figure 25, the electricity mix of this configuration consists of electricity from batteries, PV and mini wind turbines. A surface area of 1.02km^2 of PV, 4033 mini wind turbines and a battery capacity of 296.3MWh is installed in this solution, where the PV, the mini wind turbines and the battery represent 22.2%, 42.2% and 28.7% of the overall costs respectively. This displays the significant investment and footprint required for mini wind turbines for only providing a small share of 18.5% of the yearly electricity supply. It can be observed that during summer, PV are the most important electricity source, whereas during winter, mini wind turbines have an equal or even higher share of the generated electricity. Moreover, the battery is 49.1% larger than the largest battery from the selected Pareto configurations. It is again used primarily as a daily storage unit with no significant seasonal variations.



Figure 25: Weekly moving average of electricity supply distribution and electricity supply from storage technologies for scenario with battery and no connection to electricity or natural gas grid

It must be noted that the potential of PV power to be generated in summer is higher than the actual electricity generated. Throughout the year, only 37.6% of the potential PV electricity is used at the airport itself, while the additional potential electricity is not generated due to curtailment of the PV. It may be promising for future research to analyse whether this additional electricity could be fed into the national grid at profitable prices for the airport. When selling the potential excess electricity at varying grid prices, a maximal revenue of $7.3M \notin$ y would be possible, which would reduce the yearly total costs of this configuration by 22.5%.

Next to the electricity mix over the full year, it is interesting to look into the battery's performance specifically, which is used more than in the previous scenarios. Figure 26 shows the battery use profile together with the input data for the solar radiation and the wind speed, both for the full year as well as for two typical days, a winter and a summer day, with different solar radiation and wind speed characteristics.

Computing the overall amplitude for seasonal and daily variation, it is possible to see that the daily variation of the battery state of charge is approximately as large as the seasonal variation for the summer day and nearly twice as large for the winter day. Furthermore, when looking at the change in state of charge at the end of one day, only a minor difference can be observed compared to the beginning of the day. This indicates that the battery is used more as a daily than as a seasonal storage unit, as no significant state of charge is accumulated over the year. The daily storage behaviour allows the battery to store electricity for the night when no solar radiation is available.

When comparing the state of charge to the solar radiation and wind speed profile, as expected, it can be seen that the battery is charging when there is a lot of solar radiation and wind available and that it supplies electricity during off-peak periods. This can be seen particularly well for the daily profile of the winter day, where the battery charges during the full time period in which solar radiation is available, in order to store



Figure 26: Battery state of charge (a) for the full year (weekly moving average) and (b) for a typical winter day (typical day 6) and a typical summer day (typical day 10) combined with solar radiation and wind speed

enough energy for the night. In the case of the summer day, the battery only charges at the end of the day when solar radiation and wind speed reduce to insufficient levels for providing the required energy for the airport. From Figure 25, it can furthermore be observed that daily storage is particularly important during winter days with less wind energy generated. As the battery is mainly used as a daily storage unit, the seasonal profile in Figure 26 does not display those clear parallels. Nevertheless, a lower average state of charge can be observed during summer, which confirms that the use of the battery is most important during extreme winter days.

5.1.4 Sensitivity Analysis

From the Pareto analysis, it can be observed that all units occur in at least one of the selected scenarios but that some technologies only occur at a low weight on the investment cost, for example, the mini wind turbine, which only occurs for $\alpha^{inv} = 0.1$. This is due to the high investment cost of the mini wind turbine compared to the other local electricity generation technologies. In fact, with the implemented costs and the weather characteristics of Amsterdam, the investment costs required for installing a PV with an output of 1MWh per year are $332 \in$ whereas a mini wind turbine with an equivalent yearly electricity output requires an investment of 2, 483 \in . Thus, mini wind turbines would require a significant decrease in investment costs to be competitive local electricity generation technologies. Figure 27 displays how the usage of the mini wind turbine changes at decreased investment costs when minimising the total costs of the energy system.



Figure 27: Resulting electricity supply distribution (a) and overall yearly energy system costs (b) for sensitivity analysis on investment costs of mini wind turbine

From the sensitivity analysis, it can be observed that, only with a decrease of 76% in investment costs, the mini wind turbine will be used as an additional local electricity generation technology. First, it only complements the PV in producing local electricity. When installing additional mini wind turbines, the size of the battery is reduced as less electricity needs to be stored from day- to nighttime. At a cost reduction of 83%, the share of mini wind turbines in the generation of electricity is surpassing the share of solar cells, which is in line with the investment cost ratio between mini wind turbines and PV, the latter having an 87% lower investment cost per MWh produced. Moreover, it can be observed that a decreased cost for the mini wind turbines allows for a decrease in dependency on natural gas combined cycles and grid electricity in the airport's electricity mix and thus allows for a reduction in operating costs and overall costs at an increased investment cost.

Next, the sensitivity of the usage of storage technologies at reduced investment costs can be looked at. First, the effect of reducing the investment cost of batteries in optimisations minimising the total costs of the energy system is analysed. The resulting electricity mix and cost distribution are shown in Figure 28.



Figure 28: Resulting electricity supply distribution (a) and overall yearly energy system costs (b) for sensitivity analysis on investment costs of battery

When optimising the energy system for minimum costs, already at the pre-defined costs, the battery ensures approximately 6% of the yearly electricity supply. It can be seen that this share gradually increases when the investment costs of the battery decrease. Meanwhile, the share of the combined cycle from natural gas decreases, while the share of PV electricity increases, indicating that producing and storing solar power becomes more profitable than producing electricity from fossil resources. At a 60% reduction of the investment cost of the battery and PV increased from 6% and 25% to 30% and 37%, respectively, in the yearly electricity supply, whereas the importance of the combined cycle decreased from 43% to 11%. This shows that a decreased battery investment cost allows the decarbonization of the airport's energy system and the reduction of operating costs at a lower increase in investment costs. At investment cost reductions higher than 80%, the grid electricity increases its share in the electricity supply while the importance of the PV decreases, indicating that the battery is now increasingly used to store electricity bought at low prices for usage in periods with high electricity prices while producing less electricity on site.

Unlike the battery, hydrogen storage is not observed to be useful when minimising the total costs of the energy system. Figure 29 displays the sensitivity of the hydrogen storage system use when only including the hydrogen system as storage technology and decreasing the costs of all hydrogen technologies, being the electrolysers, the storage tank and the fuel cell.

It can be observed that the hydrogen storage system starts being used at an investment cost reduction of 76%. Similar to the battery, together with the PV, it allows for the reduction of the share of the combined cycle in the electricity mix, as well as the reduction of the share of grid electricity. After decreasing the hydrogen technology investment costs by 94%, the hydrogen fuel cell and PV increase their share in the electricity mix from 0% and 24% to 12% and 37%, respectively, whereas the share of the combined cycle and grid electricity decreased from 48% and 28% to 25% and 27%, respectively. Similarly to the battery, at higher cost reductions, the importance of the grid electricity in the electricity mix rises, whereas the importance of the PV and the combined cycle (further) decreases. Overall, the hydrogen storage system also decarbonises the electricity supply system and reduces operating and overall costs, but only to a smaller extent than the battery and only at significant cost



Figure 29: Resulting electricity supply distribution (a) and overall yearly energy system costs (b) for sensitivity analysis on investment costs of hydrogen technologies

reductions of at least 76% for the hydrogen technologies. These reduction values must however be interpreted with caution and only in combination with the assumed costs for the technologies, as, like all other results on costs and generally all results of the optimisation, differences in the assumptions could considerably influence the conclusions.

5.2 Production of Synthetic Fuels at the Airport

As is shown in Figure 7 (on page 12), next to the energy technologies fulfilling heating, cooling and electricity demands, a pathway for the production of synthetic fuels is implemented in the superstructure. Optimisations are performed to analyse if integrating those technologies at the airport and selling the synthetic fuels would be economically beneficial for the airport. Based on the overall kerosene demand at Amsterdam Schiphol in 2017 of 3.6Mt [146] as well as the annual aircraft movements of 514,625 in 2017 [147] and 464,727 in 2023 [148], a kerosene demand of 3.25Mt is computed for 2023.

In the first analysis, it is assumed that the production of synthetic fuels at Amsterdam Schiphol would fulfil 1% of the annual kerosene demand with a constant production rate over the year of 3.71t/h. As the models [117] and associated cost functions [119] are defined for an order of magnitude of 1t/h, this production rate is in a valid range of magnitude. It must be noted that the models employed produce a fuel mix with only 1% of kerosene, as is shown in Table 3 (on page 18). For future research, it would be relevant to include synthetic fuel separation optimised for producing a maximal output share of kerosene. For this study, it is assumed that the fuel mix can be sold at the prices without taxes of the individual fuels, which are displayed in Table 3.

First, the pathway of producing synthetic fuels from biogasification is analysed. The profitability of producing synthetic fuels at airports depends on the investment costs of the technologies as well as the feedstock and output product prices. As both the price for the biomass and the price for synthetic fuels are uncertain, the total energy system costs are computed for a range of values. The biomass price varied from $5 \notin /t$ to $85 \notin /t$, as this price range represented most of the current biomass supply in the Netherlands [97]. For the synthetic fuel price obtained as a profit, a pricing factor is applied, indicating if 100%, 80% or 60% of the profit achievable at current fuel prices goes as revenue to the airport. Figure 30 displays the cost difference of the energy system for the different combinations of those two variables. The cost difference is defined as the difference in total costs, including investment costs, operating costs, and revenues from selling the synthetic fuels, with respect to the previously determined cost-optimal energy system, which has a cost of $9.43M \notin$. It thus indicates the extent to which the production of synthetic fuels would favour or burden the airport.

From Figure 30 it can be observed that, at the assumed costs, the production of synthetic fuels would be beneficial to the airport in most scenarios. In fact, only when assuming that only 60% of the fuel prices would arrive as profit for the airport and at conservative biomass prices of $65 \notin$ /t or higher, the airport would need to invest up to $5.0M\notin$ /y more than it would earn as revenue. When assuming the lowest biomass price of $5\notin$ /t, the airport would have a yearly revenue of $8.3M\notin$ /y. This possible revenue could even increase to $22.0M\notin$ /y if it is assumed that the airport would receive the full equivalent fuel price as a profit. The latter is composed of revenue of $34.5M\notin$ /y while having additional investment costs of $10.0M\notin$ /y and an additional operating cost of $2.5M\notin$ /y.



Figure 30: Difference in total energy system costs (for different biomass prices and synthetic fuel price factors) after addition of biogasification pathway compared to system with optimal cost without synthetic fuel production $(9.5M \notin /y)$

The second option implemented for the production of synthetic fuels is co-electrolysis from captured CO_2 , for example from emissions of nearby industries, as was described previously in Figure 7. As the CO_2 is assumed to have a zero cost, the only variable considered for the co-electrolysis pathway is the synthetic fuel pricing factor. The resulting cost difference with respect to the cost-optimal energy system without the production of synthetic fuels can be seen in Figure 31.



Figure 31: Difference in total energy system costs (for different synthetic fuel price factors) after addition of co-electrolysis pathway compared to system with optimal cost without synthetic fuel production $(9.43M \notin /y)$

It can be observed that the co-electrolysis pathway is significantly more expensive than the biogasification pathway. Instead of creating a profit for the airport, it increases its overall energy system costs by a factor higher than 5. When assuming that the full fuel price is received as a profit by the airport, an additional yearly cost of $42.6M \notin /y$ is required. When observing the cost distribution for this configuration, it can be seen that the energy resources in terms of grid electricity and natural gas, mostly required in the production of the synthetic fuels, have a yearly cost of $36.4M \notin /y$, while only a profit of $34.3M \notin /y$ is generated per year from selling the fuels. Moreover, the required technologies in the co-electrolysis pathway have an annualised investment cost of $32.3M \notin /y$, which does not yet include the additional costs from the supplementary capacities of combined cycles and PV.

Overall, the biogasification pathway displays promising results for producing synthetic fuels and, thus also, sustainable aviation fuel at the airport, whereas co-electrolysis does not seem to be a competitive option at the current technology development and costs. Due to the increasing importance of reducing emissions in aviation and, in consequence, the demand for sustainable aviation fuels, and with the current production capacities being not sufficient [149], it can be interesting for an airport to further investigate this option. For example, it could be analysed whether the production of sustainable aviation fuel from municipal waste could be profitable, as is proposed by Seiple at al. [150], especially considering the location of Amsterdam Schiphol within the densely populated Randstad region.

The two implemented pathways are to be built on for future analysis of the production of sustainable aviation fuels at the airport. First, the cost of the required additional infrastructure at the airport can be added to the cost of the technology, allowing to draw more precise conclusions on the profitability of the biogasification option. Further analysis could also include the investigation of different specific feedstocks available to the airport and their performance in producing sustainable aviation fuels. Furthermore, varying output rates, for example depending on the current electricity price, could be looked into, thus profiting from time periods of surplus electricity, as is proposed by Chen et al. [151]. Finally, the production at the airport site should be compared to different production locations, possibly closer to the feedstock, to investigate whether there is a benefit to produce the fuels at the airport.

5.3 Energy System Optimisation of Other European Airports

In the previous subsections, the energy system optimisation framework was applied to the case study of Amsterdam Schiphol airport. In this subsection, the analysis is extended to the five other European hub airports as presented in section 4.3. For all of the airports, the optimal configuration in terms of total costs is computed, both with all technologies in section 5.3.1 as well as with only technologies running on renewable resources in section 5.3.2. An overview of the technology capacities obtained in the optimised solutions for the different airports is included in Appendix C, which allows to verify the costs and the scale of the installed capacities.

5.3.1 Total Costs Optimised Solutions

As can be seen in section 4.3 (on page 25), the different European airports have different energy demands due to the airport building characteristics, the number of flights and passengers as well as the climate zone they are located in. Furthermore, the latter entails that they also have different potentials in terms of solar and wind power they can generate at their location. In consequence, it can be interesting to observe whether those different environments provoke different optimal energy system configurations when optimising for total costs. Thus, for all of the airports the energy system is optimised for minimal total costs and the resulting configurations are shown in Figure 32. For the optimisations, batteries are allowed as storage units, as these are observed to perform better in terms of cost from the analysis for Amsterdam Schiphol airport.



Figure 32: Comparison of energy system solutions for case-study European hub airports optimised for minimal total costs with batteries as available energy storage option

From the results, a most frequent configuration can be observed. For all airports except Oslo, the resulting energy system is composed of a combined cycle of natural gas, a relatively small boiler, a CO_2 network, PV, a small battery and electricity from the grid, with different relative sizes of these components depending on for example the solar radiation and the price of grid electricity. Oslo is the only airport with a different configuration for its energy system, not using a CO_2 network but a boiler, a combined cycle of natural gas, heat pumps and chillers for heating and cooling. This is due to the long airport buildings with respect to the overall surface area, making the required piping of the CO_2 network significantly more expensive.

From the key performance indicators, as displayed in Figure 33, it can be observed that Munich ranks lowest for the total costs and GWP per passenger with $0.23 \notin pax$ and $0.64 \text{kgCO}_2 \text{eq}/\text{pax}$ due to the large surface area per passenger, which needs to be supplied with energy. Zurich, for example, has a 38.9% lower surface area per passenger to supply with energy. Looking at the share of renewable energy, it can be seen that Oslo scores highest with 51.9% due to the large share of renewable energy in the Norwegian electricity grid. It is followed by the airport of Madrid with only 25.3%, which can slightly benefit of its large solar power potential due to the Mediterranean climate zone it is located in. Regarding self-sufficiency, Rome and Madrid perform best with 22.9% and 24.2% respectively, being the two cities with the highest solar power potential. On the other hand, Oslo has the lowest self-sufficiency of only 10.5% due to its low solar power potential. For the exergy efficiency of the airport, the airports are observed to perform similarly well. The airports with a high share of grid electricity perform best with exergy efficiencies of up to 63.4% in the case of Amsterdam.



Figure 33: Comparison of key performance indicators for energy systems of case-study European hub airports optimised for minimal total costs with batteries as available energy storage option

5.3.2 Total Costs Optimised Solutions with only Renewable Technologies

In contrast to the previous results, for the optimisations performed in this subsection, the use of natural gas boilers and combined cycles is prohibited. The energy system is again optimised for minimal total costs. The different configurations obtained can be seen in Figure 34.



Figure 34: Comparison of energy system solutions for case-study European hub airports optimised for minimal total costs with batteries as available energy storage option and no natural gas

The configurations for the different airports with only renewable technologies look very similar. For all airports except Oslo, a combination of the CO_2 network, the battery, PV and electricity from the grid is employed. Again, Oslo has a configuration different from the other airports by not using a battery. This is most probably due to the small amount of PV installed at the airport, as Oslo has only a low solar power potential, and consequently the small amount of intermittent electricity generated which can be saved from day to night. Between the other airports, the distribution of technologies looks similar.

The key performance indicators are shown in Figure 35. From the results, the lowest yearly GWP is observed for Oslo and Zurich with $0.06 \text{kgCO}_2 \text{eq}/\text{pax}$ and $0.01 \text{kgCO}_2 \text{eq}/\text{pax}$ respectively due to the large share of renewable electricity in the electricity mix of the countries. Overall, the GWP decreases due to the avoided emissions from the combined cycle. The renewable energy share is again very high for Oslo with 95.5%. Despite having a very low GWP, Zurich scores with 70.1% significantly lower in terms of renewable energy share, as the electricity mix of Switzerland includes a large share of nuclear power, which is not renewable. In terms of self-sufficiency, the airports located in the Mediterranean climate zone with high solar power potential score higher with a self-sufficiency of 61.0% for Rome, whereas Oslo, located in the Boreal climate zone, heavily depends on the electricity grid and only has a self-sufficiency of 12.1%. The same effect can be observed for the installed PV capacities: While in Madrid and Rome 0.25km^2 and 0.23km^2 are employed respectively, the configuration of Oslo only uses 0.04km^2 of PV. This also affects the installed battery capacity, which is largest for Madrid with 65.6 MWh, while in Oslo no battery is employed.



Figure 35: Comparison of key performance indicators for energy systems of case-study European hub airports optimised for minimal total costs with batteries as available energy storage option and no natural gas

Overall, two main configurations occur in the optimisations for the different airports. These are the combination of a CO_2 network, PV and a battery when only optimising with renewable technologies as well as an additional combined cycle from natural gas when allowing all technologies. Comparing these configurations to the different solutions in section 5.1.1, it can be seen that they coincide with the most frequent configurations in the Pareto fronts when optimising with and without batteries. Thus, it can be concluded that those energy system configurations are not only robust to different objectives while optimising, but also to different airport environments. Furthermore, in the Pareto analysis, it is seen that they do not only perform well in terms of costs, but also for their GWP and self-sufficiency, thus displaying an overall good performance for optimising the energy system of an airport.

Several points should be noted after interpreting the results for the different airports. First, for the five new airports, the specific energy demands are not validated by airport data, unlike for Amsterdam Schiphol. Only the overall order of magnitude is validated based on a set of 20 European airports. In addition, the building material and insulation might be different for all airports, which has a significant impact on the energy demand and is not yet taken into account by the energy demand model. Furthermore, the currently implemented configurations of the energy systems differ depending on the location, so that different pathways and transformation costs are needed for converting the respective airports' energy systems. Nevertheless, the model gives a good overview of feasible configurations depending on the location and airport configuration and provides a basis for future studies.

6 Conclusions

Airports are a significant source of emissions and energy consumption, with the yearly energy consumption of Amsterdam Schiphol being equivalent to the annual electricity consumption of 100,000 Dutch households. Consequently, optimising the energy systems of airports can contribute to reducing the aviation's environmental impact. The goal of this study was to develop an energy system model to analyse to what extent European airports, such as Amsterdam Schiphol airport, could benefit from the concept of renewable energy hubs in reducing their sustainability impact and increasing their energy supply self-sufficiency while fulfilling the varying energy demand profile from airport operations and infrastructure. It was found that a fully renewable energy system is possible at a limited total cost increase of approximately 20%. Self-sufficiencies of around 80% were observed to be viable in those energy system configurations combining photovoltaic cells (PV), a CO_2 network, batteries and electricity from the grid, which display robust results for varying airport environments.

The airport energy demand was estimated for the terminal and pier buildings and the airfield handling equipment. Combined, those two account for 61% of the energy consumption at Amsterdam Schiphol. Other important consumers are, for example, office buildings, hotels or hangars. The buildings' energy demand was split into heating, cooling, and ventilation; these were computed based on an estimate for the building envelope, hot water and remaining electricity demand. The energy consumption of the ground support equipment was determined based on reference data from actual ground support equipment manufacturers. The input parameters were defined as the number of flights and passengers, the building surface area and wall length, and weather data to make the model applicable to any airport. The demand was determined for an hourly profile, and the yearly energy demand was validated with actual airport energy demand data from Amsterdam Schiphol. The energy demand model allowed to obtain comparable results also for other European airports.

In the second step, an optimisation superstructure was constructed using the OSMOSE energy system optimisation framework developed at the Industrial Process and Energy Systems Engineering (IPESE) group of EPFL. The energy system can be optimised for the operating costs, investment costs and global warming potential (GWP) while respecting unit sizes, ensuring resource balance and obtaining a minimum energy requirement from the heat cascade. The energy demand data was clustered in typical days to reduce the computational load for mixed integer linear programming problems. The superstructure of possible energy technologies included multiple heating, cooling and electricity generation technologies, a battery and a hydrogen system for electricity storage as well as two pathways for producing synthetic fuels at the airport.

The optimisation framework was demonstrated by constructing different possible energy system configurations using a Pareto analysis for operating and investment costs. It was observed that the lowest overall costs of $9.3M\notin/y$, consisting of operating and annualised investment costs of $3.9M\notin/y$ and $5.5M\notin/y$ respectively, were obtained in a configuration employing a CO₂ network, PV, a combined cycle from natural gas and grid electricity as well as a battery for electricity storage. With this configuration, a scope-1 and -2 GWP of $24.8ktCO_2/y$, a renewable electricity share of 18.3% and a self-sufficiency of 17.8% were obtained. The latter can be significantly improved when energy storage technologies are used and when the combined cycle for heat and electricity production is avoided. When comparing the results from the optimisations for different airports, it was observed that most airports would employ energy system configurations similar to the ones proposed for Amsterdam Schiphol, showing the robustness of this combination of energy technologies.

A fully renewable energy system for the airport is possible at an overall cost increase of approximately 20%. However, the electricity grid still produces an essential amount of emissions, which must be decarbonized to avoid all scope-2 emissions. Results from different airports proved that an energy system consisting of a CO_2 network, PV, a battery and electricity from the grid is competitive to non-renewable energy systems in different environments. However, making the airport fully self-sufficient and independent of the grid is not competitive due to 248.3% higher total costs for this configuration. Self-sufficiencies of approximately 80% are achievable for the fully renewable energy system at total cost increases of only 60%.

Compared to the hydrogen system, battery storage is the most promising energy storage technology due to its higher round-trip efficiency and lower overall cost. The battery storage system is already used slightly for minimal-cost solutions and its usage increases in the case of possible future investment cost reductions. The hydrogen storage system would only be employed in minimal-cost configurations if its investment costs are reduced by 76%. A similar observation can be made for the mini wind turbines, which would only be employed as a local electricity production technology at investment cost reductions of approximately 76%, whereas the PV are already competitive at the current costs.

Finally, the models for producing synthetic fuels at the airport showed that those could be a possible future source of revenue for an airport. While the co-electrolysis pathway currently has too high investment and operating costs, the biogasification pathway from biomass would allow the production of synthetic fuels with profit, from synthetic fuels sales. At a biomass price of $5 \notin /t$ and while considering the maximal assumed revenue achievable from selling the synthetic fuels, more than $20M \notin /y$ could be obtained per year from selling 1% of the sustainable aviation fuel demand at Amsterdam Schiphol, more than twice as high as the minimal overall annual energy system costs. While those results look promising, further analyses of the possible installation of those technologies at the airport is needed to validate those conclusions.

7 Recommendations

Overall, this study displays the potential of this energy system optimisation framework. The framework allows to analyse and compare possible airport energy system configurations in varying environments. Due to the broad applicability of the model and the wide set of possible technologies, however, several limitations exist and need to be kept in mind when interpreting its results. First, the heating, ventilation and cooling demand is estimated based on a simplified building model and does, for example, not take into account specific building material or insulation. Furthermore, the demand of other airports than Amsterdam Schiphol is not validated with specific heating, cooling and electricity demands of these airports but only its magnitude is confirmed by comparison with other European airports. For the optimisation results, it must be noted that the energy system is not optimised for all key performance indicators, but a set of feasible configurations is analysed and their key performance indicators are displayed. Next, the optimisation results depend on assumptions for technology characteristics and costs, which might have a significant influence on the results, as can be seen in the sensitivity analysis. Moreover, the pathway for transitioning from the current towards more renewable and self-sufficient energy systems at different airports needs to be developed. Future work can address these gaps and continue to build on this model.

First, the method for estimating the energy demand could be extended to the remaining parts of the airport outside of the terminal and pier buildings and the ground support equipment. It could additionally include multiple actors present in the airport environment, such as hotels, office buildings, etc. Demand clusters could be identified around the airport site, where an integration of the energy systems could be beneficial due to the proximity of complementing demands. Next to optimising the energy system for a pre-defined demand, possible demand reduction pathways could also be investigated. Furthermore, closer cooperation with airports would allow to improve and validate the hourly profiles generated and to obtain more detailed data for the building envelope. Cooperation with airports would, furthermore, also allow to obtain more specific information on current energy systems and the characteristics of the technologies currently employed, which would allow the design of specific pathways for renewing and improving airport energy systems for the future. More technologies could be added to the superstructure, such as thermal storage, already employed at multiple Amsterdam Schiphol piers [152]. Exploring more long-term storage units, such as methane storage using power-to-methane processes [153] for reducing the dependency on the grid, could be interesting, as in this study especially daily storage units displayed good performances. Possible future technologies could be investigated and added to the superstructure, such as nuclear micro-reactors, which are suggested for the airport environment and are beneficial, especially in remote locations [154]. Promising technologies, such as the CO_2 network, could be further analysed for their use in the airport environment. Assuming future investment cost reductions for the hydrogen technologies, a combined battery and hydrogen storage system could also be investigated. Moreover, selling electricity from the airport to the grid could be considered, as well as investing in energy production and storage technologies in the airport region. The latter depends on the airport's location and could, for example, be near wind farms or hydropower plants.

Next, the potential of ground support equipment for an optimized energy system can be considered more closely. On the one hand, the effects of optimized ground support equipment operations, for example in the form of improved charging schedules, on charging demand and its distribution throughout the day can be considered. On the other hand, it can also be analysed to what extent using ground support equipment as a vehicle-to-grid technology, as is proposed in [50], would lead to a better performance of the overall energy system, for example by replacing battery capacities used for daily energy storage.

Finally, future analysis based on the model constructed could continue investigating in depth the option of producing synthetic fuels at the airport site. Additional resources, such as municipal waste from nearby cities [150] or CO_2 captured in the airport's ventilation system, could be analysed. The technology models implemented can be further adapted to the airport environment, and additional infrastructure costs and operational requirements can be considered. Finally, the fuel separation unit needs to be investigated in more detail and optimised for a maximal production of sustainable aviation fuels which can be sold at the airport.

References

- R. Florida, C. Mellander, and T. Holgersson. "Up in the air: the role of airports for regional economic development". In: *The Annals of Regional Science* 54.1 (2015), pp. 197–214. DOI: 10.1007/s00168-014-0651-z.
- [2] Eurocontrol. A Place to Stand: Airports in the European Air Network. EUROCONTROL Trends in Air Traffic 3. ISSN: 1993-1425. 2007.
- [3] Airports Council International Europe. Passenger traffic reaches nearly 95% of pre-pandemic levels in 2023. 2024. URL: https://www.aci-europe.org/media-room/477-passenger-traffic-reaches-nearly-95-of-pre-pandemic-levels-in-2023.html (visited on 01/17/2025).
- [4] N. J. Ashford, S. Mumayiz, and P. H. Wright. Airport Engineering: Planning, Design, and Development of 21st Century Airports. 4th ed. Hoboken, New Jersey, USA: Wiley, 2011. ISBN: 978-0-470-39855-5.
- [5] Schiphol. Annual Report 2023. en. Schiphol, The Netherlands, 2024.
- KLM. CO2 emission and compensation price per destination. 2019. URL: https://img.static-kl. com/m/500c345c9ba5a839/original/CO2-emission-and-compensation-price-per-destination-2019.pdf (visited on 01/19/2025).
- [7] Centraal Bureau voor de Statistiek. Energy consumption private dwellings; type of dwelling and regions.
 2023. URL: https://www.cbs.nl/en-gb/figures/detail/81528ENG (visited on 01/19/2025).
- [8] Centraal Bureau voor de Statistiek. Households today. 2024. URL: https://www.cbs.nl/en-gb/ visualisations/dashboard-population/households/households-today (visited on 01/17/2025).
- [9] Centraal Bureau voor de Statistiek. Inwoners per gemeente. 2024. URL: https://www.cbs.nl/nlnl/visualisaties/dashboard-bevolking/regionaal/inwoners (visited on 01/19/2025).
- [10] Fraport AG Frankfurt Airport Services Worldwide. Umwelterklärung 2022. Frankfurt am Main, Germany, 2023.
- [11] Flughafen München GmbH. Flughafen München Integrierter Bericht 2022. München-Flughafen, Germany, 2023.
- [12] Flughafen Zürich AG. Integrierter Bericht 2022. Zürich-Flughafen, Switzerland, 2023.
- [13] Flughafen Wien AG. Umwelterklärung 2021 Wien. Wien-Flughafen, Austria, 2022.
- [14] Aeroporti di Roma. 2022 Integral Annual Report. Roma, Italy, 2023.

- [15] SEA Group Milan Airports. Sustainability Report 2022. Aeroporto Milano Linate, Italy, 2024.
- [16] Avinor. Annual and Sustainability Report 2023. Oslo, Norway, 2024.
- [17] Finavia. Annual and Responsibility Report 2023. 2024.
- [18] iGA Istanbul Airport. Sustainability Report 2021-2022. Istanbul, Turkey, 2022.
- [19] Groupe ADP. Traffic Paris Aéroport and Groupe ADP airports. 2024. URL: https://www.parisaeroport. fr/en/group/finance/investor-relations/traffic (visited on 06/20/2024).
- [20] Aeroports de Paris. Universal Registration Document 2023. Paris, France, 2024.
- [21] AENA. Consolidated Management Report 2023. Madrid, Spain, 2024.
- [22] Gatwick Airport. Progress on our 2020 goals. Gatwick, UK.
- [23] ANA Environment and Sustainability Office. ANA Sustainability Report ANA 2022 Together for Positive Mobility. Lisboa, Portugal, 2023.
- [24] Athens International Airport. Annual & Sustainability Report. Athens, Greece, 2024.
- [25] Manchester Airports Group. Manchester Airport Corporate Social Responsibility Progress Update 2022/2023. Manchester, UK, 2023.
- [26] Manchester Airports Group. Traffic statistics / Manchester Airport. 2024. URL: https://www.manchesterairport. co.uk/about-us/publications/traffic-statistics/ (visited on 06/20/2024).
- [27] Manchester Airports Group. Stansted Airport Corporate Social Responsibility Progress Update 2022/2023. Manchester, UK, 2023.
- [28] Manchester Airports Group. December's strong performance rounds off 2022. 2023. URL: https:// mediacentre.stanstedairport.com/london-stansted-rounds-off-2022-with-strong-passengerperformance-in-december/ (visited on 06/20/2024).
- [29] Copenhagen Airports. Annual Report 2023. Kastrup, Denmark, 2024.
- [30] Copenhagen Airports. Strong growth at Copenhagen Airport: Close to 27 million travellers. 2024. URL: https://www.cph.dk/en/about-cph/investor/traffic-statistics/2024/01/strong%20growth% 20at%20copenhagen%20airport%20close%20to%2027%20million%20travellers (visited on 06/20/2024).
- [31] Brussels Airport Company. Environmental Report 2019. Zaventem, Belgium, 2019.
- [32] Swedawia Airports. Annual & Sustainability Report 2023. Stockholm Arlanda, Sweden, 2024.
- [33] International Energy Agency. Europe Sources of electricity generation. 2022. URL: https://www.iea. org/regions/europe/electricity (visited on 01/17/2025).
- [34] L. Middelhauve. "On the role of districts as renewable energy hubs". PhD thesis. Lausanne, Switzerland: EPFL, 2022.
- [35] M.-J. Yoo, L. Lessard, M. Kermani, et al. "OsmoseLua An Integrated Approach to Energy Systems Integration with LCIA and GIS". In: *Computer Aided Chemical Engineering* 37 (2015). Ed. by K. V. Gernaey, J. K. Huusom, and R. Gani, pp. 587–592. DOI: https://doi.org/10.1016/B978-0-444-63578-5.50093-1.
- [36] M. Schaafsma. "Airports and Cities in Networks". In: disP The Planning Review 39.154 (2003), pp. 28– 36. DOI: 10.1080/02513625.2003.10556854.
- [37] Airports Council International World. Top 10 busiest airports in the world shift with the rise of international air travel demand. 2024. URL: https://aci.aero/2024/04/14/top-10-busiest-airports-inthe-world-shift-with-the-rise-of-international-air-travel-demand/ (visited on 06/29/2024).
- [38] D. Schaar and L. Sherry. "Analysis of airport stakeholders". In: 2010 Integrated Communications, Navigation, and Surveillance Conference Proceedings. 2010. DOI: 10.1109/ICNSURV.2010.5503233.
- [39] M. Janić. Landside Accessibility of Airports: Analysis, Modelling, Planning, and Design. Cham, Switzerland: Springer International Publishing, 2019. ISBN: 978-3-319-76149-7.
- [40] C. G. Corlu, R. de la Torre, A. Serrano-Hernandez, et al. "Optimizing Energy Consumption in Transportation: Literature Review, Insights, and Research Opportunities". In: *Energies* 13.5 (2020), p. 1115. DOI: 10.3390/en13051115.
- [41] F. Greer, J. Rakas, and A. Horvath. "Airports and environmental sustainability: a comprehensive review". In: *Environmental Research Letters* 15.10 (2020), p. 103007. DOI: 10.1088/1748-9326/abb42a.
- [42] S. Ortega Alba. "Characterization, analysis and optimization of energy demand patterns in airports". PhD thesis. Santander, Spain: University of Cantabria, 2017.

- [43] L. Lin, X. Liu, T. Zhang, et al. "Energy consumption index and evaluation method of public traffic buildings in China". In: Sustainable Cities and Society 57 (2020), p. 102132. DOI: 10.1016/j.scs.2020. 102132.
- [44] B. Li, W. Zhang, J. Wang, et al. "Research and Analysis on Energy Consumption Features of Civil Airports". In: *IOP Conference Series: Earth and Environmental Science* 94.1 (2017), p. 012134. DOI: 10.1088/1755-1315/94/1/012134.
- [45] X. Gu, J. Xie, C. Huang, et al. "Prediction of the spatiotemporal passenger distribution of a large airport terminal and its impact on energy simulation". In: Sustainable Cities and Society 78 (2022), p. 103619. DOI: 10.1016/j.scs.2021.103619.
- [46] T. de Rubeis, I. Nardi, D. Paoletti, et al. "Multi-year consumption analysis and innovative energy perspectives: The case study of Leonardo da Vinci International Airport of Rome". In: Energy Conversion and Management 128 (2016), pp. 261–272. DOI: 10.1016/j.enconman.2016.09.076.
- [47] G. Xianliang, X. Jingchao, L. Zhiwen, et al. "Analysis to energy consumption characteristics and influencing factors of terminal building based on airport operating data". In: Sustainable Energy Technologies and Assessments 44 (2021), p. 101034. DOI: 10.1016/j.seta.2021.101034.
- [48] O. F. Yildiz, M. Yilmaz, and A. Celik. "Reduction of energy consumption and CO2 emissions of HVAC system in airport terminal buildings". In: *Building and Environment* 208 (2022), p. 108632. DOI: 10. 1016/j.buildenv.2021.108632.
- [49] G. Lodewijks, Y. Cao, N. Zhao, et al. "Reducing CO Emissions of an Airport Baggage Handling Transport System Using a Particle Swarm Optimization Algorithm". In: *IEEE Access* 9 (2021), pp. 121894– 121905. DOI: 10.1109/ACCESS.2021.3109286.
- [50] X. Liu, M. Li, X. Liu, et al. "Quantifying energy flexibility potential of ground electric vehicles in an airport with real behavior data". In: Sustainable Cities and Society 105 (2024), p. 105331. DOI: 10.1016/j.scs.2024.105331.
- [51] D. Sigler, Q. Wang, Z. Liu, et al. "Route optimization for energy efficient airport shuttle operations A case study from Dallas Fort worth International Airport". In: Journal of Air Transport Management 94 (2021), p. 102077. DOI: 10.1016/j.jairtraman.2021.102077.
- [52] M. Zoutendijk and M. Mitici. "Fleet scheduling for electric towing of aircraft under limited airport energy capacity". In: *Energy* 294 (2024), p. 130924. DOI: 10.1016/j.energy.2024.130924.
- [53] Schiphol. New electric equipment for parked planes at Schiphol. 2023. URL: https://news.schiphol. com/new-electric-equipment-for-parked-planes-at-schiphol/ (visited on 01/13/2025).
- [54] R. Radhakrishnan. Analysis on the energy distribution and feasibility of positive energy district at Stavanger airport. Master thesis. Stavanger, Norway: University of Stavanger, 2020.
- [55] R. Zafar, A. Mahmood, S. Razzaq, et al. "Prosumer based energy management and sharing in smart grid". In: *Renewable and Sustainable Energy Reviews* 82 (2018), pp. 1675–1684. DOI: 10.1016/j.rser. 2017.07.018.
- [56] Y. Zhang, P. Si, Y. Feng, et al. "Operation strategy optimization of BCHP system with thermal energy storage: A case study for airport terminal in Qingdao, China". In: *Energy and Buildings* 154 (2017), pp. 465–478. DOI: 10.1016/j.enbuild.2017.08.059.
- [57] Y. Tian, X. Zhang, and N. Balta-Ozkan. "Multi-Energy Systems in Green Airports". In: CIRED 2021 -The 26th International Conference and Exhibition on Electricity Distribution. Vol. 2021. 2021, pp. 2043– 2047. DOI: 10.1049/icp.2021.1561.
- [58] S. Jin and Y. Li. "Analyzing the performance of electricity, heating, and cooling supply nexus in a hybrid energy system of airport under uncertainty". In: *Energy* 272 (2023), p. 127138. DOI: 10.1016/j.energy. 2023.127138.
- [59] Q. Jia, H.-q. Hu, and J.-l. Yu. "The model construction in airport renewable energy system scale planning based in Multi-Criteria Decision Making". In: *Energy Reports* 7 (2021), pp. 92–98. DOI: 10.1016/j.egyr. 2021.08.068.
- [60] S. Baek, H. Kim, and H. J. Chang. "Optimal Hybrid Renewable Airport Power System: Empirical Study on Incheon International Airport, South Korea". In: Sustainability 8.6 (2016), p. 562. DOI: 10.3390/ su8060562.
- [61] M. Prussi, L. Laveneziana, D. Misul, et al. "A model-based approach to long-term energy planning: the case-study of the Turin Airport". In: *Journal of Physics: Conference Series* 2648.1 (2023), p. 012034. DOI: 10.1088/1742-6596/2648/1/012034.

- [62] S. Thiem, V. Danov, M. Metzger, et al. "Project-level multi-modal energy system design Novel approach for considering detailed component models and example case study for airports". In: *Energy* 133 (2017), pp. 691–709. DOI: 10.1016/j.energy.2017.05.159.
- [63] E. Cardona, P. Sannino, A. Piacentino, et al. "Energy saving in airports by trigeneration. Part II: Short and long term planning for the Malpensa 2000 CHCP plant". In: *Applied Thermal Engineering* 26.14 (2006), pp. 1437–1447. DOI: 10.1016/j.applthermaleng.2006.01.020.
- [64] B. Kılkış and Ş. Kılkış. "New exergy metrics for energy, environment, and economy nexus and optimum design model for nearly-zero exergy airport (nZEXAP) systems". In: *Energy* 140 (2017), pp. 1329–1349. DOI: 10.1016/j.energy.2017.04.129.
- [65] F. Belfiore. "District heating and cooling systems to integrate renewable energy in urban areas". PhD thesis. Lausanne, Switzerland: EPFL, 2021.
- [66] E. Cardona, A. Piacentino, and F. Cardona. "Energy saving in airports by trigeneration. Part I: Assessing economic and technical potential". In: *Applied Thermal Engineering* 26.14 (2006), pp. 1427–1436. DOI: 10.1016/j.applthermaleng.2006.01.019.
- [67] M. C. Kirca, A. McGordon, and T. Q. Dinh. "Multi-Input Multi-Output Model of Airport Infrastructure for Reducing CO2 Emissions". In: 2020 IEEE Vehicle Power and Propulsion Conference (VPPC). ISSN: 1938-8756. Nov. 2020, pp. 1–6. DOI: 10.1109/VPPC49601.2020.9330949.
- [68] Y. Xiang, H. Cai, J. Liu, et al. "Techno-economic design of energy systems for airport electrification: A hydrogen-solar-storage integrated microgrid solution". In: Applied Energy 283 (2021), p. 116374. DOI: 10.1016/j.apenergy.2020.116374.
- [69] H. Zhao, Y. Xiang, Y. Shen, et al. "Resilience Assessment of Hydrogen-Integrated Energy System for Airport Electrification". In: *IEEE Transactions on Industry Applications* 58.2 (2022), pp. 2812–2824. DOI: 10.1109/TIA.2021.3127481.
- [70] Schiphol. Energiebeoordeling SNBV 2023. Schiphol, The Netherlands, 2024.
- [71] Schiphol. Energiebalansen Asset Management 2023. Schiphol, The Netherlands, 2024.
- [72] Swiss Society of Engineers and Architects. SIA 2024 Raumdatenblätter. 2024. URL: https://www.sia. ch/de/cms/dienstleistungen/normenundordnungen?item=7979#7956 (visited on 12/03/2024).
- [73] N. Djongyang, R. Tchinda, and D. Njomo. "Thermal comfort: A review paper". In: Renewable and Sustainable Energy Reviews 14.9 (2010), pp. 2626–2640. DOI: 10.1016/j.rser.2010.07.040.
- [74] OpenStreetMap contributors. *Planet dump retrieved from https://planet.osm.org.* 2017. URL: https://www.openstreetmap.org (visited on 12/03/2024).
- [75] ASHRAE. ANSI/ASHRAE/ASHE Addendum a to ANSI/ASHRAE/ASHE Standard 170-2017: Ventilation of Health Care Facilities. ISSN: 1041-2336. Atlanta, USA, 2020.
- [76] The Engineering ToolBox. Air Thermophysical Properties. 2003. URL: https://www.engineeringtoolbox. com/air-properties-d_156.html (visited on 12/03/2024).
- [77] C. Agudelo-Vera, S. Avvedimento, J. Boxall, et al. "Drinking Water Temperature around the Globe: Understanding, Policies, Challenges and Opportunities". In: Water 12.4 (2020), p. 1049. DOI: 10.3390/ w12041049.
- B. Lévesque, M. Lavoie, and J. Joly. "Residential water heater temperature: 49 or 60 degrees Celsius?" In: The Canadian Journal of Infectious Diseases 15.1 (2004), pp. 11–12. DOI: 10.1155/2004/109051.
- [79] The Engineering ToolBox. Water Thermophysical Properties. 2003. URL: https://www.engineeringtoolbox. com/water-thermal-properties-d_162.html (visited on 12/03/2024).
- [80] J. Turner. Are airports presenting a wide enough food and drink offering? 2020. URL: https://www. airport-technology.com/features/food-and-drinks-offering-at-airports/ (visited on 12/03/2024).
- [81] L. Wierda, W. Li, and R. Kemna. Ventilation Units. Ecodesign Impact Accounting Overview Report 2023. European Commission, DG Energy, 2023.
- [82] The Engineering ToolBox. Air Change Rates in typical Rooms and Buildings. 2005. URL: https://www. engineeringtoolbox.com/air-change-rate-room-d_867.html (visited on 12/03/2024).
- [83] KLM. Selected Tripdata Ground Support Equipment August 2024. 2024.
- [84] Orientitan. Products / Ground Support Equipment. URL: http://uld-equipment.com/products.html (visited on 01/08/2025).
- [85] K. Timmermans. Electric Ground Support Equipment at Airports. Master thesis. Delft, Netherlands: TU Delft, 2023.

- [86] M. Schäfer, M. Strohmeier, V. Lenders, et al. "Bringing up OpenSky: A large-scale ADS-B sensor network for research". In: *IPSN-14 Proceedings of the 13th International Symposium on Information Processing* in Sensor Networks. 2014, pp. 83–94. DOI: 10.1109/IPSN.2014.6846743.
- [87] Schiphol. Traffic Review 2023. 2024. URL: https://www.annualreportschiphol.com/2023/summary (visited on 01/26/2025).
- [88] P. Zippenfenig. "Open-Meteo.com Weather API". In: Zenodo (2023). DOI: https://doi.org/10.5281/ ZENOD0.7970649.
- [89] M. v. d. Sluys and P. v. Kan. SolTrack: a free, fast and accurate routine to compute the position of the Sun. 2022. DOI: 10.48550/arXiv.2209.01557.
- [90] B. Burger and Fraunhofer-Institut für Solare Energiesysteme ISE. *Energy-Charts.* 2024. URL: https://www.energy-charts.info/index.html (visited on 01/05/2025).
- [91] European Environment Agency. Greenhouse gas emission intensity of electricity generation, country level. 2024. URL: https://www.eea.europa.eu/en/analysis/indicators/greenhouse-gas-emissionintensity-of-1/greenhouse-gas-emission-intensity-of-electricity-generation-countrylevel (visited on 01/05/2025).
- [92] T. Schütz, M. H. Schraven, M. Fuchs, et al. "Comparison of clustering algorithms for the selection of typical demand days for energy system synthesis". In: *Renewable Energy* 129 (2018), pp. 570–582. DOI: 10.1016/j.renene.2018.06.028.
- [93] D. Lepour, J. Loustau, C. Terrier, et al. "REHO: A Decision Support Tool for Renewable Energy Communities". In: Journal of Open Source Software 9 (2024), p. 6734. DOI: 10.21105/joss.06734.
- [94] Investing. Dutch TTF Natural Gas Futures Historical Prices. 2023. URL: https://www.investing.com/ commodities/dutch-ttf-gas-c1-futures-historical-data (visited on 10/09/2024).
- [95] S. Chen, A. Bouteska, T. Sharif, et al. "The Russia–Ukraine war and energy market volatility: A novel application of the volatility ratio in the context of natural gas". In: *Resources Policy* 85 (2023), p. 103792. DOI: 10.1016/j.resourpol.2023.103792.
- [96] Tata Steel. Tata Steel IJmuiden. 2025. URL: https://www.tatasteelnederland.com/en/about-tatasteel/tata-steel-ijmuiden (visited on 01/11/2025).
- [97] S2Biom. Tools for biomass chains. 2012. URL: https://s2biom.wenr.wur.nl/web/guest/home (visited on 01/11/2025).
- [98] A. Anurag, J. Zhang, J. Gwamuri, et al. "General Design Procedures for Airport-Based Solar Photovoltaic Systems". In: *Energies* 10.8 (2017), p. 1194. DOI: 10.3390/en10081194.
- [99] A. S. Wallerand. "Integration of solar energy with industrial processes". PhD thesis. Lausanne, Switzerland: EPFL, 2018.
- [100] A. De Vita, P. Capros, L. Paroussos, et al. EU reference scenario 2020: energy, transport and GHG emissions : trends to 2050. Luxembourg, Luxembourg: Publications Office of the European Union, 2021. ISBN: 978-92-76-39356-6.
- [101] L. Cuadra, I. Ocampo-Estrella, E. Alexandre, et al. "A study on the impact of easements in the deployment of wind farms near airport facilities". In: *Renewable Energy* 135 (2019), pp. 566–588. DOI: 10.1016/j.renene.2018.12.038.
- [102] J. Bukala, K. Damaziak, H. R. Karimi, et al. "Modern small wind turbine design solutions comparison in terms of estimated cost to energy output ratio". In: *Renewable Energy* 83 (2015), pp. 1166–1173. DOI: 10.1016/j.renene.2015.05.047.
- [103] A. Rosato, A. Perrotta, and L. Maffei. "Commercial Small-Scale Horizontal and Vertical Wind Turbines: A Comprehensive Review of Geometry, Materials, Costs and Performance". In: *Energies* 17.13 (2024), p. 3125. DOI: 10.3390/en17133125.
- [104] International Renewable Energy Agency. Electricity Storage and Renewables: Costs and Markets to 2030.
 Abu Dhabi, UAE: International Renewable Energy Agency, 2017. ISBN: 978-92-9260-038-9.
- [105] S. Fazlollahi. "Decomposition optimization strategy for the design and operation of district energy systems". PhD thesis. Lausanne, Switzerland: EPFL, 2014.
- [106] G. Wernet, C. Bauer, B. Steubing, et al. "The ecoinvent database version 3 (part I): overview and methodology". In: *The International Journal of Life Cycle Assessment* 21 (2016), pp. 1218–1230. DOI: 10.1007/s11367-016-1087-8.
- [107] H. Pieper, I. Krupenski, W. Brix Markussen, et al. "Method of linear approximation of COP for heat pumps and chillers based on thermodynamic modelling and off-design operation". In: *Energy* 230 (2021), p. 120743. DOI: 10.1016/j.energy.2021.120743.

- [108] S. Henchoz, C. Weber, F. Maréchal, et al. "Performance and profitability perspectives of a CO2 based district energy network in Geneva's City Centre". In: *Energy* 85 (2015), pp. 221–235. DOI: 10.1016/j. energy.2015.03.079.
- [109] R.-A. Suciu. "Fifth generation district energy systems for low carbon cities". PhD thesis. Lausanne, Switzerland: EPFL, 2019.
- [110] C. Weber and D. Favrat. "Conventional and advanced CO2 based district energy systems". In: Energy 35.12 (2010), pp. 5070–5081. DOI: 10.1016/j.energy.2010.08.008.
- [111] J. Ivy. Summary of Electrolytic Hydrogen Production: Milestone Completion Report. Colorado, USA: National Renewable Energy Laboratory, 2004.
- [112] F. M. Sapountzi, J. M. Gracia, C. J. (.-J. Weststrate, et al. "Electrocatalysts for the generation of hydrogen, oxygen and synthesis gas". In: *Progress in Energy and Combustion Science* 58 (2017), pp. 1– 35. DOI: 10.1016/j.pecs.2016.09.001.
- [113] X. Wei, S. Sharma, A. Waeber, et al. "Comparative life cycle analysis of electrolyzer technologies for hydrogen production: Manufacturing and operations". In: *Joule* 8.12 (2024), pp. 3347–3372. DOI: 10. 1016/j.joule.2024.09.007.
- [114] H. Nami, O. B. Rizvandi, C. Chatzichristodoulou, et al. "Techno-economic analysis of current and emerging electrolysis technologies for green hydrogen production". In: *Energy Conversion and Management* 269 (2022), p. 116162. DOI: 10.1016/j.enconman.2022.116162.
- [115] X. Wei, S. Sharma, J. Van herle, et al. "Analysis and optimization of solid oxide fuel cell system with anode and cathode off gas recirculation". In: *Renewable and Sustainable Energy Reviews* 208 (2025), p. 115015. DOI: 10.1016/j.rser.2024.115015.
- [116] V. Cigolotti and M. Genovese. Stationary fuel cell applications: Current and future technologies Costs, Performance and Potential. IEA Technology Collaboration Programme on Advanced Fuel Cells, 2021.
- [117] European Commission. Directorate General for Energy., EnergyVille., Vito., et al. AIDRES, "Advancing industrial decarbonization by assessing the future use of renewable energies in industrial processes": assessment and geo mapping of renewable energy demand for technological paths towards carbon neutrality of EU energy intensive industries : methodology and results in support to the EU industrial plants database. Luxembourg, Luxembourg: Publications Office of the European Union, 2023. ISBN: 978-92-68-07186-1.
- [118] C. Rubio-Maya, J. Uche-Marcuello, A. Martínez-Gracia, et al. "Design optimization of a polygeneration plant fuelled by natural gas and renewable energy sources". In: *Applied Energy* 88.2 (2011), pp. 449–457. DOI: 10.1016/j.apenergy.2010.07.009.
- [119] J. Granacher. "Decision support for the design of integrated biorefineries". PhD thesis. Lausanne, Switzerland: EPFL, 2023.
- [120] A. M. Petersen, F. Chireshe, O. Okoro, et al. "Evaluating refinery configurations for deriving sustainable aviation fuel from ethanol or syncrude". In: *Fuel Processing Technology* 219 (2021), p. 106879. DOI: 10.1016/j.fuproc.2021.106879.
- [121] A. Allangawi, E. F. H. Alzaimoor, H. H. Shanaah, et al. "Carbon Capture Materials in Post-Combustion: Adsorption and Absorption-Based Processes". In: C - Journal of Carbon Research 9.1 (2023), p. 17. DOI: 10.3390/c9010017.
- [122] X. Zhu, W. Xie, J. Wu, et al. "Recent advances in direct air capture by adsorption". In: Chemical Society Reviews 51.15 (2022), pp. 6574–6651. DOI: 10.1039/D1CS00970B.
- [123] M. G. Plaza, S. Martínez, and F. Rubiera. "CO2 Capture, Use, and Storage in the Cement Industry: State of the Art and Expectations". In: *Energies* 13.21 (2020), p. 5692. DOI: 10.3390/en13215692.
- [124] W. Zhang, H. Liu, Y. Sun, et al. "Parametric study on the regeneration heat requirement of an aminebased solid adsorbent process for post-combustion carbon capture". In: Applied Energy 168 (2016), pp. 394–405. DOI: 10.1016/j.apenergy.2016.01.049.
- [125] A. D. Celebi. "Methodology for the identification of promising integrated biorefineries". PhD thesis. Lausanne, Switzerland: EPFL, 2019.
- [126] D. R. Nhuchhen, P. Basu, and B. Acharaya. "A Comprehensive Review on Biomass Torrefaction". In: International Journal of Renewable Energy & Biofuels 2014 (2014), p. 506376. DOI: 10.5171/2014. 506376.
- [127] S. Bube, N. Bullerdiek, S. Voß, et al. "Kerosene production from power-based syngas A technical comparison of the Fischer-Tropsch and methanol pathway". In: *Fuel* 366 (2024), p. 131269. DOI: 10. 1016/j.fuel.2024.131269.

- [128] European Commission. Weekly Oil Bulletin. 2023. URL: https://energy.ec.europa.eu/data-andanalysis/weekly-oil-bulletin_en (visited on 12/17/2023).
- [129] ECHEMI. Kerosene International Price. 2024. URL: https://www.echemi.com/pip/kerosene-pid_ Rock27024.html (visited on 12/19/2024).
- [130] A. Al-Moubaraki and I. Obot. "Corrosion Challenges in Petroleum Refinery Operations: Sources, Mechanisms, Mitigation, and Future Outlook". In: *Journal of Saudi Chemical Society* 25 (2021), p. 101370. DOI: 10.1016/j.jscs.2021.101370.
- [131] D. Dardor, D. Flórez-Orrego, C. Terrier, et al. "ROSMOSE: A web-based decision support tool for the design and optimization of industrial and urban energy systems". In: *Energy* 304 (2024), p. 132182. DOI: 10.1016/j.energy.2024.132182.
- [132] F. Maréchal and B. Kalitventzeff. "Process integration: Selection of the optimal utility system". In: Computers & Chemical Engineering 22 (1998), S149–S156. DOI: 10.1016/S0098-1354(98)00049-0.
- [133] I. Hannula and E. Kurkela. Liquid transportation fuels via large-scale fluidised-bed gasification of lignocellulosic biomass. Espoo, Finland: VTT Technical Research Centre of Finland, 2013. ISBN: 978-951-38-7978-5.
- [134] Greenhouse Gas Protocol (WRI & WBCSD). Corporate Value Chain (Scope 3) Accounting and Reporting Standard. 2011.
- [135] J. Szargut. Exergy Method Technical and Ecological Applications. Southampton, UK: WIT Press, 2005. ISBN: 978-1-85312-753-3.
- [136] O. Ozgener and L. Ozgener. "Exergy and reliability analysis of wind turbine systems: A case study". In: *Renewable and Sustainable Energy Reviews* 11.8 (2007), pp. 1811–1826. DOI: 10.1016/j.rser.2006. 03.004.
- [137] Schiphol. Facts about Amsterdam Airport Schiphol. 2023. URL: https://www.schiphol.nl/en/routedevelopment/page/amsterdam-airport-schiphol-airport-facts/ (visited on 01/12/2025).
- [138] KLM. Network and alliances. 2025. URL: https://www.klm.ch/en/information/corporate/networkalliances (visited on 01/12/2025).
- [139] European Environment Agency. Observed climate zones in the period 1975-1995 (left) and 1996-2016 (right). 2019. URL: https://www.eea.europa.eu/en/analysis/maps-and-charts/observedclimate-zones-in-the (visited on 01/08/2025).
- [140] A. Ceglar, M. Zampieri, A. Toreti, et al. "Observed Northward Migration of Agro-Climate Zones in Europe Will Further Accelerate Under Climate Change". In: *Earth's Future* 7.9 (2019), pp. 1088–1101. DOI: 10.1029/2019EF001178.
- [141] Sociedad Mercantil Estatal para la Gestión de la Innovación y las Tecnologías Turísticas, S.A.M.P. Air traffic. 2023. URL: https://www.dataestur.es/en/transport/air-traffic-aena/ (visited on 01/08/2025).
- [142] Flughafen München GmbH. Traffic Report 2023. München-Flughafen, Germany, 2024.
- [143] Avinor. Traffic Statistics. 2023. URL: https://avinor.no/en/corporate/about-us/statistics/ traffic-statistics (visited on 01/08/2025).
- [144] Aeroporti di Roma. Traffic data. 2023. URL: https://www.adr.it/web/aeroporti-di-roma-en/bsntraffic-data (visited on 01/08/2025).
- [145] Zurich Airport. Medienmitteilung Archive. 2023. URL: https://newsroom.flughafen-zuerich.ch/en/ ?h=1&t=medienmitteilung (visited on 01/08/2025).
- [146] R. Terwel and J. Kerkhoven. Carbon Neutral Aviation With Current Engine Technology: The Take-Off of Synthetic Kerosene Production in the Netherlands - The Story and the Report. Quintel Intelligence B.V. & Kalavasta B.V., 2018.
- [147] Schiphol. Traffic Review 2017. Schiphol, The Netherlands, 2018.
- [148] Schiphol. Traffic and transport figures per month. 2024. URL: https://www.schiphol.nl/en/schipholgroup/transport-and-traffic-statistics/ (visited on 01/05/2025).
- B. Wang, Z. J. Ting, and M. Zhao. "Sustainable aviation fuels: Key opportunities and challenges in lowering carbon emissions for aviation industry". In: Carbon Capture Science & Technology 13 (2024), p. 100263. DOI: 10.1016/j.ccst.2024.100263.
- [150] T. Seiple, Y. Jiang, L. J. Snowden-Swan, et al. "Cost-Effective Opportunities to Produce Sustainable Aviation Fuel from Low-Cost Wastes in the U.S". In: ACS Sustainable Chemistry & Engineering 11.33 (2023), pp. 12326–12335. DOI: 10.1021/acssuschemeng.3c02147.

- [151] G. Chen, O. Lugovoy, P. Piris-Cabezas, et al. Intermittent production of electricity-based synthetic jet fuel as a demand-side management strategy for grid decarbonization. SSRN Scholarly Paper. Rochester, NY, 2024. DOI: 10.2139/ssrn.5029916.
- [152] Schiphol. Thermal Energy Storage at the Airport. URL: https://www.schiphol.nl/en/sustainability/ at-the-airport/sustainable-climate-control-at-the-airport/ (visited on 01/28/2025).
- [153] L. Wang, J. Dull, F. Marechal, et al. "Trade-off designs and comparative exergy evaluation of solidoxide electrolyzer based power-to-methane plants". In: 44.19 (2019), pp. 9529–9543. DOI: 10.1016/j. ijhydene.2018.11.151.
- [154] J. Buongiorno. Flexible Siting Criteria and Staff Minimization for Micro-Reactors. English. Cambridge, MA, USA: Massachusetts Inst. of Technology (MIT), 2022. DOI: 10.2172/1898105.
- [155] Nowtricity. CO2 emissions per kWh in Norway. 2024. URL: https://www.nowtricity.com/country/ norway/ (visited on 12/20/2024).
- [156] Nowtricity. CO2 emissions per kWh in Switzerland. 2024. URL: https://www.nowtricity.com/country/ switzerland/ (visited on 12/20/2024).

Appendices

A Verification of the Optimisation Framework

The optimisation framework developed is verified using a diverse set of tests, which confirm the right implementation of the units and the correct functioning of the constraints described in section 3.4. From every optimisation, the parameters used, the decision variables and the unit results, including their respective cost and emissions, are saved to ensure the traceability of the results and to allow model verification. The main verification tests are described in this section.

First, the technological definitions of the units are verified. The specific values of the in and outgoing resource streams, if possible, and the specific cost and emission values at unit load are calculated by hand and compared with the output optimisation data. The ratios between specific cost and resource stream magnitude are compared with values from the literature to validate the implemented technology characteristics. Furthermore, by implementing dummy demands for the units in OSMOSE, the correct functioning of the units within the optimisation environment is confirmed. For models depending on varying input profiles, such as the solar radiation for the PV, it is checked whether the output streams correctly reflect the input profile.

Furthermore, the balance of resources in the optimisation framework is verified. First, dummy demands are implemented for the different layers, which allows to trace the decisions in terms of units implemented. It is verified whether all units are supplying their respective demand, whether the streams are correctly connecting the units and whether the scales of the streams are coherent. Varying configurations of technologies are tested to ensure that no model deficiencies prohibit using specific technology configurations. Moreover, to verify the superstructure's robustness, it is exposed to extreme conditions, such as having no connection to external resources.

In addition to verifying the correct implementation of the technologies, the units' costs and impact definitions are verified. With the installed unit loads obtained from the optimisations and the specific cost and impact definitions, the overall system costs and impact can be confirmed for simple configurations in which the computations are traceable. It is verified whether the overall yearly costs are correctly computed from the results of the typical days. Furthermore, it is confirmed that the varying electricity prices and assigned grid emissions are rightfully used to determine the system's operating cost.

Most attention is needed to verify the correct working of the storage units. Simple test cases using only two periods with a small number of time instances and pre-defined resource unavailability in certain time instances are constructed, which force the system to use the storage options and allow the verification of the storage levels at the end of each time instance by hand. Next, it is verified that the installed storage capacity is adequate for receiving the allocated energy. When extending the periods to the full 12 typical days, the functioning of the storage units under extreme conditions, such as, for example, including full periods without electricity supply, wind and solar radiation, is verified.

B Input Data for European Hub Airports

In section 4, the input data for the case-study of Amsterdam Schiphol airport and the energy demand profiles for the six European hub airports are displayed. This appendix displays the input data including the considered buildings as well as the passenger, flight, weather, electricity and input energy demand profiles for the five other European hub airports, which are Madrid Barajas in section B.1, Munich in section B.2, Oslo-Gardermoen in section B.3, Rome-Fiumicino in section B.4 and Zurich in section B.5.

B.1 Madrid Barajas



Figure B.1: Input data for Madrid-Barajas in 2023 including (a) the map of the considered terminal and pier buildings (using data from [74]), (b) the number of flights [86] and passengers (monthly passenger numbers from [141]), (c) the temperature and solar radiation [88], (d) the wind speed and sunshine duration [88], (e) the electricity price [90] and emissions (average yearly emissions per kWh from [91], yearly profile of renewable energy share from [90]) in Spain and (f) the estimated energy demand profiles



Figure B.2: Input data for Munich in 2023 including (a) the map of the considered terminal and pier buildings (using data from [74]), (b) the number of flights [86] and passengers (monthly passenger numbers from [142]), (c) the temperature and solar radiation [88], (d) the wind speed and sunshine duration [88], (e) the electricity price [90] and emissions (average yearly emissions per kWh from [91], yearly profile of renewable energy share from [90]) in Spain and (f) the estimated energy demand profiles

B.3 Oslo-Gardermoen



Figure B.3: Input data for Oslo-Gardermoen in 2023 including (a) the map of the considered terminal and pier buildings (using data from [74]), (b) the number of flights [86] and passengers (monthly passenger numbers from [143]), (c) the temperature and solar radiation [88], (d) the wind speed and sunshine duration [88], (e) the electricity price [90] and emissions (average yearly emissions per kWh from [155], yearly profile of renewable energy share from [90]) in Spain and (f) the estimated energy demand profiles



B.4 Rome-Fiumicino

Figure B.4: Input data for Rome-Fiumicino in 2023 including (a) the map of the considered terminal and pier buildings (using data from [74]), (b) the number of flights [86] and passengers (monthly passenger numbers from [144]), (c) the temperature and solar radiation [88], (d) the wind speed and sunshine duration [88], (e) the electricity price [90] and emissions (average yearly emissions per kWh from [91], yearly profile of renewable energy share from [90]) in Spain and (f) the estimated energy demand profiles



Figure B.5: Input data for Zurich in 2023 including (a) the map of the considered terminal and pier buildings (using data from [74]), (b) the number of flights [86] and passengers (monthly passenger numbers from [145]), (c) the temperature and solar radiation [88], (d) the wind speed and sunshine duration [88], (e) the electricity price [90] and emissions (average yearly emissions per kWh from [156], yearly profile of renewable energy share from [90]) in Spain and (f) the estimated energy demand profiles
Installed	
logies	
Technologies	
of	
apacities	
Cal	
U	

In this appendix, the technology capacities are shown which are obtained from the optimisations for the selected configurations for Amsterdam Schiphol in section 5.1 and for the five other European hub airports in section 5.3. They can be found in the tables in section C.1 and section C.2 respectively.

C.1 Energy System Optimisation Amsterdam Schiphol

	_
2	5
	re
•	-
ŗ	ī.
	In
	nt
	ē
¢	Ħ
	5
	re.
ĥ	d D
ĥ	
-	ğ
1	
	ОШ
د	Ĕ
	ŝ
	ysı
-	Ę,
	μ
	r a
	(1)
5	цĎ
e	Ξ
	r L
c	ē
	\mathbf{S}
	on
	Ĕ
-	Ξ
	š
-	ರ
	g
_	ĕ
	ŝ
	e,
5	t P
	ц
•	Г
	0
	16S
	ogles
	ologies
	nnologies
•	schnologies
-	technolo
	d technolog
	technolo
-	ed technolog
-	ed technolog
-	e installed technolog
-	stalled technolog
	the installed technolog
	the installed technology
	the installed technolog
	ties for the installed technolog
· · · · · · · · · · · · · · · · · · ·	cities for the installed technolog
· · · · · · · · · · · · · · · · · · ·	ties for the installed technolog
	cities for the installed technolog
	e capacities for the installed technolog
	cities for the installed technolog
	the capacities for the installed technology
	of the capacities for the installed technolog
	of the capacities for the installed technolog
	new of the capacities for the installed technology
	of the capacities for the installed technolog
	erview of the capacities for the installed technolog
	: Uverview of the capacities for the installed technology
	erview of the capacities for the installed technolog
	: Uverview of the capacities for the installed technology
	C.I. : Uverview of the capacities for the installed technological sector of the secto
	C.I. : Uverview of the capacities for the installed technological sector of the secto
	able C.I. Uverview of the capacities for the installed technolog
	C.I. : Uverview of the capacities for the installed technological sector of the secto
	able C.I. Uverview of the capacities for the installed technolog
	able C.I. Uverview of the capacities for the installed technolog

	Electi	Electricity generation and storage	eration e	He	Heating an	and cooling	<u>භ</u>			CO ₂ network	twork			Т П	Hydrogen technologies	n techn	ologies	
	(solar cell area) \mathbf{PV}	Mini wind turbine (sənidrut bniw rədmun)	Battery (electricity stored)	Boiler natural gas (heat produced)	Combined cycle natural gas (electricity produced)	Heat pump (electricity used)	Chiller (electricity used)	Central plant winter (electricity used)	Central plant summer (heat exchanged)	Distribution pipes (length pipes and excavation)	Air conditioning (heat exchanged)	Refrigeration (electricity used)	Heating (electricity used)	(electricity required) AEC	electricity required) (electricity required)	(electricity required) SOEC	(electricity produced) SOFC	Hydrogen tank (H ₂ stored)
	km^2	1	MWh	MW	MM	MM	MW	MW	MW	km	MM	MM	MM	MW	MM	MW	MM	MWh
B01+CHI	0	0	0	16.88	0	0	11.77	0	0	0	0	0	0	0	0	0	0	0
CC+B0I+CHI	0	0	0	9.3	9.02	0.11	11.77	0	0	0	0	0	0	0	0	0	0	0
HP+CHI	0	0	0	0	0	7.55	11.77	0	0	0	0	0	0	0	0	0	0	0
$PV+CO_2N$	0.15	0	0	0	0	0	0.15	0.74	12	11.54	5.87	2.23	3.68	0	0	0	0	0
PV+CC+C02N	0.13	0	0	1.05	10.9	0	0	0	12.21	11.54	8.79	2.21	1.38	0	0	0	0	0
$PV+WT+CO_2N$	0.74	2534.37	0	0	0	0	7.79	1.17	2.81	11.54	4.88	0.42	3.68	0	0	0	0	0
BAT+PV+CC+C02N	0.14	0	9.39	1.66	10.11	0	0	0.16	12.21	11.54	8.04	2.21	1.27	0	0	0	0	0
$BAT+PV+CO_2N$	0.21	0	50.55	0	0	0	0	0.71	12.21	11.54	5.87	2.26	3.68	0	0	0	0	0
XBAT+PV+CC+C02N	0.37	0	146.18	0.38	6.88	0	0	0.56	12.21	11.54	2.39	2.22	2.25	0	0	0	0	0
XBAT+PV+CO ₂ N	0.9	0	198.73	0	0	0	0	1.06	12.21	11.54	5.87	2.26	3.68	0	0	0	0	0
H ₂ S+PV+CC+C0 ₂ N	0.53	0	0	0.05	10.72	0	0	0.38	15.22	11.54	13.97	2.08	1.98	18.89	0	0.32	4.76	104.68
XH ₂ S+PV+CC+C0 ₂ N	1.13	0	0	0.25	8.86	0	0	0.47	26.73	11.54	21.77	1.88	1.77	66.64	0	1.57	15.46	276.88
Independent, energy system	1 02	4032.54	206.95	0		0	0	000	19.91	11 54	қ <u>8</u> 7	9.06	3.68	C	0	0	Ċ	

Airports
European
of Other
otimisation
Energy System Op
Energy
С.2

Table 2: Overview of the capacities for the installed technologies in the selected solutions for further analysis from the Pareto front in Figure 20

	Heating (besu yiiity used)	MW	1.27	1.64	2.03	0	0.58	0.79	3.68	4.86	5.23	2.86	2.68	2.10
	Refrigeration (electricity used)	MW	2.21	3.76	2.56	0	2.54	1.24	2.26	3.83	2.09	0.80	2.59	1.26
CO2 network	Air conditioning (heat exchanged)	MW	8.04	11.89	5.96	0	7.28	3.66	5.87	7.85	4.84	3.51	4.18	3.28
CO ₂ n	Distribution pipes (noitersza and excavation)	km	11.54	12.56	14.37	0	11.37	6.09	11.54	12.56	14.37	7.90	11.37	6.09
	(heat exchanged) Central plant summer	MW	12.21	26.49	15.83	0	15.23	7.77	12.21	26.49	16.40	6.60	15.23	77.7
	Central plant winter (electricity used)	MW	0.16	0	0.32	0	0	0.04	0.71	0.59	0.97	0.56	0.40	0.36
ng	Chiller (electricity used)	MW	0	0	0	6.28	0	0	C	0	0	0	0	0
Heating and cooling	qmuq teat (beat yump) (beat yump)	MW	0	0	0	2.66	0	0	0	0	0	0	0	0
ating a	Combined cycle natural gas (electricity produced)	MW	10.11	12.90	9.51	5.38	8.05	5.08	0	0	0	0	0	0
He	Boiler natural gas (heat produced)	MW	1.66	2.03	4.57	2.74	1.80	1.28	0	0	0	0	0	0
generation orage	Battery (electricity stored)	MWh	9.39	10.56	3.96	0	4.70	0.83	50.55	65.61	43.16	0	56.37	13.35
	midut baiw iniM (sənidut baiw rədmua)	1	0	0	0	0	0	0	0	0	0	0	0	0
Electricity and s	(solar cell area) PV	km^2	0.14	0.17	0.11	0.05	0.12	0.07	0.21	0.25	0.17	0.04	0.23	0.10
			EHAM	LEMD	EDDM	ENGM	LIRF	HZSH	EHAM	LEMD	EDDM	ENGM	LIRF	HZST
				Battery for energy	storage and all	non-storage	technologies			Battery for energy	storage and only	renewable	technologies	

Literature Study

Introduction

Airports are complex infrastructures. Many different stakeholders are involved in the operation of airports, and thus, many different parties require energy at an airport. Consequently, optimising an airport's energy systems involves different considerations when defining the demand, choosing energy technologies and setting up the requirements. The literature review aims to give an overview of relevant work for energy systems of airports. In this Chapter 1, the airports with their stakeholders and division in space are introduced. Next, Chapter 2 gives an overview of the energy demands at an airport. Chapter 3 then explains what studies have already been performed related to airport energy system optimisations. Subsequently, Chapter 4 presents possible technologies to be used with their advantages and disadvantages. Finally, Chapter 5 ends with identifying the research gap for this study and Chapter 6 outlines the research.

1.1. Characteristics of an Airport

Aviation is one of the most important drivers of globalisation, and cities with large airports or, even hub airports are often important economic centres [4]. Airports thus play a key role in the transportation system in the 21st century. An airport is defined as "a place where aircraft regularly take off and land, with buildings for passengers to wait in" [5]. The busiest airport in the world is Atlanta (USA), which was employed by 104.6 million passengers in 2023 [6]. Airports accommodate up to 775,000 aircraft movements per year, with again Atlanta (USA) being the busiest airport in 2023 [6]. Finally, they move up to 4.3 million tons of freight per year, with Hongkong (China) leading this statistic in 2023 [6].

Due to the considerable number of passengers, freight, and aircraft that are moved, different stakeholders are involved in an airport. These range from the passengers departing or arriving to the federal government where the airport is located [2]. The first group of stakeholders are the passengers, who are interested in on-time flights, smooth operation, and good and diverse services [2]. Their interests are more personal, except for the interest of having a connection hub that offers a diverse set of destinations. On the other hand, there are air carriers and general aviation, which are the direct users of the airport. They are concerned that the needs of their passengers and clients are satisfied while paying low operation costs and ensuring the safety of their operation [2].

The third group of stakeholders are not customers but the airport itself or service providers located at the airport. They want to satisfy all needs of passengers and air carriers to stay attractive compared to other airports and to increase the number of passengers while also maximising revenue and increasing environmental sustainability [2]. They must do this by ensuring the infrastructure is up to current standards and the employees are well-trained and satisfied [2]. Next, there are multiple stakeholders providing services at the airport, such as restaurants, parking operators, providers of transportation at the airport or service providers for the air carriers, which are interested in the growing number of passengers but also minimum fees they need to pay to the airport [2].

Finally, there are many outside stakeholders who also have interests related to airport activities. First, investors [2] who desire optimal (economical) performance and growth in terms of passengers and revenue. Organisations and communities in the region, as well as local and federal governments, want a maximum number of destinations, an airport up to the current technological standards, the possibility

for growth and safe operation [2]. However, communities, local governments and environmental NGOs also desire minimum noise pollution and emissions and, thus, a minimum impact of the airport on the environment [2].

Overall, the goals of the different stakeholders range from an efficient and smooth operation of the airport, minimal costs, growing passenger numbers, economic growth and maximum revenue to a minimised impact of the airport on its environment.

Due to the large number of stakeholders involved, an airport structure is very complex, containing several facilities. An exemplary layout is shown in Figure 1.1 [7]. As can be seen in the figure, the airport can be divided into two different parts: the landside area (coloured in red) and the airside area (coloured in orange) [8]. The landside area primarily includes all areas where passengers or freight move. They, for example, contain passenger terminal buildings, freight terminals, parking lots, railway stations, etc. [9]. The airside area consists of the areas where aircraft are moving. It includes, for example, the terminal airspace, the aircraft stands, taxiways, and runways [9].



Figure 1.1: Typical airport infrastructure layout, not up to scale (Reproduced from [7])

 \sum

Airport Energy Demands

The extensive range of stakeholders and numerous facilities at an airport result in significant energy consumption. In fact, Schiphol airport consumes an equivalent amount of electricity to that of 50000 households [10], thus being equal to the electricity consumption of a small city. Considering that most energy consumed at the moment is still non-renewable, this leads to a considerable potential for reducing the environmental impact of an airport.

The environmental impact of an airport can be reduced by employing different strategies. First, the energy demand can be reduced through efficient and optimised operation. Corlu et al. [11] studied different options for optimising the energy consumption in transportation systems. For air transportation, the most significant reductions can be achieved through aircraft trajectory optimisations and airport ground operations, such as taxiway optimisation, runway scheduling, or integrated approaches. Operational changes can be implemented quickly and lead to fast reductions in energy consumption.

The second option for reducing the environmental impact of an airport requires the introduction of new technologies. As can be seen in Figure 1.1, airports have many different facilities, all consuming energy and resources. Greer et al. [12] thus suggested the introduction of renewable electricity sources, the deployment of electric vehicles and equipment, efficient water management, the introduction of energy-efficient fixtures as, for example, light emitting diodes or the choice of durable materials in construction to make an airport more sustainable. This could come along with an efficient and optimised renewable energy system design.

Before optimizing and reducing an airport's energy consumption, it is crucial to understand its energy needs. This section explains the energy demand of an airport by first providing an overview of the energy demand at European airports in Section 2.1. Section 2.2 then illustrates the characteristics of energy demand using specific airports as examples. Section 2.3 discusses how to determine the overall energy demand of an airport. Next, Section 2.4 explains the specific energy demands in different areas of the airport. Finally, Section 2.5 defines the main influencing factors, and Section 2.6 presents methods for quantifying energy consumption and its impact.

2.1. Overview of Energy Consumption of Real Airports

Energy consumption significantly varies between different airports. It is hard to compare the specific consumptions, as the metrics provided by airports are varying, and often, little data is available. The main parameters that can be found for most airports are the total yearly energy consumption, the number of aircraft movements at an airport, and the total number of passengers per year, the latter two being correlated with each other. The energy consumption is plotted against the passenger numbers in 2.1. It must be noted that not only the consumption of individual airports but also of airport groups, such as, for example, the Spanish airports of AENA and the Aéroports de Paris, both operating more than 20 airports, are given. This leads to points with significantly higher energy consumption, movement and passenger numbers than the average, but the specific energy consumptions are comparable.

From Figure 2.1 it can be observed that there are a few outliers with a high yearly energy consumption, most notably the airports of Milan (yearly total energy consumption of 452 GWh) and Munich (387 GWh). This could be explained by different definitions of what is included in the overall energy con-



Figure 2.1: Yearly total energy consumption of 20 large European airports and airport groups vs. number of passengers at airport [13]–[36]

sumption and by the use of different energy resources with lower efficiencies. In general a linear trend can be observed for the yearly total energy consumption with respect to the number of passengers. The same trend can be observed when plotting the yearly energy consumption with respect to aircraft movements.

2.2. Characteristics of Energy Consumption of Real Airports

Having seen the overall energy consumption of different airports, the different usages of energy as well as the resources consumed can be analysed. For both of these aspects, available data is quite different for every airport, and only a few of them provide detailed information. Regarding the usage of energy, six airports provide more detailed data on how much energy is consumed for heating, cooling, as electricity or as fuel [15]–[17], [19], [20], [25]. As a first trend, it was observed that electricity usage is the most important energy consumer with approximately 57%, followed by heating usage with approximately 31% and cooling and fuel usage both with approximately 12%. However, those numbers must be used with caution due to the possibly different definitions of usage for the different airports.

Next, the resources consumed by the energy system were analysed. Again, only for six airports detailed data on the consumption of resources was found [13], [14], [18], [21], [27], [28]. The results are shown in Figure 2.2. Significant differences can be observed for the airports. At Frankfurt airport, 69% of all consumed energy resources are diesel, while it only has a minor share in all other airports. In Munich and Milan, more than 72% of the consumed energy resources they consume, representing approximately 63% to 79% of the consumed resources. Again, it must be noted that these results should be interpreted carefully, as the way the data is provided differs by every airport.



Figure 2.2: Comparison of indicated energy resource consumption distribution and consumption per passenger at different European airports [13], [14], [18], [21], [27], [28]

One last metric which is given for many airports is the share of renewable energy used at an airport. For the analysed airports, this varied from approximately 15% at Zurich airport [15] to approximately 80% for the airports in Spain [25]. However, also these values must be taken with caution: for the Spanish airports, it has, for example, been assumed that the electricity is guaranteed to be 100% renewable [25], whereas in Zurich, only 28% of the consumed electricity was considered to be renewable [15]. Overall, the importance of providing consistent metrics for the energy consumption of all airports was observed, as the results were difficult to compare and are difficult to use as a general base of comparison for later obtained results.

2.3. Determining Overall Airport Energy Demand

In Section 2.1, the energy demand for 20 of the largest European airports or airport groups was presented. A trend in energy demand per passenger and air movement was observed, though there were still significant differences between the different airports.

A similar study was performed by Costa et al. [37], where the energy consumption of European airports in 2012 was reviewed. In this study, data was obtained for 51% of the 113 studied sites. Significantly different energy consumptions were observed, but an average of 9.29kWh/passenger/year was obtained, which is slightly higher than for the average airports shown in Figure 2.1. It was also observed that larger airports such as Heathrow, Charles de Gaulle or Frankfurt have 50% or 100% higher energy consumption per passenger than smaller airports. Possible improvements identified were replacing lighting technologies, improving the heating, ventilation and air conditioning (HVAC) system, which causes 80% of the energy consumption and implementing renewable energy technologies.

Overall, airports have a wide range of different facilities with different energy demands. The PhDthesis written by Ortega Alba [38] analysed in detail the different energy demands at the Santander Airport in Spain. It explained and defined the energy demand in three steps. First, it characterised the airport based on general administrative, statistical or meteorological parameters and overall information on the airport, its management and its operation characteristics. In the second step, energy data was analysed by identifying and classifying the main energy users of the airport and by defining an energy inventory and balance. Finally, in the third step, the electric pattern is described. For this, general, hourly, weekly, monthly, seasonal as well as yearly energy demands were analysed, and profiles were obtained. Several concepts were employed to characterise the airport's energy demand. The energy demand was divided into airside and landside demands. The airside part of the airport is the part where the passenger is the primary customer, whereas on the airside part, the aircraft and its related activities are the primary customer. Energy demands in both parts could be further subdivided into different clusters and demands, such as lighting, data processing, HVAC, signalling and information, etc. It was found that the energy demand is independent of the day in the week but that the time during the day and the season have a significant impact.

Another study defining the overall energy demand of an airport was the Master's thesis written by Radhakrishnan [39] on the energy system of Stavanger airport in Norway. In this study, the energy consumption, distribution and demand of the airport were determined and compared with the theoretically required energy for electric aviation. It was found that the demand for electric aviation is significantly higher than the demand for the airport itself. Photovoltaic cells (PV) were found to not be enough to provide all the electricity required, and correct placement and inclination of the PV were identified to be important. Batteries and hydrogen storage were observed to be unprofitable, as it is cheaper to sell and buy electricity from the grid, even if the energy demand in winter is significantly higher than in summer. Finally, energy management measures, such as insulation, were also important and could lead to a reduction of energy consumption of up to 25%.

Rubeis et al. [40] analysed the energy consumption, including electricity, thermal energy and fuels, of Leonardo da Vinci International Airport in Rome. For their approach to implementing an energy management system, they determined both the energy production as well as the consumption in different airport areas (terminaly, offices, runways, etc.) for the three different demand types. This allowed to obtain a detailed division of the energy consumption and to assess significance of the different energy consumption during summer due to air conditioning and thermal demand during winter due to heating demands. Furthermore, improved energy management was observed to reduce electricity demand, whereas the thermal demand increased with an increasing number of passengers. Different technolo-

gies were proposed for the energy system, such as a high concentrator solar plan, mini and micro wind turbines and electrochemical storage and a Smart Grid system was optimised for load management. It was suggested to apply similar energy system solutions also for other European airports.

2.4. Demands of Different Airport Areas

The energy demand of an airport is composed of several components. From the previous studies, the following three main demands were observed: electricity, thermal energy for heating and cooling and fuel. Another option for dividing the demand is to only consider the demand of specific sub-areas of the airport. This was done in several studies considering, for example, the energy demand of airport buildings, airport operations, etc. Most importantly, the energy demand can be divided into activities taking place on the landside and airside area of an airport, as was explained in Section 1.1.

Figure 2.3 attempts to give an overview over all energy demands which were identified in different studies [38], [40]–[43]. They are clustered into four main groups: buildings, vehicles, aircraft and ground. Each of the group is then split into smaller sub-groups with similar energy usage characteristics. Every sub-group is categorised in either landside or airside energy demands. For all sub-groups, an overview of the included activities/facilities is given as well as a summary of the types of energy demands.

2.4.1. Landside Area

As could be observed from S. Ortega Alba [38], the largest contribution of the landside energy demand comes from the terminal buildings. The study divided this demand into different demand sources, which are heating, ventilation and air conditioning, lighting, information and communication technology, data centres, signalling, security, electromechanical and other various equipment around the airport. Next to the terminal buildings, parking and urban zones were observed as the second consumer on the landside area. Similar observations can be made from de Rubeis et al. [40], where the most significant energy consumption at the airport was observed from the terminal buildings.

From Xianliang et al. [44], the cooling plant as well as the HVAC system were observed to have the most important contribution towards the terminal building energy demand, making up approximately 32% and 17% of the total energy consumption. They were followed by offices and commercial areas with approximately 13% and 11% of the total energy consumption. Lighting, advertising, elevators, boarding bridges, the luggage system, etc., only displayed minor energy consumption contributions. When comparing the different energy demands, the HVAC system had an uneven distribution of consumption in space and time and was influenced by several factors, for example, the meteorological data or the flow of passengers, the former explaining also the important contribution of the cooling plant, as the studied airport was located in a warm climate zone.

The demand on the energy system can be lowered by applying methods of energy consumption reduction. Yildiz et al. [45] studied different options, which allowed the reduction of both the consumption and the emissions of the airport terminal building. In the studied airport, the HVAC system was observed to have the most important contribution to the overall consumption. Promising reduction strategies concern this system, such as improving the system's efficiency, changing the setpoint temperature, installing additional heat exchangers, or replacing the circulation pumps.

Next to the buildings with heating, cooling, ventilation, air conditioning, lighting and various other demands spread over the building, there are individual systems with very specific energy demands on the landside area of the airport. One of them is the baggage handling system, which energy consumption is analysed in multiple studies. Kierzkowski and Kisiel [46] created a simulation model of the check-in part of the baggage handling to determine its energy consumption, which is dependent on the structure of the baggage handling system, the flight timetable and the allocated resources for each flight. The baggage handling system was also an object of analysis for Lodewijks et al. [47], who observed that with an optimised baggage handling system and specially optimised operations, CO₂ emissions could be reduced substantially. Those small models modelling individual subsystems could be used as modules for an extended model of the energy consumption of the airport systems [46].

2.4.2. Airside Area

The second part of an airport is the airside area, which is primarily characterised by the vehicles operating on the taxiways, aprons and around the gates. Several types of vehicles operate in an airport,



Figure 2.3: Overview of airport energy demands retrieved from literature review [38], [40]-[43]

most of them in the airside area, as it can be seen in Figure 2.4 [42]. The vehicles found at the landside area are mostly related to the travel of the passengers to the airport or to the commute of the airport employees. On the other side, vehicles in the airside area of the airport are related to loading the aircraft, towing the aircraft, preparing the aircraft for take-off or for other tasks in the airside area.

The study performed by Liu et al. [42] performed an extensive investigation of the energy consumption of ground vehicles at the example of a Chinese airport. They observed that the electrification rate is twice as high for airside (25%) than for landside (12%) vehicles, while there are more landside than airside vehicles at an airport. Furthermore, they mention the important storage potential of airside vehicles, which could help achieve an airport energy system based on renewable resources. Vehicles on the airside part of the airport, in particular, have the potential for such a storage system, while the overall storage potential of all vehicles would be nearly sufficient for running the airport fully on solar energy.

Bao et al. [48] focused on one specific type of vehicle, the towing tractor, but suggested extending their model also to other ground support equipment types. They investigated the performance of a mixed fleet scheduling of both conventional and electric towing tractors, where the electric towing tractor was also assumed to be able to replace the aircraft auxiliary power unit (APU) while towing. The ideal



Figure 2.4: Ground vehicles which can be found normally at an airport in the airside and landside area (Reproduced from [42])

ratio between conventional and electric towing tractors, as well as the performance in terms of costs and flight delay, was observed to vary significantly based on the scenario, which is characterised, for example, by the distances to be driven and the number of flights to be served.

Multiple current projects focus on decreasing the energy demand of ground vehicles through efficient operation, routing and scheduling. An example of this is the study performed by Sigler et al. [49], which optimised the airport shuttle routes to operate them more energy efficiently. Optimisation of the shuttle routes allowed to achieve an energy reduction at the cost of an increased average passenger wait time. Another example is the fleet scheduling of electric towing tractors, as was done by Zoutendijk and Mitici [50]. While using more electric vehicles in order to reach climate targets, their electricity demand also increased. Thus, it is important to optimise the vehicle schedules to operate them under a limited electricity capacity. From the optimised schedules, they observed that the highest charging occurs between 15:00 and 17:00 when assuming regular operations. With increasing battery capacities, it was possible to only charge during the night, which, according to their results, saves 10% of the emissions, as no fast charging during the day is required.

2.5. Factors Defining Airport Energy Demand

Airport energy consumption is influenced by several factors. These can be the size of the airport, as is observed in Section 2.1, the flight schedule or the environment the airport is located in. For example, a correlation can be observed between the terminal area and the passenger flow, and they both also determine the overall airport energy consumption [51]. As expected, larger airports have higher energy consumption but typically lower consumption per passenger [52].

Li et al. [52] studied the influence of climate zones on airport energy consumption. The authors observed that the highest consumptions occur in freezing and cold climate environments. These are followed by airports located in climate zones with hot summers and cold winters, hot summers and warm winters and mild climates in this respective order. Overall, the energy consumption in a freezing and cold climate was two to four times higher than in mild climates. Comparable studies confirmed the observation that the highest energy consumptions occur in areas with hot summers and/or cold winters [51]. This can well be explained by the fact that heating and air conditioning represent around 35-50% of the overall airport energy consumption [51]. Costa et al. [37] furthermore suggested looking at weather data as an additional factor defining energy consumption.

Lin et al. [51] also looked at the energy consumption in different regions. For airports in Europe and Japan with an annual passenger flow higher than 10 million passengers, significantly higher energy consumptions were observed than for airports of the same size in China. Overall, energy consumption could be split into two parts: basic consumption, which is the energy required for the transportation functions of the building, and variable consumption, which is dependent on the climate and thus significantly varies between different airports.

Another determining factor is the spatio-temporal distribution of the passengers within the terminal.

Gu et al. [53] introduced a model which allows to estimate the number of people at different areas in the airport at different times, using a set of 14 different equations characterising the passenger distribution at different locations, such as the check-in area, the security check area, etc. Those distributions then allowed to calculate the energy demand for HVAC. 11.3% lower energy demands were obtained when applying the spatiotemporal model for designing the energy consumption, which showed the potential of adapting energy demand to the passenger distribution. However, the model was still limited to specific types of airports and needs to be extended to other airports that are different in terms of size or the processes to be globally applicable. Furthermore, the impact of baggage, the effect of passenger relationships, holidays or differences in queuing time can be considered to improve the model further.

2.6. Methods for Quantifying Airport Energy System

Different methods are observed to quantify the overall energy demand of an airport. When looking at the airport buildings, the energy consumption per unit area is often used and considered as the best indicator [51]. When looking at the overall energy consumption, it is often given per passenger. However, more methods for quantifying the energy consumption of airports could be beneficial. Most importantly, they need consistent and accurate measurement methods and uniform standards in the creation of statistics [52], in order to make the energy consumption results comparable between different airports and to create a base of comparison when developing new energy systems.

Greer et al. [12] looked at different metrics which are used to assess the environmental sustainability of an airport. The greenhouse gas emissions of the energy system are often employed to assess the environmental performance of an airport. Alternatively, there are however also other metrics: The LEED (Leadership in Energy and Environmental Design) is used to assess buildings and gives points to buildings based on their design, the materials used, etc. [12]. Moreover, the LCA (life-cycle assessment) considers the environmental impact from the extraction of materials for production up to the use of different components and can be used to assess different construction components of the airport, such as typically for the airport pavement but also for energy technologies [12].

3

State-of-the-Art of Airport Energy System Optimisation

Optimisation of energy systems for airports has already been performed in multiple studies. An overview of similar studies can be seen in Table 3.1. Most studies only focus on optimising the (terminal) building energy system, analysing how hybrid, renewable or combined heating, cooling and power energy systems are performing, whereas only a few studies treat the airport operations or the aircraft energy demands. As key performance indicators (KPI), mostly the overall or operational costs and/or the environmental impact are presented.

Case	Focus		Scope			Meth	od			KPI sho	wn
study	rocus	Buildings	Opera- tions	Aircraft	Model	Objective	Grid con- nection	Fossil fuels	Costs	Env. impact	Others
[41]	Energy system	Yes			MILP	Cost	Yes	Yes	Yes		
[54]	Hydrogen		Yes		MILP	Cost	Yes		Yes		Technolo- gy sizes
[55]	Hybrid energy sys- tem	Yes			HFSP	Cost	Yes	Yes	Yes		Supply structure
[<mark>56</mark>]	Renewable energy system	Yes			MCDM	Cost, ren. Energy	Yes				
[57]	Hydrogen solar- storage		Yes		MILP	Cost	Yes		Yes		
[58]	Combined heating, cooling, power	Yes			no opt.		Yes	Yes	Yes	Yes	
[59]	Combined heating, cooling, power	Yes			LP	Cost	Yes	Yes	Yes	Yes	
[60]	Nearly-zero exergy airport	Yes			REMM	Exergy	Yes	Yes		Yes	Exergy
[<mark>61</mark>]	Building cooling heating and power	Yes			SQP	Op. Costs	Yes	Yes	Yes		
[62]	Review of different methods	Yes	Yes	Yes	no opt.		Yes	Yes			
[63]	Energy model exis- tent airport	Yes	Yes		LP	Cost	Yes	Yes	Yes	Yes	
[43]	Ground support equipment		Yes		no opt.		Yes				Charact. El. Sys.
[64]	Power system	Yes			HOMER	Cost	Yes		Yes	Yes	
[65]	Energy hub	Yes	Yes	Yes	PSO	Cost, env. Impact	Yes	Yes	Yes	Yes	
[65]	Thermal energy system	Yes			MILP	Cost	Yes	Yes	Yes		

Table 3 1	Overview of	f studies or	airport	enerav	system	optimisations
10010 0.11		3100103 01	anport	chicigy .	system	opunisations

Different mathematical approaches are available for energy system optimisation: these can be, for example, linear, dynamic, mixed integer linear or agent-oriented programming, fuzzy logic or heuristic methods [66]. A variety of these are used in different airport energy system optimisation studies as shown in Table 3.1:

- The most commonly used optimisation method is linear programming (LP), which allows to find an optimal solution based on an objective function and a set of constraints, which are both defined as a linear function of the employed decision variables [67].
- Mixed-integer linear programming (MILP) is a specific type of LP, where both integer as well as non-integer variables are allowed [67].
- In sequence quadratic programming (SQP), multiple quadratic model optimisations of linearised constraints are performed after each other to find the optimal solution in an iterative progress [61].
- For particle swarm optimisation (PSO), a swarm of solutions is generated, wherein every solution moves towards an optimal solution throughout an iterative process based both on its own optimal solution as well as the optimal solution of the swarm until reaching a global optimum [68].
- Hybrid fuzzy-stochastic programming (HFSP) combines fuzzy programming, joint probabilistic constraint programming and Monte Carlo techniques and manages uncertainties by modelling them as fuzzy sets and as probability distributions, thus leading to improved performance [55]. However, large-scale problems are hard to solve using this technique.
- The method of Rational Exergy Management Model (REMM) is a novel method based on the exergy on the energy supplied and required and restrains the consumed primary energy as well as the emissions based on the match between supplied and required energy [60].
- Multiple Criteria Decision Making (MCDM) is a more systematic approach of approaching problem with a clearly defined decision-making process [56].

Different optimisation tools are available for energy system optimisation and were employed in the different studies. One example is the Hybrid Optimisation Model for Electric Renewable (HOMER), which was developed by the National Renewable Energy Laboratory in the United States and which allows to simulation many configurations of energy systems based on climate and technical component data [64]. Another option is the OSeMOSYS framework, which allows to model energy systems from village up to global scales in a modular structure also allowing the implementation of long-term dynamics such as changes in energy prices or demand [63].

Regarding the design variables, these are often the sizes of the different energy technologies employed. However, different more specific design variables might be employed to limit the number of design variables and thus reduce complexity of the optimisation problem. For instance, Kilkis and Kilkis [60] employed only three main design variables, being the co-generation engine capacity and peak power, the allocation of generated power supply and the mixing ratio between natural gas and biogas. This allowed to solve the optimisation problem using "a simple search method" [60].

3.1. Buildings

Buildings are a frequent scope of analysis in literature. An overview of the studies optimising the building energy system is shown in Table 3.2, including both the demands and the technologies considered in those studies.

From Table 3.2, it can be seen that most studies focus on the heating, cooling and power supply of the airport buildings [41], [55], [59], [60]. Only a few studies focus solely on individual demands, for example, the electric power supply [56], [64], electric power supply combined with cooling [61] or fulfilling the thermal demand of an airport [69]. In addition to the building demands, a few studies are also considering the electricity demand of vehicles, which are either simple light-duty and maintenance vehicles [63], the ground service equipment or a possible additional load due to electric aircraft [65].

Jin and Li [55] found that their hybrid energy system combining technologies using natural gas and electricity was most sensitive to changes in the cooling demands, as cooling is required during nearly half of the year and as it relies on technologies consuming electricity which is more costly than the different types of fuel. Cardone et al. [59] furthermore found that the improvements from new energy systems get more important when the demand for heating and cooling is increasing. In general, for the energy demand of an airport, it is crucial to consider seasonal and daily variations, as the assumption of a fixed load, which is often made to reduce computational complexity, can lead to erroneous or inaccurate results [64]. It is also important to consider seasonal and daily variations in the energy sources,

 Table 3.2: Demands considered and technologies employed in different airport optimisation studies considering the airport building energy demands

Airport and	reference	15 air- ports [41]	Chang- sha [55]	none [56]	Mal- pen- sa [58] [59]	Schi- hol [60]	Qing- dao [61]	Tu- rin [63]	In- che- on [64]	Kan- sai [65]	Ge- ne- va [69]
	Heating	Yes	Yes		Yes	Yes		Yes		Yes	Yes
	Cooling	Yes	Yes		Yes	Yes	Yes	Yes		Yes	Yes
Demands	Power	Yes	Yes	Yes	Yes	Yes	Yes	Yes	Yes	Yes	103
	Mobility	163	103	163	163	163	103	163	163	Yes	
	WODIIty									163	
	PV	Yes	Yes	Yes		Yes			Yes	Yes	Yes
	Wind	Yes		Yes					Yes	Yes	
	Geothermal power			Yes							
Electricity	Gas turbine	Yes	Yes		Yes		Yes			Yes	
Generation	Steam turbine	Yes									
	Biodiesel generator			Yes							
	Thermoelectric generator									Yes	
	Internal comb. Engine Gas boiler										
		Yes	Yes		Yes			Yes		Yes	Yes
	Diesel boiler							Yes			
Heating	Electric boiler/heater	Yes									Yes
Heating	Heat boiler									Yes	
	Electric heat pump	Yes									
	Ground source heat pump		Yes			Yes					
0 11	Absorption chiller	Yes	Yes		Yes	Yes	Yes				
Cooling	Compression chiller	Yes	Yes		Yes		Yes	Yes			Yes
	Combined heating & cooling	Yes									
Combined	Combined heating & power					Yes					
	HVAC	Yes									
	Methane reactor									Yes	
Hydrogen	Electrolyzer									Yes	
	Hydrogen-oxygen fuel cell									Yes	
	Hot water storage	Yes	Yes			Yes					Yes
	Chilled water storage	Yes	Yes				Yes				
Storage	Ice thermal energy storage	Yes				Yes					
-	Electrochemical storage	Yes		Yes					Yes	Yes	
	Hydrogen storage tank									Yes	
Others	Desalination	Yes									

which influence the availability and costs of the energy generated from the different technologies employed.

Next to the demands, Table 3.2 displays the technologies employed in the related studies. Multiple different technology options are considered for electricity generation, heating and cooling, respectively. Many studies furthermore add the option of energy storage, which could be achieved using thermal (hot water, chilled water, ice), electrochemical and hydrogen storage systems. Hydrogen generation and usage are only modelled in one study, which also includes the demands of mobility and thus uses hydrogen as a renewable alternative for mobility applications [65]. Different to the energy generation and storage technologies, desalination is added as a possible water source alternative, which can be employed when the charges for buying the required volume of water are too expensive [41].

Overall, it can be seen that PV power is the most commonly suggested electricity generation method next to electricity coming from the grid. However, PV panels might be a risk for the pilot due to glare effects, which need to be carefully investigated before deciding on where PV panels can be placed at an airport [60]. Common alternatives for electricity generation are wind energy and gas turbines. For heating and cooling, gas boilers or ground source heat pumps and absorption or compression chillers are implemented most often.

Although the use of PV systems might be limited because of glare effects, they should not be dismissed entirely. Instead, careful consideration should be given to their placement, as they are highly appealing due to their often lower levelised cost of electricity compared to grid prices [41]. Wind energy can be a promising alternative to PV in achieving emission-free or nearly zero-exergy airports [60]. Wind

energy can be generated both on-site and off-site from the airport.

In general, many studies observe the potential of combined technologies: these could, for example, be combustion engines for combined heating and power [41], combined heating, cooling and power units based on a combination of a gas turbine, boiler, absorption chiller and an auxiliary electric chiller [58] or combined cooling, heating and power systems based on a gas turbine and an absorption chiller [55]. The costs of these systems can be reduced through an optimal mix of fossil and alternative fuels [60]. As an emission-free alternative, ground-source heat pumps also show comparable performances [55]. Looking at energy storage, it was seen that thermal energy storage is the most attractive option due to their lower investment costs compared to electrochemical energy storage [41].

The different studies also show the importance of considering variations in the grid: Jin and Li [55] found that the share of electricity from the grid varied between 86.2% and 48% depending on the current electricity price. Thus, the electricity and natural gas prices (and of other resources) and their variations over the day are crucial for identifying the optimal strategy [61]. Zhang et al. [61] observed that economically, it is more attractive to use electricity price decreased in Qingdao in off-peak hours during the night. Cheap electricity bought during off-peak hours can further be used to store chilled water obtained from the absorption for periods with higher energy prices.

From most of the studies, it is observed that creating an optimised energy system allowed for a significant reduction in costs. This demonstrates that the economic performance of the system relies on a well-designed operation strategy [61]. In fact, Zhang et al. [61] observed a 24% operation cost reduction for summer days due to their optimised building cooling, heating and power system with integrated thermal storage. Thiem et al. [41] found that for an integrated energy system with electricity generated on-site, up to 61% of total costs can be saved compared to the case of only using energy from the grid. Also, Cardona et al. [59] found in 2006 that a combined heating, cooling and power plant re-powering in 2010 for Malpensa airport would be profitable with a payback period of only 3.2 years.

Cardona et al. [58] additionally observed that even a non-optimised combined heating, cooling and power plant already allowed to achieve energy intensity, cost and emissions reductions, the latter being approximately 16.7% compared to the non-combined conventional system. However, the most cost-effective is not always identical to the most energy-efficient energy system [61]. In fact, Cardone et al. [59] observed a profit decrease of 583000€/year to achieve a 2.01% increase in the primary energy saving index, which was required to fulfil the energy saving requirements set by the European Union (EU). Thus, aiming for slightly less than the optimal profit allowed for much better energy efficiency and environmental performance. As a general comment, it can, however, be noted that those optimisation results should be taken with caution, as these simulations have limitations and as assumptions were made, which may lead to a good but not exactly the optimal result [64].

Next to employing an accurate model, it is also important to consider governmental regulations when developing new energy systems. An example of this can be seen at Incheon International Airport in South Korea [64], where the government has a policy of "eco-friendly airport operations and preparations for a green future", requiring the large-scale deployment of renewable energy and of measures to increase energy efficiency of airports.

Several general conclusions and recommendations can be drawn from the previous studies. Firstly, several data sources are used when developing an energy system model. Relying on more recent sources enables more precise parameter estimations, leading to more accurate results [55]. Additionally, when possible, the model should be validated against real data, as was done by Prussi et al. [63]. Secondly, energy systems are rarely overhauled abruptly. Therefore, long-term road-maps are essential to plan an airport's energy transition properly [64]. These road-maps can include retrofitting renewable energy technologies into existing energy systems, as was explored by Belfiore [69]. Finally, economies of scale must be considered when designing larger energy systems, as they often demonstrate better economic performance than multiple smaller ones [64]. However, systems with strong economic performance also tend to have a higher risk of violating constraints [55], creating a trade-off between robustness and cost-effectiveness.

3.2. Operations

Two main pathways are identified for transitioning airport operations, particularly ground support equipment, to renewable energy sources: implementing a fully electric system or integrating hydrogen. Kirca et al. [43] focused on the first of the two, analysing the potential for the electrification of the ground support equipment across all categories. These included pushback vehicles, service vehicles (fuel, catering, etc.), loading vehicles (conveyor, passenger stairs, etc.) and power supply for the aircraft. The model determined the optimum between the number of ground support equipment vehicles required and the required recharging instances for varying battery capacities. It was found that the charging schedule plays a crucial role. For instance, charging during the day, in addition to nighttime, reduced the microgrid's peak load by 23% and shortened its duration by 28%. Overall, the electrification of the ground support equipment avoided 60% of the CO_2 emissions. In the future, it was recommended that options for vehicle-to-grid storage, better grid profiles, and electrification of a full airport be looked at.

In addition to a fully electric system, hydrogen can be used to fuel vehicles or added as a storage resource. The latter was investigated in two related studies by Zhao et al. [54], and Xiang et al. [57], where hydrogen storage and supply as electricity after conversion in a fuel cell was introduced next to the PV technologies, the electricity grid and batteries. Next to the vehicles, the APU of the aircraft was also considered. In the latter study, hydrogen was also used to power the APU [57]. Using renewable technologies and hydrogen allowed to reduce the total annual costs by 41.6% and the emissions by 67.29% [57]. Furthermore, even if the hydrogen technologies seemed to be expensive at first, they allowed to achieve lower cost and environmental impact in the long term when their investment paid off [57]. Together with PV technologies and battery storage systems, they also allow to compensate for less available electricity from the grid [54]. The most economical solution further improves when having a higher solar irradiance, a lower cost for hydrogen systems and increasing oxygen prizes, whereas it remains unchanged for varying prices of electricity from the grid, emissions from the grid and the carbon tax [57].

3.3. Overall Airport

Tian et al. [65] focused on optimising the entire energy system of the airport, trying to satisfy the building heating, cooling and power demands, light demands over the airport, the demands for ground support equipment and even the demands for possible future electric aircraft. As shown in Table 3.2 they employed various technologies for the energy system, ranging from electricity generation to regular boilers and hydrogen technologies. The authors concluded that the daily variation of demands could be beneficial as the periods with no demand correspond to periods with no sun and, thus, no electricity production from PV. However, the edge times (07:00-08:00 and 18:00-20:00) are critical because of the mismatch between demand and the limited availability of sunlight during those periods. Hence, wind and natural gas generators are interesting alternative resources for stabilizing the electricity supply during periods with less PV output. Combining electricity generation technologies from waste heat, gas turbines, and hydrogen fuel cells displayed how heat and electricity usage and generation can be interconnected. Finally, by including the charging needs of electric aircraft, it was observed that their demand represents a large part of the total load, requiring the use of grid electricity in addition to local electricity sources.



Energy Technologies Suitable for Airport Environment

In the previous section, an overview of different studies on optimising airport energy systems is given. In Table 3.2, the energy technologies employed in those studies are presented. A variety of technologies are observed in the optimised superstructures. Through the choice of innovative locally applicable technologies, the energy system can benefit from its environment, such as, for example, through the introduction of ocean energy in coastal regions [62].

Consequently, this section attempts to provide an outline of the technologies that could possibly be employed in an airport energy system. This will be done by giving an overview of the technologies for electricity generation in Section 4.1, for heating and cooling in Section 4.2 and for energy storage in Section 4.3. Their respective advantages and disadvantages will be outlined. Furthermore, possible technologies for the production of hydrogen and sustainable aviation fuel (SAF) are explored in Section 4.4 and 4.5.

4.1. Electricity Generation

Local electricity generation technologies can be used at the airport or on sites close to the airport and thus allow for the airport's self-sufficiency. They allow to avoid variable electricity grid availability and costs, as was seen previously in Section 3.1. The most commonly used renewable electricity technologies are PV and wind turbines, which, however, have limitations in the airport environment, as seen amongst others in the subsequent list. Smaller wind turbine technologies would thus, for example, be beneficial, as their possible placements are less restricted than for conventional wind turbines [70]. An alternative resource with underestimated potential [71] are, for example, micro-nuclear power plants, which are ideal for sites such as airports with high energy consumption and which require flexible and reliable energy supply [71]. Alternatively, fuel cells [72] and geothermal electricity generation [73] are flexible electricity sources which can supply the airport with electricity reliably. Finally, hydropower and ocean energy are two other possible sources of electricity which are reliable [74] or have synergies with solar or wind energy [75], but they require a river, mountains or an ocean nearby and are thus limited to the geographical constraints of the airport. The following list gives an overview of the advantages and disadvantages of all technologies.

ΡV

Advantages:

- Renewable [76]
- Large technological development and cost reductions over the past 20 years [76]
- Competitive levelised cost of electricity [76]
- Large surface areas available for PV at the airport [77]

Wind Turbines

Advantages:

- Renewable [81]
- Large areas available at airports [82]
- Significant technological progress over the past 20 years [81]
- Competitive with conventional electricity sources [83]

Nuclear Micro-Reactors

Advantages:

- Emission-free [85]
- Large and underestimated economic potential [71]
- Flexible sources of energy [71]
- Ideal for high-intensity energy consumers such as airports and remote locations [71]
- Small footprint [86]
- Long life [86]
- Use passive cooling independent of electricity supply [85]

Hydropower

Advantages:

- Renewable [87]
- Available when intermittent resources are unavailable [87]
- Reliable [74]
- Advanced technology [74]

Disadvantages:

- Long return on investment time [76]
- Rare and hazardous materials used [78]
- Currently often disposed at end-of-life [78]
- Glare effects might disturb pilots [79]
- Possible radar interference [77]
- Intermittent solar radiation [80]

Disadvantages:

- High wind turbines physically interfere with airspace [82]
- The current trend towards longer blades [81]
- Cluttering effects and radio interference [82]
- Complex recycling of wind turbine blades due to carbon fibres [84]
- Intermittent wind energy [81]

Disadvantages:

- Partly low TRL [86]
- Safety concerns for nuclear energy [85]
- Disposal and isolation of nuclear waster [85]
- Nuclear power abandoned in some countries
 [85]

- Negative impact on local environment [87]
- High investment costs [87]
- Possible effects of seasonality [87]
- Only possible to use with adequate geography [74]

Ocean Energy

Advantages:

- Renewable [88]
- Economies of scale expected to decrease costs [75]
- Possible synergies with offshore wind energy [75]

Fuel Cells

Advantages:

- Can be used for combined heating [72]
- High electrical efficiencies compared to other combined technologies [72]
- Can quickly start running in case of varying demands in energy [72]

Geothermal Electricity

Advantages:

- Renewable [73]
- Available all around the year and the Earth [73]
- Competitive costs [73]
- Flexible [73]

Disadvantages:

- Not far in development [88]
- Require proximity of the airport to the ocean
- Require a significant amount of materials [88]

Disadvantages:

- Emissions depend on production of fuel used [72]
- Reliability of fuel cells can be an issue [89]

Disadvantages:

- Electricity generation requires medium- to high-temperature geothermal sources [73]
- Possibly difficult process to obtain permit [73]
- Energy waste and environmental concerns can be problematic [90]
- Risk of induced seismicity [91]

4.2. Heating and Cooling

Next to electricity, the second main demand of the airport buildings is heating and cooling. The subsequent list presents commonly used technologies for building heating. First, the widespread technologies of fossil gas or oil boilers are presented, which, however, come with the drawback of a significant environmental impact [92]. Electric boilers [93] and biomass boilers [94] are presented as renewable alternatives for the same type of technology. Solar thermal collectors [95] or air-sourced heat pumps [96] are widespread technologies which could be used instead of boilers. Furthermore, geothermal heat pumps show very high efficiencies [97] and could therefore be an option, they are however currently less used than air-sourced heat pumps [97]. Finally, the combined cooling, heating and power production is a very promising option, which could be used with different resources [98].

Gas Boiler

Advantages:

- Widespread technology [92]
- Relatively cheap [99]

- High environmental impact [92]
- Ban on gas boilers announced in many countries [100]
- Natural gas prices are volatile and depend on extracting countries [101]

Oil Boiler

Advantages:

- Widespread technology [92]
- Relatively cheap [99]

Disadvantages:

- High environmental impact (50% higher than for gas boilers) [92]
- An on oil boilers announced in many countries [100]
- Oil price are volatile and depend on geopolitical conflicts [102]

Disadvantages:

- Much lower efficiency than heat pumps, up to four times more energy for the same amount of heat [93]
- Shorter lifetime than heat pumps [93]

Disadvantages:

- The low energy density of biomass [94]
- Higher particulate emissions than natural gas or oil [94]
- Crops might be used, and thus, this technology could compete with food production [94]

Disadvantages:

- Large areas required for solar thermal collectors [95]
- High investment costs [95]

Disadvantages:

• COP is dependent on the environment and air temperature and thus highly variable [96]

Electric Boiler

Advantages:

- Renewable [93]
- Lower installation costs than heat pump [93]
- Boilers are widespread technology [93]

Biomass Boiler

Advantages:

- Renewable [94]
- Biomass is relatively cheap [94]
- Waste can be reused [94]

Solar Thermal Collectors

Advantages:

- Renewable [95]
- Widespread technology [95]
- On-site heat production

Air-Sourced Heat Pumps

Advantages:

- Renewable [96]
- Currently, the most widespread heat pump technology [97]

Geothermal Heat Pump

Advantages:

- Renewable [96]
- High efficiencies [97]
- Underground temperature is relatively stable and thus only small COP variations [96]

Combined (Cooling), Heating and Power

Advantages:

- Renewable resources can be implemented [98]
- Allows the integration of different types of energy production efficiently by, for example, recovering waste heat from electricity production for heating [98]
- High overall efficiency [98]

Disadvantages:

Disadvantages:

- High costs for drilling and installation [97]
- Risk of induced seismicity [91]
- Environmental performance depends on the fuel used

Next to the heating technologies, the following list gives an overview of possible technologies for cooling. Absorption and compression chillers are two of the most common technologies used for cooling [103]. Compression chillers seem to be the better option for large-scale cooling applications [103]. A smaller alternative for active cooling is a packaged terminal air conditioner [104]. Night ventilation and wind towers are included as two passive cooling options, which could be combined with active cooling technologies to reduce their energy consumption [104].

Night Ventilation

Advantages:

- Passive cooling technology [104]
- Can be combined with active cooling [104] to reduce energy consumption

Wind Tower

Advantages:

- Passive cooling technology [104]
- Simple technology with no moving parts [104]
- Wind towers could rotate to orient with respect to wind direction [106]
- Reduce energy consumption for cooling [106]

Disadvantages:

- Outside air can enter the building [104]
- The emissions from which can cause health problems [105]

Disadvantages:

 Cooling performance depends on wind velocity and direction [104]

Packaged Terminal Air Conditioner

Advantages:

- Common active cooling technology for commercial buildings [104]
- Simple installation [104]
- Relatively small [104]

Absorption Chiller

Advantages:

- Can use various heat sources such as waste or solar heat [108]
- Flexible [108]
- No global warming contribution [104]

Compression Chiller

Advantages:

- Dominant over absorption chiller for small or medium cooling application [108]
- High energy efficiency [103]
- Compact [103]
- Better for high-capacity cooling [103] such as at airports

4.3. Energy Storage

Energy storage could significantly help to reduce the effect of intermittent renewable energy sources on the energy system [109]. Different forms of energy storage exist, which will be presented in ??. For short-term storage, flywheels are preferred, but they come with the drawback of very fast discharge rates [110]. Pumped hydro or compressed air, on the other side, show low discharge rates and are suitable for long-term storage but come with the drawback of low energy and power density [111]. Batteries and supercapacitors are a mature and not yet mature technology respectively to store electricity directly without pumping water or compressing hydrogen and thus have faster response times [111], but are more expensive than, for example, pumped hydropower [99].

Multiple options exist for hydrogen storage. First, hydrogen can be stored in a pressurised form, requiring a large volume [112]. The second option is cryogenic storage, allowing a high storage density but requiring a lot of energy [112]. Finally, metal hydride storage could be used by creating chemical bonds between hydrogen and the metal material, but this requires the use of expensive materials [112].

Three options are also presented for thermal storage. Sensible heat storage comes with low costs but also lower storage efficiencies [113]. Latent heat and chemical thermal storage cost more, but also have higher storage efficiencies, the latter however being a complex technology still in development [113].

Disadvantages:

- High investment costs [104]
- Noise pollution/vibration [104]
- Only applicable for small or medium-sized spaces [107]

Disadvantages:

- Smaller efficiency compared to compression chiller [104]
- Less expensive and less maintenance required than compression chiller [103]

- Synthetic refrigerants used which have negative environmental impact [103]
- Noise pollution [103]

Flywheel

Advantages:

- Perfect for standby power [110]
- High power levels [110]
- High energy efficiency [111]
- No depth-of-discharge effects [110]
- High cycle life [110]
- Can be used for frequency regulation [114]
- · Can help with the intermittency of renewable energy [114]

Pumped Hydropower

Advantages:

- Low self-discharge rate [111]
- See Hydropower in Section 4.1

Compressed Air

Advantages:

- Low self-discharge rates [110]
- Flexible storage durations [110]
- · Can be stored efficiently and safely underground [110]

Battery

Advantages:

- Mature technology [111]
- High energy efficiency [111]
- High energy and power density [111]
- Fast response time [111]

Supercapacitor

Advantages:

- High power density [111]
- High energy efficiency [111]
- Fast response time [111]
- Many cycles possible [111]

Disadvantages:

- High self-discharge rates (20% per hour) [110]
- Not suitable for long-term storage [110]
- High costs [110]

Disadvantages:

- Low energy and power density [111]
- See Hydropower in Section 4.1

Disadvantages:

- Still in development [110]
- Requires caverns suitable for storage [110]
- Low energy and power density [111]

Disadvantages:

- · Low lifetime in terms of cycles and years [111]
- Use of harmful materials and chemicals [115]
- Battery waste has a significant environmental impact [115]

- Low energy density [111]
- High self-discharge [111]
- Not yet mature technology [111]
- Expensive [116]
- Packing difficult [116]

Pressurised Hydrogen

Advantages:

- Can be stored in pressure vessels or underground [112]
- Low operating costs [112]

Disadvantages:

- Large volume required for storage [112]
- Leakage can be an issue [112]
- High investment costs [112]
- Expensive high-pressure tanks (700 bar) not viable [117]

Cryogenic Hydrogen

Advantages:

• The high density of storage [112]

Disadvantages:

- Utilises a lot of energy [112]
- Expensive storage vessels [112]
- Expensive liquefaction consumes a lot of energy [117]
- Boil-off must be analysed [117]

Metal Hydrides Hydrogen

Advantages:

- Multiple different materials can be used [112]
- Some metal hydrides employ largely available materials which are easily recyclable [118]

Sensible Heat

Advantages:

- Low costs, as normally only composed of tank and charging/discharging equipment [113]
- Simple system [113]
- Underground storage possible [113]
- Commercial [113]

Latent Heat/Phase-Change

Advantages:

- High capacities [113]
- High storage efficiencies of 75-90% [113]
- Discharging temperature can be set [113]

Disadvantages:

- Issues such as managing the heater and expensive materials [112]
- Actual hydrogen storage capacity lower than values from theory [118]

Disadvantages:

- Medium storage efficiencies of 50-90% [113]
- Capacity limited by storage medium specific heat [113]
- Large space requirement [110]

- High costs due to required technology for heat transfer [113]
- Complex technology [113]
- In development [113]

Chemical Thermal Storage

Advantages:

High storage efficiencies of 75-100% [113]

Disadvantages:

- High costs due to required technology for heat transfer [113]
- Expensive materials, containers and auxiliary equipment required [113]
- Complex technology [113]
- Stability can be an issue [113]
- In development [113]

4.4. Production of Hydrogen

Hydrogen technologies are increasingly used across the world, playing a key role in the energy transition, especially because of hydrogen's energy storage potential [119]. Consequently, there is an increasing demand for green hydrogen. Besides energy storage applications, hydrogen could be used in vehicles operating at the airport and as aviation fuel in the future [62]. Currently, hydrogen is mostly produced using fossil resources such as natural gas through steam methane reforming (SMR) [120]. However, this process has high direct emissions of 9.35kgCO₂eq per kilogram of hydrogen produced [120].

It is possible to produce green hydrogen using water or biomass as a primary resource [121]. In the first case, water electrolysis using alkaline electrolysers is the most used technology, presenting efficiencies between 62-82%¹ [122]. Other electrolysers include polymer electrolyte membranes (PEM) (67-82% efficiency¹ [122]), anion exchange membranes, and solid oxide electrolysers (SOEC) [123]. Although presenting promising efficiencies (<110%¹ [122]), SOECs operate at temperatures of 700-1000°C [122] and are still too expensive, requiring further developments to reach a commercial state [123]. Some advancements have also been made in thermochemical water splitting, photoelectrolysis and photolysis to produce hydrogen with water and heat or sunlight, respectively, but are also still far from reaching technological maturity [122].

From biomass, hydrogen can be produced through biogasification or pyrolysis, with efficiencies of 30-60% and 35-50%, respectively [121]. From the latter two processes, CO and CH₄ are obtained. Through processing, these two products allow a rise in the production of hydrogen in SMR and watergas shift reactions [121]. Using biogasification, a reduction of 78.1% of CO2 emissions is achieved compared with conventional SMR [120]. Furthermore, those emissions are considered carbon-neutral [121]. Developments have also been made for biomass hydrothermal liquefaction [124], allowing hydrogen production at similar efficiencies (85-90%) to SMR [121]. Other technologies from biomass, such as dark fermentation or photofermentation, are still far from reaching technological maturity [122].

4.5. Production of Sustainable Aviation Fuels

In the previous subsection, the option of hydrogen as an aviation fuel was mentioned. However, the most probable future green aviation fuel will be SAF. In 2015, the International Air Transport Association published the first edition of its SAF roadmap [125]. At the same time, several airlines signed agreements with SAF suppliers [125] to have sufficient resources as they expected SAF to become increasingly important in the future. In fact, in 2023 in the EU the European Council adopted a law which requires 2% of SAF in aviation fuel by 2025, 6% by 2030 and 70% by 2050 [126], showcasing the importance of the development and large-scale deployment of production technologies for SAF, possibly also at airports, to achieve these ambitious goals.

There are multiple paths for the production of SAF: In general, they can be classified as biofuels, which are produced from renewable materials on a carbon basis, or as synthetic fuels, which are typically produced from syngas [127]. For both of them, multiple pathways are currently in development or even approved [127]. For biofuels, these are, for example [127]:

¹"Voltage efficiency (%) based on HHV of hydrogen (may be greater than 100%)" [122]

- Fischer-Tropsch: Syngas is produced from gasification at high temperatures, which is then converted in a catalyzed thermochemical reaction to liquid and gaseous hydrocarbons. Final refinement and distillation allows to obtain aviation fuel.
- Hydroprocessed Esters and Fatty Acids: Biogenic material is purified, the resulting oil is deoxygenated, subsequently cracking and isomerisation of the hydrocarbons takes place, before the aviation fuel is obtained through distillation and separation.
- Hydroprocessed Fermented Sugars: Sugar is transformed into hydrocarbons using fermentation, which is then hydroprocessed in an alkane. Aviation fuel is finally obtained from distillation.

To obtain synthetic fuels, there are two main options [127]:

- Power-to-Liquids: Hydrogen obtained from the electrolysis of water is combined with carbon dioxide. The mixture can subsequently be synthesised and synthetic fuel is obtained using upgrading of the synthesised mixture.
- Sun-to-Liquids: Concentrated solar radiation allows the synthesis of a mixture of water and carbon dioxide. Synthetic fuel can subsequently be obtained again by upgrading the obtained synthesised mixture.

Cabrera and de Sousa [127] observed, that synthetic fuels at the moment still require further improvements in technology, for example, for co-electrolysis, to reduce costs to a competitive level to biofuels. However, in the future, they are expected to be economically more attractive than biofuels. Still, according to them, both biofuel and synthetic fuel pathways will be important to reach a sufficient production level for achieving the goals set for the use of SAF.

5

Research Gap

It is observed from the literature review that most studies on airport energy systems are either centred on studying the terminal building or on the fuel and electricity supply to ground equipment or other vehicles at the airport. A similar observation was made by Zhou [62], who found that most studies were centred on determining the airport energy demand by designing HVAC systems or energy systems for electric vehicles at an airport. Only a few studies have looked at the entire airport, including buildings, ground support equipment, and possibly other vehicles and aircraft demands. Furthermore, previous studies often looked at the optimisation of a very specific combination of energy technologies, for example, by focusing on renewable electricity generation sources with a battery or at combined heating, cooling and power units. An optimisation with multiple possible technologies, where the optimisation selects the optimal ones, was done less frequently. This approach of using a superstructure is often employed in process integration [128] for obtaining multiple different configurations of technologies, with different configurations being optimal for different external conditions.

In consequence, the two observations just mentioned lead to the research gap, which is the basis of this study. The research gap consists of the combination of the following six points:

- Optimisation of a fully integrated airport energy system, including the airport buildings, the ground support equipment, other vehicles and possibly aircraft demands.
- Build-up of an extensive superstructure for the airport, which includes all possible technologies that can be used at an airport.
- Analysis of the robustness of the different combinations of energy technologies under varying operational circumstances.
- Sensitivity analysis of the resulting airport energy system on different technological or environmental parameters.
- Inspection of the effect of connecting the airport's energy system to its environment, such as close-by cities or industries.
- Examination of a possible production of SAF or hydrogen on or close to the airport.

In summary, the identified research gap lies in an all-encompassing approach to determining optimal energy systems for airports of varying sizes and environmental conditions. This will allow to assess which technologies are most robust under varying circumstances, to identify what technological or economical improvements are needed for the technologies to become attractive in the future, and to evaluate the feasibility of a fully-renewable and self-sufficient energy system for an airport. Overall, this study can serve as an advisory guide in the design of future energy systems, with the optimal system providing a benchmark for airport energy systems. Additionally, it will propose a range of metrics to quantify the resulting configuration, ensuring comparability for future research.

6

Research Plan

After finalisation of the literature review, the research plan is developed. It is constructed with the goal of answering the research question, as is outlined in Section 6.1. From the list of research sub-questions, the organisation of the thesis is set up in Section 6.2.

6.1. Research Question

The research question follows from the research gap explained in Chapter 5. It is defined as follows.

"To what extent could European airports, such as Schiphol airport, benefit from the concept of renewable energy hubs in reducing their sustainability impact and increasing their energy supply self-sufficiency while fulfilling the varying energy demand profile from airport operations and infrastructure?"

Several sub-questions can be derived from the research question, which are guiding the research. They can all be further divided into a set of tasks, which allows to answer the respective sub-question. The eight sub-questions defined for this thesis and their tasks are as follows:

- 1. What are the most important parameters influencing the overall energy demand of an international airport?
 - 1.1 Analyse airport energy demand division
 - 1.2 Identify and quantify main airport energy demands
 - 1.3 Identify factors influencing airport energy demand
 - 1.4 Identify the time-wise distribution of energy demands and influencing factors
 - 1.5 Define the relation between airport energy demands and factors influencing the demand
 - 1.6 Create a model calculating energy demands based on pre-defined factors for different airports

2. Which technologies are suitable for providing energy for an airport and how can these technologies be combined in an energy system?

- 2.1 Identify possible energy technologies
- 2.2 Choose realistic and promising technologies for implementation in a superstructure
- 2.3 Implement models for energy technologies in OSMOSE (with associated costs and emissions)
- 2.4 Verify and validate implementation of energy technologies
- 2.5 Build energy system superstructure
- 2.6 Verify and validate energy system superstructure

3. Is the introduction of energy storage technologies beneficial for levelling the intermittency of electricity supply as well as seasonal energy supply variations?

- 3.1 Identify possible energy storage technologies
- 3.2 Choose most suitable energy storage technologies
- 3.3 Implement models for energy storage in OSMOSE (with associated costs and emissions)
- 3.4 Verify and validate implementation of energy storage technologies
- 3.5 Confirm correct time-variations of energy demands and technologies
- 3.6 Determine new optimal energy system with storage technologies
- 3.7 Analyse effect of added storage options on results
- 4. Is it possible for an airport to have a fully self-sufficient energy system or what energy resources need to be imported in all cases?
 - 4.1 Optimize fully self-sufficient energy system with resources at the airport
 - 4.2 Determine (if needed) what amount of resources (for example electricity from the grid) needs to be imported
 - 4.3 Propose (if required) options for local energy production near to the airport for supplying the required demands

5. Is it an economically competitive option for an airport to fully rely on renewable energy technologies and resources for its full energy system?

- 5.1 Optimize fully-renewable energy system with developed superstructure
- 5.2 Optimize energy system also including non-renewable energy technologies
- 5.3 Compare costs and emissions of both energy system options
- 5.4 With research hypotheses 4, assess whether zero-emissions energy system is possible
- 5.5 With research hypotheses 7, look at what changes in costs would make renewable energy system attractive
- 5.6 Analyse and discuss different types of sustainability

6. Which combination of energy technologies is most robust to different airport environments and to changes in the airport operational circumstances?

- 6.1 Based on the energy model defined, determine energy demand for multiple airports of different sizes and in different environments
- 6.2 Identify technologies which can(not) be used in different environments or the varying efficiencies
- 6.3 Determine optimal energy system for the defined airports
- 6.4 Analyse changes in the chosen technologies in different environments
- 7. How sensitive is the optimal energy system to variations in the energy technology characteristics or costs?
 - 7.1 Determine most significant possible variations in terms of costs or technology characteristics for implemented technologies
 - 7.2 Run sensitivity analysis on identified characteristics
 - 7.3 Analyse impact of variations
 - 7.4 Determine what cost reductions, efficiency improvements, etc. are needed for technologies to be chosen
- 8. Is it an option to produce SAF on the airport and does it have an advantage compared to importing it from outside the airport's energy system?
 - 8.1 Identify possible SAF production technologies
 - 8.2 Choose most suitable SAF production technologies

- 8.3 Implement models for SAF production in OSMOSE (with associated costs and emissions)
- 8.4 Verify and validate implementation of SAF production technologies
- 8.5 Determine demand of SAF on airports
- 8.6 Optimize energy system with additional SAF demand and production technologies
- 8.7 Determine energy quantities that would need to be imported or additional energy production site sizes that would be required for supplying energy demands for SAF

6.2. Organisation

The research sub-questions and their related tasks can be used for defining the thesis schedule, for which a Gantt chart is developed, as can be seen in Figure 6.1. In the first six weeks, the literature was reviewed, and this research plan was developed. In the subsequent seven weeks until the midterm review, a first working demand model and energy system optimisation framework is constructed and verified. Subsequently, the next 14 weeks until the greenlight review are used for extending the base model and producing results. Finally, during the last five weeks, the thesis is submitted, and the defence is prepared.

								T						1						_	_			line		T		sis	T	
							(w)						2											03.01. (Internal Goal Deadline		17.01. (Greenlight Review)		04.02. (Hand-in written thesis		(in)
							05.07. 14.07. (Literature Review)						04.10. 11.10. (Midterm Review)											oal D		Rev		ritte		18.02. (Thesis Defence)
							ure						E E											al Ge		light		in v		Def
							erat						dtei											ern		eeu		-pue		esis
							. 5						. Ŭ											E.		Ð.		. Ĕ		Ē
	Midterm or		07.06.	14.06.	21.06.	28.06.	05.07. 14.07.	19.07.	26.07.	23.08.	20.09.	27.09.	04.10.	18.10.	25.10.	01.11.	08.11.	15.11.	22.11.	29.11. 06.12	13.12.	20.12.	27.12.	3.01	10.01.	7.01	24.01.	1.02	11.02.	3.02
	final		6	Ę.	3	8	8 7	Ä	5	ŝ	2(3	3 2	Ä	3	0	ő	ä	5 13	~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~	8 #	20	3	8	Ä	÷	3 6	5 8	닅	Ä
1. Parameters influencing energy demand?	Midterm	10 weeks				_	_		-			_		-		_	-	-	-			-			_	-	_	-	-	-
1.1. Analyze airport energy demand division	Midterm																													
1.2. Identify and quantify main demands	Midterm																													
1.3. Identify factors influencing demand 1.4. Identify time-wise distribution energy demands and factors	Midterm Midterm							-																						
1.5. Define relation airport energy demands and factors	Midterm																													
1.6. Create model	Midterm																													
2. Suitable energy technologies?	Midterm	18 weeks																												
2.1. Identify possible technologies 2.2. Choose initial technologies	Midterm Midterm																													
2.3. Implement technology models (+costs&emissions)	Midterm																													
2.4. Verify and validate implementations	Midterm													1												1			1	
2.5. Build superstructure	Midterm													1																
2.6. Verify and validate superstructure	Midterm																							_						
2.3.2. Add additional technology models (+costs&emissions) 2.4.2. Verify and validate additional implementations	Final Final							1												-									-	\vdash
3. Benefit of energy storage?	Midterm	13 weeks		-	_	_	_		-	_		-				_	_	-	_	_	_				_	-		_	+	-
3.1. Identify possible energy storage technologies	Midterm													L															L	
3.2. Choose suitable energy storage technologies	Midterm																													
3.3. Implement models energy storage	Midterm												_																	-
3.4. Verify and validate implementations 3.5. Confirm time-variations demands and technologies	Midterm Midterm																													
3.6. Determine new optimal energy system	Midterm																													
3.7. Analyze effect added storage options	Midterm																													
4. Self-sufficient energy system possible?	Midterm	6 weeks										_																		
4.1. Optimize fully self-sufficient energy system 4.2. Determine additional required resources	Midterm Midterm																							_						
4.2. Determine additional required resources 4.3. Propose options nearby energy production	Final																													
5. Renewable energy system competitive?	Midterm	9 weeks																												
5.1. Optimize fully-renewable energy system	Midterm																													
5.2. Optimize non-renewable energy system 5.3. Compare costs and emissions	Midterm Midterm																									-				
5.4. Assess possibility zero-emissions energy system	Midterm																													
5.5. What techn./cost changes needed for attractive renewable system	Final																													
5.6. Analyze sustainability	Final				_												_	_			_									
6. Robustness energy systems different environments & operations? 6.1. Energy demand for multiple airports	Final Final	5 weeks															-													
6.2. Technologies possible to use at different airports	Final																													
6.3. Determine optimal energy systems	Final																													
6.4. Analyze chosen technologies in different environments	Final																													
7. Sensitivity energy/technology costs and characteristics?	Final	8 weeks																												
7.1. Determine possible variations of costs or technology characteristics 7.2. Run sensitivity analysis	Final																													
7.3. Analyze impact of variations	Final																													
7.4. Determine cost reductions, efficiency improvements, etc. for techno																														
8. Produce SAF on site?	Final Final	3 weeks																												
8.1. Identify possible SAF production technologies 8.2. Choose suitable SAF production technologies	Final													1															1	
8.3. Implement models for SAF production	Final							1						1															1	
8.4. Verify and validate implementations	Final													1															1	
8.5. Determine demand of SAF	Final Final													1															1	
8.6. Optimize new energy system 8.7. Determine required additional energy resources	Final																							_						
0. Other thesis tasks	Both	32 weeks												1												1				
0.1. Write literature review chapter	Midterm																													
0.2. Write methodology chapter for thesis	Midterm																													
0.3. Write draft version thesis 0.4. Write final thesis (includes proofreading)	Final Final							1																						
0.5. Prepare thesis defence	Final							1						1												-1				
		· ·												·	·							·		_						
LEGEND	1																													
Building model																														
Analysing results Writing																														
Overall thesis deliverable deadline																														
Internal sub-question deadline																														

Figure 6.1: Gantt chart outlining Master's thesis organisation

Bibliography

- R. Florida, C. Mellander, and T. Holgersson. "Up in the air: the role of airports for regional economic development". In: *The Annals of Regional Science* 54.1 (2015), pp. 197–214. DOI: 10.1007/s00168-014-0651-z.
- [2] D. Schaar and L. Sherry. "Analysis of airport stakeholders". In: 2010 Integrated Communications, Navigation, and Surveillance Conference Proceedings. 2010. DOI: 10.1109/ICNSURV. 2010.5503233.
- [3] EPFL. Industrial Process and Energy Systems Engineering. URL: https://www.epfl.ch/ labs/ipese/ (visited on 02/02/2025).
- [4] M. Schaafsma. "Airports and Cities in Networks". In: *disP The Planning Review* 39.154 (2003), pp. 28–36. DOI: 10.1080/02513625.2003.10556854.
- [5] Cambridge Dictionary. airport. 2024. URL: https://dictionary.cambridge.org/de/ worterbuch/englisch/airport (visited on 07/10/2024).
- [6] Airports Council International World. Top 10 busiest airports in the world shift with the rise of international air travel demand. 2024. URL: https://aci.aero/2024/04/14/top-10busiest-airports-in-the-world-shift-with-the-rise-of-internationalair-travel-demand/ (visited on 06/29/2024).
- [7] R. Aehnelt. Sample infrastructure of a typical airport. Larger airports usually contain more runways and terminals. 2011. URL: https://en.wikipedia.org/wiki/Airport#/media/ File:Airport infrastructure.png (visited on 07/10/2024).
- [8] L. Sherry. Introduction to Airports Design and Operations. George Mason University, 2009.
- [9] M. Janić. *Landside Accessibility of Airports: Analysis, Modelling, Planning, and Design*. Cham, Switzerland: Springer International Publishing, 2019. ISBN: 978-3-319-76149-7.
- [10] Schiphol. Reducing Energy Consumption. URL: https://www.schiphol.nl/en/schipholgroup/page/reducing-energy-consumption/ (visited on 07/10/2024).
- [11] C. G. Corlu, R. de la Torre, A. Serrano-Hernandez, et al. "Optimizing Energy Consumption in Transportation: Literature Review, Insights, and Research Opportunities". In: *Energies* 13.5 (2020), p. 1115. DOI: 10.3390/en13051115.
- [12] F. Greer, J. Rakas, and A. Horvath. "Airports and environmental sustainability: a comprehensive review". In: *Environmental Research Letters* 15.10 (2020), p. 103007. DOI: 10.1088/1748-9326/abb42a.
- [13] Fraport AG Frankfurt Airport Services Worldwide. *Umwelterklärung 2022*. Frankfurt am Main, Germany, 2023.
- [14] Flughafen München GmbH. *Flughafen München Integrierter Bericht 2022*. München-Flughafen, Germany, 2023.
- [15] Flughafen Zürich AG. Integrierter Bericht 2022. Zürich-Flughafen, Switzerland, 2023.
- [16] Flughafen Wien AG. Umwelterklärung 2021 Wien. Wien-Flughafen, Austria, 2022.
- [17] Aeroporti di Roma. 2022 Integral Annual Report. Roma, Italy, 2023.
- [18] SEA Group Milan Airports. Sustainability Report 2022. Aeroporto Milano Linate, Italy, 2024.
- [19] Avinor. Annual and Sustainability Report 2023. Oslo, Norway, 2024.
- [20] Finavia. Annual and Responsibility Report 2023. 2024.
- [21] iGA Istanbul Airport. Sustainability Report 2021-2022. Istanbul, Turkey, 2022.

- [22] Groupe ADP. Traffic Paris Aéroport and Groupe ADP airports. 2024. URL: https://www. parisaeroport.fr/en/group/finance/investor-relations/traffic (visited on 06/20/2024).
- [23] Aeroports de Paris. Universal Registration Document 2023. Paris, France, 2024.
- [24] Schiphol. Annual Report 2023. en. Schiphol, The Netherlands, 2024.
- [25] AENA. Consolidated Management Report 2023. Madrid, Spain, 2024.
- [26] Gatwick Airport. Progress on our 2020 goals. Gatwick, UK.
- [27] ANA Environment and Sustainability Office. ANA Sustainability Report ANA 2022 Together for Positive Mobility. Lisboa, Portugal, 2023.
- [28] Athens International Airport. Annual & Sustainability Report. Athens, Greece, 2024.
- [29] Manchester Airports Group. Manchester Airport Corporate Social Responsibility Progress Update 2022/2023. Manchester, UK, 2023.
- [30] Manchester Airports Group. Traffic statistics | Manchester Airport. 2024. URL: https://www. manchesterairport.co.uk/about-us/publications/traffic-statistics/ (visited on 06/20/2024).
- [31] Manchester Airports Group. *Stansted Airport Corporate Social Responsibility Progress Update* 2022/2023. Manchester, UK, 2023.
- [32] Manchester Airports Group. December's strong performance rounds off 2022. 2023. URL: https: //mediacentre.stanstedairport.com/london-stansted-rounds-off-2022with-strong-passenger-performance-in-december/ (visited on 06/20/2024).
- [33] Copenhagen Airports. Annual Report 2023. Kastrup, Denmark, 2024.
- [34] Copenhagen Airports. Strong growth at Copenhagen Airport: Close to 27 million travellers. 2024. URL: https://www.cph.dk/en/about-cph/investor/traffic-statistics/2024/ 01/strong%20growth%20at%20copenhagen%20airport%20close%20to%2027% 20million%20travellers (visited on 06/20/2024).
- [35] Brussels Airport Company. Environmental Report 2019. Zaventem, Belgium, 2019.
- [36] Swedawia Airports. Annual & Sustainability Report 2023. Stockholm Arlanda, Sweden, 2024.
- [37] A. Costa, L. Blanes Restoy, C. Donnelly, et al. "Review of EU airport energy interests and priorities with repect to ICT, energy efficiency and enhanced building operation". In: Manchester, UK, 2012.
- [38] S. Ortega Alba. "Characterization, analysis and optimization of energy demand patterns in airports". PhD thesis. Santander, Spain: University of Cantabria, 2017.
- [39] R. Radhakrishnan. *Analysis on the energy distribution and feasibility of positive energy district at Stavanger airport*. Master thesis. Stavanger, Norway: University of Stavanger, 2020.
- [40] T. de Rubeis, I. Nardi, D. Paoletti, et al. "Multi-year consumption analysis and innovative energy perspectives: The case study of Leonardo da Vinci International Airport of Rome". In: *Energy Conversion and Management* 128 (2016), pp. 261–272. DOI: 10.1016/j.enconman.2016. 09.076.
- [41] S. Thiem, V. Danov, M. Metzger, et al. "Project-level multi-modal energy system design Novel approach for considering detailed component models and example case study for airports". In: *Energy* 133 (2017), pp. 691–709. DOI: 10.1016/j.energy.2017.05.159.
- [42] X. Liu, M. Li, X. Liu, et al. "Quantifying energy flexibility potential of ground electric vehicles in an airport with real behavior data". In: Sustainable Cities and Society 105 (2024), p. 105331. DOI: 10.1016/j.scs.2024.105331.
- [43] M. C. Kirca, A. McGordon, and T. Q. Dinh. "Multi-Input Multi-Output Model of Airport Infrastructure for Reducing CO2 Emissions". In: 2020 IEEE Vehicle Power and Propulsion Conference (VPPC). ISSN: 1938-8756. Nov. 2020, pp. 1–6. DOI: 10.1109/VPPC49601.2020.9330949.

- [44] G. Xianliang, X. Jingchao, L. Zhiwen, et al. "Analysis to energy consumption characteristics and influencing factors of terminal building based on airport operating data". In: Sustainable Energy Technologies and Assessments 44 (2021), p. 101034. DOI: 10.1016/j.seta.2021. 101034.
- [45] O. F. Yildiz, M. Yilmaz, and A. Celik. "Reduction of energy consumption and CO2 emissions of HVAC system in airport terminal buildings". In: *Building and Environment* 208 (2022), p. 108632. DOI: 10.1016/j.buildenv.2021.108632.
- [46] A. Kierzkowski and T. Kisiel. "Simulation Model for the Estimation of Energy Consumption of the Baggage Handling System in the Landside Area of the Airport". In: *Energies* 15.1 (2022), p. 256. DOI: 10.3390/en15010256.
- [47] G. Lodewijks, Y. Cao, N. Zhao, et al. "Reducing CO□ Emissions of an Airport Baggage Handling Transport System Using a Particle Swarm Optimization Algorithm". In: *IEEE Access* 9 (2021), pp. 121894–121905. DOI: 10.1109/ACCESS.2021.3109286.
- [48] D.-W. Bao, J.-Y. Zhou, Z.-Q. Zhang, et al. "Mixed fleet scheduling method for airport ground service vehicles under the trend of electrification". In: *Journal of Air Transport Management* 108 (2023), p. 102379. DOI: 10.1016/j.jairtraman.2023.102379.
- [49] D. Sigler, Q. Wang, Z. Liu, et al. "Route optimization for energy efficient airport shuttle operations – A case study from Dallas Fort worth International Airport". In: *Journal of Air Transport Management* 94 (2021), p. 102077. DOI: 10.1016/j.jairtraman.2021.102077.
- [50] M. Zoutendijk and M. Mitici. "Fleet scheduling for electric towing of aircraft under limited airport energy capacity". In: *Energy* 294 (2024), p. 130924. DOI: 10.1016/j.energy.2024. 130924.
- [51] L. Lin, X. Liu, T. Zhang, et al. "Energy consumption index and evaluation method of public traffic buildings in China". In: Sustainable Cities and Society 57 (2020), p. 102132. DOI: 10.1016/ j.scs.2020.102132.
- [52] B. Li, W. Zhang, J. Wang, et al. "Research and Analysis on Energy Consumption Features of Civil Airports". In: *IOP Conference Series: Earth and Environmental Science* 94.1 (2017), p. 012134. DOI: 10.1088/1755-1315/94/1/012134.
- [53] X. Gu, J. Xie, C. Huang, et al. "Prediction of the spatiotemporal passenger distribution of a large airport terminal and its impact on energy simulation". In: Sustainable Cities and Society 78 (2022), p. 103619. DOI: 10.1016/j.scs.2021.103619.
- [54] H. Zhao, Y. Xiang, Y. Shen, et al. "Resilience Assessment of Hydrogen-Integrated Energy System for Airport Electrification". In: *IEEE Transactions on Industry Applications* 58.2 (2022), pp. 2812–2824. DOI: 10.1109/TIA.2021.3127481.
- [55] S. Jin and Y. Li. "Analyzing the performance of electricity, heating, and cooling supply nexus in a hybrid energy system of airport under uncertainty". In: *Energy* 272 (2023), p. 127138. DOI: 10.1016/j.energy.2023.127138.
- [56] Q. Jia, H.-q. Hu, and J.-I. Yu. "The model construction in airport renewable energy system scale planning based in Multi-Criteria Decision Making". In: *Energy Reports* 7 (2021), pp. 92–98. DOI: 10.1016/j.egyr.2021.08.068.
- Y. Xiang, H. Cai, J. Liu, et al. "Techno-economic design of energy systems for airport electrification: A hydrogen-solar-storage integrated microgrid solution". In: *Applied Energy* 283 (2021), p. 116374. DOI: 10.1016/j.apenergy.2020.116374.
- [58] E. Cardona, A. Piacentino, and F. Cardona. "Energy saving in airports by trigeneration. Part I: Assessing economic and technical potential". In: *Applied Thermal Engineering* 26.14 (2006), pp. 1427–1436. DOI: 10.1016/j.applthermaleng.2006.01.019.
- [59] E. Cardona, P. Sannino, A. Piacentino, et al. "Energy saving in airports by trigeneration. Part II: Short and long term planning for the Malpensa 2000 CHCP plant". In: *Applied Thermal Engineering* 26.14 (2006), pp. 1437–1447. DOI: 10.1016/j.applthermaleng.2006.01.020.
- [60] B. Kılkış and Ş. Kılkış. "New exergy metrics for energy, environment, and economy nexus and optimum design model for nearly-zero exergy airport (nZEXAP) systems". In: *Energy* 140 (2017), pp. 1329–1349. DOI: 10.1016/j.energy.2017.04.129.

- [61] Y. Zhang, P. Si, Y. Feng, et al. "Operation strategy optimization of BCHP system with thermal energy storage: A case study for airport terminal in Qingdao, China". In: *Energy and Buildings* 154 (2017), pp. 465–478. DOI: 10.1016/j.enbuild.2017.08.059.
- [62] Y. Zhou. "Low-carbon transition in smart city with sustainable airport energy ecosystems and hydrogen-based renewable-grid-storage-flexibility". In: *Energy Reviews* 1.1 (2022), p. 100001. DOI: 10.1016/j.enrev.2022.100001.
- [63] M. Prussi, L. Laveneziana, D. Misul, et al. "A model-based approach to long-term energy planning: the case-study of the Turin Airport". In: *Journal of Physics: Conference Series* 2648.1 (2023), p. 012034. DOI: 10.1088/1742-6596/2648/1/012034.
- [64] S. Baek, H. Kim, and H. J. Chang. "Optimal Hybrid Renewable Airport Power System: Empirical Study on Incheon International Airport, South Korea". In: Sustainability 8.6 (2016), p. 562. DOI: 10.3390/su8060562.
- [65] Y. Tian, X. Zhang, and N. Balta-Ozkan. "Multi-Energy Systems in Green Airports". In: CIRED 2021 - The 26th International Conference and Exhibition on Electricity Distribution. Vol. 2021. 2021, pp. 2043–2047. DOI: 10.1049/icp.2021.1561.
- [66] F. A. Plazas-Niño, N. R. Ortiz-Pimiento, and E. G. Montes-Páez. "National energy system optimization modelling for decarbonization pathways analysis: A systematic literature review". In: *Renewable and Sustainable Energy Reviews* 162 (2022), p. 112406. DOI: 10.1016/j.rser. 2022.112406.
- [67] R. Zafar, A. Mahmood, S. Razzaq, et al. "Prosumer based energy management and sharing in smart grid". In: *Renewable and Sustainable Energy Reviews* 82 (2018), pp. 1675–1684. DOI: 10.1016/j.rser.2017.07.018.
- [68] H. Rhim. *How Does Particle Swarm Optimization Work*? 2024. URL: https://www.baeldung. com/cs/pso (visited on 07/11/2024).
- [69] F. Belfiore. "District heating and cooling systems to integrate renewable energy in urban areas". PhD thesis. Lausanne, Switzerland: EPFL, 2021.
- [70] V. Hariprasad, A. V. Anand, J. Sahana, et al. "Energy support to airports by wind systems". In: 2017 International conference of Electronics, Communication and Aerospace Technology (ICECA). Vol. 2. Coimbatore, India, Apr. 2017, pp. 748–751. DOI: 10.1109/ICECA.2017. 8212767.
- [71] J. Buongiorno. Flexible Siting Criteria and Staff Minimization for Micro-Reactors. English. Cambridge, MA, USA: Massachusetts Inst. of Technology (MIT), 2022. DOI: 10.2172/1898105.
- [72] T. Elmer, M. Worall, S. Wu, et al. "Fuel cell technology for domestic built environment applications: State of-the-art review". In: *Renewable and Sustainable Energy Reviews* 42 (2015), pp. 913–931. DOI: 10.1016/j.rser.2014.10.080.
- [73] International Renewable Energy Agency. *Geothermal power: Technology brief*. Abu Dhabi, UAE, 2017.
- [74] M. Balat. "Hydropower Systems and Hydropower Potential in the European Union Countries". In: Energy Sources, Part A: Recovery, Utilization, and Environmental Effects 28.10 (2006), pp. 965– 978. DOI: 10.1080/00908310600718833.
- [75] S. Astariz, A. Vazquez, and G. Iglesias. "Evaluation and comparison of the levelized cost of tidal, wave, and offshore wind energy". In: *Journal of Renewable and Sustainable Energy* 7.5 (2015), p. 053112. DOI: 10.1063/1.4932154.
- [76] V. Benda and L. Černá. "PV cells and modules State of the art, limits and trends". In: *Heliyon* 6.12 (2020). DOI: 10.1016/j.heliyon.2020.e05666.
- [77] A. Anurag, J. Zhang, J. Gwamuri, et al. "General Design Procedures for Airport-Based Solar Photovoltaic Systems". In: *Energies* 10.8 (2017), p. 1194. DOI: 10.3390/en10081194.
- [78] H. K. Salim, R. A. Stewart, O. Sahin, et al. "Drivers, barriers and enablers to end-of-life management of solar photovoltaic and battery energy storage systems: A systematic literature review".
 In: *Journal of Cleaner Production* 211 (2019), pp. 537–554. DOI: 10.1016/j.jclepro. 2018.11.229.

- [79] S. Sreenath, K. Sudhakar, and A. F. Yusop. "Solar PV in the airport environment: A review of glare assessment approaches & metrics". In: *Solar Energy* 216 (2021), pp. 439–451. DOI: 10.1016/j.solener.2021.01.023.
- [80] P. G. McCormick and H. Suehrcke. "The effect of intermittent solar radiation on the performance of PV systems". In: Solar Energy 171 (2018), pp. 667–674. DOI: 10.1016/j.solener.2018. 06.043.
- [81] D. J. Willis, C. Niezrecki, D. Kuchma, et al. "Wind energy research: State-of-the-art and future research directions". In: *Renewable Energy* 125 (2018), pp. 133–154. DOI: 10.1016/j. renene.2018.02.049.
- [82] L. Cuadra, I. Ocampo-Estrella, E. Alexandre, et al. "A study on the impact of easements in the deployment of wind farms near airport facilities". In: *Renewable Energy* 135 (2019), pp. 566– 588. DOI: 10.1016/j.renene.2018.12.038.
- [83] A. S. Darwish and R. Al-Dabbagh. "Wind energy state of the art: present and future technology advancements". In: *Renewable Energy and Environmental Sustainability* 5 (2020), p. 7. DOI: https://doi.org/10.1051/rees/2020003.
- [84] J. Beauson, A. Laurent, D. P. Rudolph, et al. "The complex end-of-life of wind turbine blades: A review of the European context". In: *Renewable and Sustainable Energy Reviews* 155 (2022), p. 111847. DOI: 10.1016/j.rser.2021.111847.
- [85] B. Zohuri. *Nuclear Micro Reactors*. Cham, Switzerland: Springer International Publishing, 2020. ISBN: 978-3-030-47224-5.
- [86] J. W. Sterbentz, J. E. Werner, M. G. McKellar, et al. Special Purpose Nuclear Reactor (5 MW) for Reliable Power at Remote Sites Assessment Report. Idaho Falls, ID, USA: Idaho National Lab. (INL), 2017. DOI: 10.2172/1410224.
- [87] S. N. Silva and J. Á. d. Castillo. "An Approach of the Hydropower: Advantages and Impacts. A Review". In: *Journal of Energy Research and Reviews* 8.1 (2021), pp. 10–20. DOI: 10.9734/ jenrr/2021/v8i130201.
- [88] A. Uihlein. "Life cycle assessment of ocean energy technologies". In: The International Journal of Life Cycle Assessment 21.10 (2016), pp. 1425–1437. DOI: 10.1007/s11367-016-1120-y.
- [89] J. Wang. "Barriers of scaling-up fuel cells: Cost, durability and reliability". In: Energy 80 (2015), pp. 509–521. DOI: 10.1016/j.energy.2014.12.007.
- [90] I. Lee, J. W. Tester, and F. You. "Systems analysis, design, and optimization of geothermal energy systems for power production and polygeneration: State-of-the-art and future challenges". In: *Renewable and Sustainable Energy Reviews* 109 (2019), pp. 551–577. DOI: 10.1016/j. rser.2019.04.058.
- [91] L. Buijze, L. v. Bijsterveldt, H. Cremer, et al. "Review of induced seismicity in geothermal systems worldwide and implications for geothermal systems in the Netherlands". In: *Netherlands Journal of Geosciences* 98 (2019). DOI: 10.1017/njg.2019.6.
- [92] P. C. Slorach and L. Stamford. "Net zero in the heating sector: Technological options and environmental sustainability from now to 2050". In: *Energy Conversion and Management* 230 (2021), p. 113838. DOI: 10.1016/j.enconman.2021.113838.
- [93] L. Gómez Pérez. *Heat Pump vs Boiler Comparison Guide* (2024). 2024. URL: https://www. boilerguide.co.uk/compare/types/boiler-vs-heat-pump (visited on 02/01/2025).
- [94] R. Saidur, E. A. Abdelaziz, A. Demirbas, et al. "A review on biomass as a fuel for boilers". In: *Renewable and Sustainable Energy Reviews* 15.5 (2011), pp. 2262–2289. DOI: 10.1016/j. rser.2011.02.015.
- [95] T. S. Ge, R. Z. Wang, Z. Y. Xu, et al. "Solar heating and cooling: Present and future development". In: *Renewable Energy* 126 (2018), pp. 1126–1140. DOI: 10.1016/j.renene.2017.06.081.
- [96] P. Carroll, M. Chesser, and P. Lyons. "Air Source Heat Pumps field studies: A systematic literature review". In: *Renewable and Sustainable Energy Reviews* 134 (2020), p. 110275. DOI: 10.1016/j.rser.2020.110275.

- [97] D. Menegazzo, G. Lombardo, S. Bobbo, et al. "State of the Art, Perspective and Obstacles of Ground-Source Heat Pump Technology in the European Building Sector: A Review". In: *Energies* 15.7 (2022), p. 2685. DOI: 10.3390/en15072685.
- [98] M. A. Bagherian and K. Mehranzamir. "A comprehensive review on renewable energy integration for combined heat and power production". In: *Energy Conversion and Management* 224 (2020), p. 113454. DOI: 10.1016/j.enconman.2020.113454.
- [99] A. De Vita, P. Capros, L. Paroussos, et al. EU reference scenario 2020: energy, transport and GHG emissions : trends to 2050. Luxembourg, Luxembourg: Publications Office of the European Union, 2021. ISBN: 978-92-76-39356-6.
- [100] S. Braungardt, B. Tezak, J. Rosenow, et al. "Banning boilers: An analysis of existing regulations to phase out fossil fuel heating in the EU". In: *Renewable and Sustainable Energy Reviews* 183 (2023), p. 113442. DOI: 10.1016/j.rser.2023.113442.
- [101] P. Kotek, A. Selei, B. Takácsné Tóth, et al. "What can the EU do to address the high natural gas prices?" In: *Energy Policy* 173 (2023), p. 113312. DOI: 10.1016/j.enpol.2022.113312.
- [102] Q. Zhang, K. Yang, Y. Hu, et al. "Unveiling the impact of geopolitical conflict on oil prices: A case study of the Russia-Ukraine War and its channels". In: *Energy Economics* 126 (2023), p. 106956. DOI: 10.1016/j.eneco.2023.106956.
- [103] R&R Refrigeration & Air Conditioning, Inc. Chillers: Vapor Compression vs. Vapor Absorption. 2023. URL: https://getcooled.com/chillers-vapor-compression-vs-vaporabsorption/ (visited on 07/13/2024).
- [104] B. R. Hughes, H. N. Chaudhry, and S. A. Ghani. "A review of sustainable cooling technologies in buildings". In: *Renewable and Sustainable Energy Reviews* 15.6 (2011), pp. 3112–3120. DOI: 10.1016/j.rser.2011.03.032.
- [105] S. H. L. Yim, M. E. J. Stettler, and S. R. H. Barrett. "Air quality and public health impacts of UK airports. Part II: Impacts and policy assessment". In: *Atmospheric Environment* 67 (2013), pp. 184–192. DOI: 10.1016/j.atmosenv.2012.10.017.
- [106] A. R. Dehghani-sanij, M. Soltani, and K. Raahemifar. "A new design of wind tower for passive ventilation in buildings to reduce energy consumption in windy regions". In: *Renewable and Sustainable Energy Reviews* 42 (2015), pp. 182–195. DOI: 10.1016/j.rser.2014.10.018.
- [107] J. S. Brar. Study, modeling, analysis, evaluation, selection and performance improvement of air conditioning system. Patiala, India: Department of Mechanical Engineering, Thapar University, 2012.
- [108] W. Hüls Güido, W. Lanser, S. Petersen, et al. "Performance of absorption chillers in field tests". In: Applied Thermal Engineering 134 (2018), pp. 353–359. DOI: 10.1016/j.applthermaleng. 2018.02.013.
- [109] J. Rugolo and M. J. Aziz. "Electricity storage for intermittent renewable sources". In: Energy & Environmental Science 5.5 (2012), pp. 7151–7160. DOI: 10.1039/C2EE02542F.
- [110] A. G. Olabi, C. Onumaegbu, T. Wilberforce, et al. "Critical review of energy storage systems". In: *Energy* 214 (2021), p. 118987. DOI: 10.1016/j.energy.2020.118987.
- [111] E. Bazdar, M. Sameti, F. Nasiri, et al. "Compressed air energy storage in integrated energy systems: A review". In: *Renewable and Sustainable Energy Reviews* 167 (2022), p. 112701. DOI: 10.1016/j.rser.2022.112701.
- [112] R. Dennis. Hydrogen Storage A brief overview of hydrogen storage options. Pittsburgh, PA, USA, 2020. URL: https://www.netl.doe.gov/sites/default/files/2020-02/ TMCES/Rich%20Dennis%20H2%20Storage%20for%20Energy%20Storage%20Meeting_ Final%20V1.pdf (visited on 07/12/2024).
- [113] I. Sarbu and C. Sebarchievici. "A Comprehensive Review of Thermal Energy Storage". In: Sustainability 10.1 (2018), p. 191. DOI: 10.3390/su10010191.
- [114] A. A. K. Arani, H. Karami, G. B. Gharehpetian, et al. "Review of Flywheel Energy Storage Systems structures and applications in power systems and microgrids". In: *Renewable and Sustainable Energy Reviews* 69 (2017), pp. 9–18. DOI: 10.1016/j.rser.2016.11.166.

- [115] M. A. Hannan, S. B. Wali, P. J. Ker, et al. "Battery energy-storage system: A review of technologies, optimization objectives, constraints, approaches, and outstanding issues". In: *Journal of Energy Storage* 42 (2021), p. 103023. DOI: 10.1016/j.est.2021.103023.
- [116] S. Karthikeyan, B. Narenthiran, A. Sivanantham, et al. "Supercapacitor: Evolution and review". In: *Materials Today: Proceedings* 46 (2021), pp. 3984–3988. DOI: 10.1016/j.matpr.2021. 02.526.
- [117] J. Andersson and S. Grönkvist. "Large-scale storage of hydrogen". In: International Journal of Hydrogen Energy 44.23 (2019), pp. 11901–11919. DOI: 10.1016/j.ijhydene.2019.03. 063.
- [118] N. Klopčič, I. Grimmer, F. Winkler, et al. "A review on metal hydride materials for hydrogen storage". In: *Journal of Energy Storage* 72 (2023), p. 108456. DOI: 10.1016/j.est.2023. 108456.
- [119] A. Kovač, M. Paranos, and D. Marciuš. "Hydrogen in energy transition: A review". In: International Journal of Hydrogen Energy 46.16 (2021), pp. 10016–10035. DOI: 10.1016/j. ijhydene.2020.11.256.
- [120] H. H. Cho, V. Strezov, and T. J. Evans. "Environmental impact assessment of hydrogen production via steam methane reforming based on emissions data". In: *Energy Reports* 8 (2022), pp. 13585–13595. DOI: 10.1016/j.egyr.2022.10.053.
- [121] P. J. Megía, A. J. Vizcaíno, J. A. Calles, et al. "Hydrogen Production Technologies: From Fossil Fuels toward Renewable Sources. A Mini Review". In: *Energy & Fuels* 35.20 (2021), pp. 16403– 16415. DOI: 10.1021/acs.energyfuels.1c02501.
- [122] R. Pinsky, P. Sabharwall, J. Hartvigsen, et al. "Comparative review of hydrogen production technologies for nuclear hybrid energy systems". In: *Progress in Nuclear Energy* 123 (2020), p. 103317. DOI: 10.1016/j.pnucene.2020.103317.
- [123] International Renewable Energy Agency. Green hydrogen cost reduction: Scaling up electrolysers to meet the 1.5C climate goal. Abu Dhabi, UAE: International Renewable Energy Agency, 2020. ISBN: 978-92-9260-295-6.
- [124] M. Elhassan, R. Abdullah, M. R. R. Kooh, et al. "Hydrothermal liquefaction: A technological review on reactor design and operating parameters". In: *Bioresource Technology Reports* 21 (2023), p. 101314. DOI: 10.1016/j.biteb.2022.101314.
- [125] IATA. *IATA Sustainable Aviation Fuel Roadmap*. Montreal, Canada Geneva, Switzerland: International Air Transport Association, 2015.
- [126] Council of the EU. RefuelEU aviation initiative: Council adopts new law to decarbonise the aviation sector. 2023. URL: https://www.consilium.europa.eu/en/press/pressreleases/2023/10/09/refueleu-aviation-initiative-council-adopts-newlaw-to-decarbonise-the-aviation-sector/ (visited on 07/13/2024).
- [127] E. Cabrera and J. M. M. de Sousa. "Use of Sustainable Fuels in Aviation—A Review". In: *Energies* 15.7 (2022), p. 2440. DOI: 10.3390/en15072440.
- [128] J. Granacher. "Decision support for the design of integrated biorefineries". PhD thesis. Lausanne, Switzerland: EPFL, 2023.